UPTAKE OF MICRONUTRIENTS BY ACACIA SENEGAL VARIETIES AND ITS POSSIBLE EFFECT ON GUM ARABIC QUALITY

BY

JOSEPH KIPKOSGEI LELON

REGISTRATION NUMBER: 180/7953/1996

A THESIS SUBMITTED IN PARTIAL FULFILLMENT FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN CHEMISTRY OF THE UNIVERSITY OF NAIROBI

2008

UNIVERSITY OF NAIROBI LIBRARY P. O. Box 30197 NAIROBI



DECLARATION

1. I declare that this thesis is of my own composition and apart from acknowledged assistance, is a record of my own research. The material has never been presented before to the University of Nairobi or any other establishment for academic award.

> Joseph Kipkosgei Lelon **Department of Chemistry**

Signed Strile

2. This thesis has been submitted with our knowledge as University Supervisors:

Professor I. O. Jumba Department of Chemistry

Professor J. K. Keter

Department of Soil Science

Dr. F.D.O. Oduor

Department of Chemistry

Signed Hit .13th Nov. 200

hm 17/11/08

DEDICATION

This thesis is dedicated to my children, Grace Rirenet, David Kipkemoi Bartilol, Michael Kiprotich and my wife Magdaline W. Lelon for their support during the entire period. My late father and mother, Lelon Cheptum Chepkener and Kabilo Lelon Cheptum Sei for their sacrifice to educate me. Without them I could not have come this far.

ABSTRACT

A study was conducted to establish the uptake of micronutrients by two *Acacia senegal* varieties established under arid and semi-arid conditions, and its subsequent effect on the quality of gum arabic exudates. Soil and gum arabic samples from the experimental sites at Solit, Kapkun, Kimorok and Maoi in Marigat division, Baringo district, were collected, dried and analysed to establish their baseline physical and chemical characteristics. Glasshouse pot experiments were conducted using the same soil and vermiculite media to determine uptake of copper (Cu), iron (Fe), manganese (Mn) and zinc (Zn) by *Acacia senegal* seedlings. A randomized, split-plot design that simulated the environmental conditions in the field was used in treatments varying from normal, low and high concentrations of the micronutrients applied.

The soil types varied significantly (P < 0.05) in the level of calcium (17.3, 5.7, 6.4 and 5.0 cmol. (⁺)/kg) and magnesium (6.1, 3.3, 4.0 and 2.7 cmol. (⁺)/kg), respectively, at the study sites. The average copper uptake (103 ppm) by *Acacia senegal* variety *senegal* in Solit on dry matter (DM) basis was significantly higher (P < 0.05) than that at Maoi (41 ppm), Kimorok (33 ppm) and Kapkun (30 ppm). Gum arabic from *Acacia senegal* variety *senegal* in Kapkun and Solit had concentrations of 45 and 40 ppm Cu, which reveal that *Acacia senegal* variety *senegal* variety *senegal* tends to take higher Cu levels (108 and 156 ppm) from the soils than that of *Acacia senegal* variety *kerensis* at Kimorok and Maoi (38 and 32 ppm). Iron and manganese uptake by the variety (654 and 638 ppm) at Kapkun was significantly higher (P < 0.05) than that of the variety at Kimorok (366 ppm Fe and 307 Mn), respectively.

Zinc uptake by *Acacia senegal* variety *kerensis* at Kimorok (533 ppm) was significantly higher (P < 0.05) than in Maoi (233 ppm), Kapkun (224 ppm) and Solit (141 ppm). The gum arabic had ash content of 2.88%. Analysis showed that the ash content had a concentration of Zn (124 ppm) which indicated that *Acacia senegal* variety *kerensis* tend to take higher concentrations of Zn (533 ppm) from the soils than that of *Acacia senegal* variety *senegal* (141 ppm).

Moisture, ash and volatile matter contents in gum arabic from *Acacia senegal* variety *senegal* were 14.9%, 3.16% and 64.24%, while *Acacia senegal* variety *kerensis* had 15.2%, 2.88% and 63.8%, respectively. Gum arabic obtained from *Acacia senegal* variety *senegal* had higher levels of copper (45 ppm), iron (1415 ppm) and manganese (109 ppm) compared to variety *kerensis* which by contrast had higher levels of zinc (124 ppm) and nitrogen (0.34 %), respectively. All these levels however fell within the ranges quoted in the International Standard Specifications (0.26% - 0.39% N, iron (730 – 2490 ppm), manganese (69 – 117 ppm), zinc (45 – 111 ppm), ash 2 - 4% and moisture 13 - 15%, respectively). The quality parameters of gum arabic may partly depend on soil characteristics, climate, and availability of the nutrients in the soils.

ACKNOWLEDGEMENTS

I would like sincerely to thank God Almighty for having granted me the opportunity to study and sustained and protected me from all tribulations. All the glory and honour go to him forever and ever.

I would like to thank very much and express my deep appreciation to my supervisors; Prof. I.O. Jumba, Prof. J. K. Keter and Dr. F.D.O. Oduor for their guidance, constructive criticism and patience.

I would also like to express my sincere gratitude to the Director, Kenya Forestry Research Institute (KEFRI) for provision of study leave, laboratory facilities, glasshouse experiments and cordial working atmosphere. This support is indeed highly appreciated.

I am very grateful to the Permanent Secretary, Ministry of Education, Science and Technology through the Secretary, National Council for Science and Technology for funding the Project without which this work could have been unattainable.

The Centre Director, KEFRI, Muguga Regional Research Centre, Mr. Ely Mwanza deserves my great appreciation for his kind assistance with purchase of research materials, moral support and provision of logistics to field sites.

My many thanks go to the former chief of Kimalel location, Mr. Chelal and the people of Kimalel and Kaibosoi locations, Marigat division of Baringo district, for the provision of the four Study Sites.

I would like to thank Mr. James B. Matata, Assistant Director, KARI- Agricultural research Fund secretariat for his moral and logistical support on monitoring and evaluation of the field project sites.

I am also very thankful to Heads of Libraries, KARI –Muguga, NARL - Nairobi, KARI headquarters and KEFRI headquarters for provision of all references cited in the thesis.

۷

I appreciate a lot the assistance of my colleagues and friends at the Department of Chemistry, UON and KEFRI for their encouragement and moral support during my studies.

I am also very grateful to all the technical staff of KEFRI Soil laboratories, particularly Omollo, Kamonde, Zadok, Mary Gathara, Corety, Kibera and others for their assistance in undertaking the soil physical and chemical analysis, field work and glasshouse data collection. My special thanks go to Mr. Nyogot for soil map illustration of study area, and to Mr. James Mwangi for statistical analysis of data using SAS and SPSS computer software.

Lastly, but not least I am deeply indebted to my beloved wife Magdaline W. Lelon and the children, Grace, David and Michael for their patience, long suffering, support and prayers. Without them my studies could have been unattainable.

CONTENTS

DECLARATION	i
DEDICATION	ii
ABSTRACT	iii
ACKNOWLEDGEMENTS	v
CONTENTS	vii
List of Tables	xi
List of Figures	xii
List of Plates	xii
List of Abbreviations	xiv

CHAPTER ONE	1
INTRODUCTION	1
1.0 Introduction and Background	1
1.1 Rationale	5
1.2 Problem Statement	6
1.3 Objectives	7
1.4 Organisation of the Work presented	8

CHAPTER TWO	9
LITERATURE REVIEW	9
2.1 Physical properties of Soil	9
2.1.3 Functions of Mineral Elements in Plants	14
2.1.4 Description and Distribution of <i>Acacia senegal</i> varieties in Kenya	28

CHAPTER THREE
MATERIALS AND METHODS
3.1 The Study Area
3.1.1 Introduction
3.1.2 Selection and Description of Study Sites
3.2.1 Soil Sampling and Pretreatment
3.3 Soil Analysis
3.3.1 Physical properties of Soils
3.4 Determination of Chemical properties of Soils41
3.4.1 Organic carbon
3.4.2 Total Nitrogen by Kjeldhal Method
3.5 Soil pH and Electrical conductivity
3.5.1 Extractable Phosphorus
3.5.2 Soil Extractable Macronutrients
3.5.3 Soil Extractable Micronutrients
3.6 Micronutrients Uptake Studies
3.6.1 Seed Collection
3.6.2 Seed Germination in the Controlled Growth Chamber
3.6.3 Glasshouse Experimental Design
3.6.4 Preparation of Hydroponics Solution Culture (Balanced Nutrient Solution Media)48
3.6.5 Preparation of Low Micronutrient Medium
3.6.6 Preparation of High Micronutrient Medium
3.6.7 Application of Nutrient Solutions, Monitoring and Data Collection Procedure
3.7 Plant sampling and Tissue Analysis
3.7.1 Determination of Micronutrients in Tree seedlings of Acacia senegal varieties

3.8 Gum arabic Analysis	52
3.8.1 Determination of the Moisture and Ash contents	52
3.8.2 Determination of Internal energy in gum arabic samples	52
3.8.3 Determination of the Volatile matter in gum arabic samples	53
3.8.4 Determination of Nitrogen content and Trace elements in gum arabic samples	53
3.9 Statistical Methods of Data Analysis	53

CHAPTER FOUR	56
CHARACTERIZATION OF SOILS AND SOIL FERTILITY STATUS	56
4.1 Physical Properties of Soils	56
4.2 Chemical Properties of Soils	61
4.3 Soil Fertility Status in the Study Sites	79
4.4 Conclusions	80

CHAPTER FIVE
STUDIES ON MICRONUTRIENTS UPTAKE BY ACACIA SENEGAL VARIETIES IN
A SIMULATED ARID AND SEMI-ARID ENVIRONMENT
5.1 Determination of Micronutrients Uptake by Acacia senegal varieties
5.2 Response of Acacia senegal varieties to Micronutrient uptake
5.2.1 Concentrations of Micronutrient uptake by Acacia senegal varieties
5.3 Conclusions

CHAPTER SIX
INFLUENCE OF ACACIA SENEGAL VARIETIES ON QUALITY OF GUM ARABIC
6.0 Introduction
6.1 Quality of Gum Arabic115
6.2 Physical properties of Gum arabic
6.3 Physical Properties of Gum arabic in the four Study sites117
6.4 Influence of Soil Composition on the Quality of Gum arabic Varieties
6.4.1 Nitrogen
6.4.2 Copper
6.4.3 Iron
6.4.4 Manganese
6.4.5 Zinc
6.5 Conclusions

CHAPTER SEVEN	
INTEGRATING DISCUSSION AND RECOMMENDATIONS	
REFERENCES	133
APPENDICES	157

List of Tables

Table 3.1 Categorizations of Nutrient Treatments 50
Table 4.1: The Physical properties of Soils
Table 4.2: Characterization Criteria for soils 61
Table 5.1: Uptake of Micronutrients by Acacia senegal seedlings, Heights and Diameter at
ground level on Normal and Dosage Treatments in Soil Medium
Table 5.2: Uptake of Micronutrients by Acacia senegal seedlings, Heights and Diameter at
ground level on Normal and Dosage Treatments in Vermiculite Medium
Table 5.3 Comparisons of Heights on Normal treatments with Low Dosage treatments93
Table 5.4 Comparisons of Heights on Normal treatments with High Dosage treatments98
Table 5.5: Diameter at ground level of tree seedlings for normal and dosage treatments 107
Table 5.6 Concentration of micronutrients uptake by plants 111
Table 5.7 Concentration Ranges of Micronutrients uptake by Acacia senegal varieties 112
Table 6.1 International Specifications of Quality parameters of Gum Arabic 115
Table 6.2: Physical properties of Gum arabic from the study sites
Table 6.3 Chemical compositions of Soils and Gum Arabic

List of Figures

Figure 3.1: Location of the Study Sites – Baringo District
Figure 4.1 Soil Texture in the Study sites57
Figure 4.2 Bulk and Particle densities in Sites
Figure 4.3 Porosity and Percolation in Sites
Figure 4.4 Soil Moisture content in Sites
Figure 4.5 Comparison of SOM
Figure 4.6 Comparison of Soil N65
Figure 4.7 Comparison of Soil P
Figure 4.8 Comparison of Soil K67
Figure 4.9 Comparison of Soil Ca
Figure 4.10 Comparison of Soil Mg70
Figure 4.11 Comparison of Soil Cu
Figure 4.12 Comparison of Soil Fe73
Figure 4.13 Comparison of Soil Mn74
Figure 4.14 Comparison of Soil Zn76
Figure 4.15: Soil Map on Characterization of Study Sites
Figure 5.1 Comparisons of Heights on Normal treatments in Soil and Vermiculite media91
Figure 5.2 Comparisons of Heights on Dosage treatments Soil and Vermiculite media92
Figure 5.3 Comparisons of Heights on Cu Normal treatments with Low Cu Dosage
Figure 5.4 Comparisons of Heights on Fe Normal treatments with low Fe Dosage
Figure 5.5 Comparisons of Heights on Mn Normal treatments with low Mn Dosage
Figure 5.6 Comparisons of Heights on Zn Normal treatments with low Zn Dosage
Figure 5.7 Comparisons of Heights on Cu Normal treatments with High Cu Dosage
Figure 5.8 Comparisons of Heights on Fe Normal treatments with high Fe Dosage

Figure 5.9 Comparisons of Heights on Mn Normal treatments with high Mn Dosage101
Figure 5.10 Comparisons of Heights on Zn Normal treatments with high Zn Dosage102
Figure 5.11 Comparisons of Diameter at ground level on Normal treatments in Soil and
Vermiculite media
Figure 5.12 Comparisons of Diameter at ground level on Dosage treatments in Soil and
Vermiculite media
Figure 6.1 Comparison of Soil Nitrogen and Gum arabic Nitrogen
Figure 6.2 Comparison of Soil Cu and Cu concentrations in Gum arabic
Figure 6.3 Comparison of Soil Fe and Fe concentrations in Gum arabic
Figure 6.4 Comparison of Soil Mn and Mn concentrations in Gum arabic
Figure 6.5 Comparison of Soil Zn and Gum arabic Zn concentrations

List of Plates

Plate 3.1: Acacia senegal var. senegal and other closely related species	5
Plate 3.2: Acacia senegal var. kerensis and other closely related species	5
Plate 3.3: Acacia senegal var. senegal and kerensis with other closely related species	5
Plate 3.4: Acacia senegal var. kerensis and other closely related species	7
Plate 3.5: Soil sampling in the Study sites	8
Plate 1 Controlled Growth Chamber for Seed Germination	5
Plate 2 Glasshouse Experiment	7
Plate 3 Application of dosage treatments to Acacia senegal seedlings	9

List of Abbreviations

ADI	Acceptable daily intake
ANOVA	Analysis of variance
ASALs	Arid and Semi arid lands
cmol (⁺)/kg	Centimoles of ions per kilogram
Dgl	Diameter at ground level
FAO	Food and Agricultural Organization
GOK	Government of Kenya
Ht	Heights
IPAL	Integrated Project in Arid Lands
JECFA	Joint Expert Committee on Food Additives (FAO/WHO)
KARI	Kenya Agricultural Research Institute
KEFRI	Kenya Forestry Research Institute
KK	Kimorok
KN	Kapkun

LSM	Least Significant Means
MI	Maoi
MTD	Maximum tolerated dose
mS/cm	millisiemens per centimetre
NARL	National Agricultural Research Laboratories
NAS	National Academy of Sciences (U S A)
PPM	Parts per million
S.E.	Standard error
SOM	Soil organic matter
ST	Solit
WHO	World Health Organization

-

CHAPTER ONE

INTRODUCTION

1.0 Introduction and Background

Kenya's total land surface area is approximately 590,000 square kilometres. About 80% of this area is arid and semi-arid lands (ASALs), which support 20% of human population with over 50% livestock population and significant proportion of wildlife (KEFRI, 1992; GOK, 1993). The remaining 20% of the land surface sustains 80% of Kenya's population and is being extensively exploited for agricultural production and plantation forestry. This reflects an imbalance since the area cannot sustain such a high population and its requirements of fuelwood, shelter and food production (Republic of Kenya, 1989a, b). In addition, the high population growth rate in the high and medium potential areas has exceeded the carrying capacity resulting into migration of people to the fragile ASAL ecosystem in search of settlement and farming (Lusigi, 1984).

The ASALs are characterised by low and erratic rainfall (150 - 750 mm annually), high potential evaporation rates (1900- 2500 mm/day), high mean annual temperatures (28° C) and poor soils (Sombroek *et al.*, 1982). The soils are variable, shallow, rocky and low in organic matter, moisture content and water holding capacities which reduce infiltration rates thereby increasing water run-off and soil erosion (Sombroek *et al.*, 1982).

Pastoral and nomadic communities inhabit the ASALs and their livelihood greatly depends on livestock and small-scale agriculture. These communities are often subjected to various shocks and disruptions of drought, famine, lack of fodder, physical insecurity fuelled by ethnic conflict through livestock rustling, banditry, wildlife conflict, and pests and diseases, resulting in great loss of livestock (Barrow, 1996). The high current movement of people from high and medium potential areas to the fragile ASAL areas in search of settlement has led to widespread over-exploitation of plant resources (Herlocker, 1999). The adverse effects of deforestation include loss of soil structure, decrease in soil fertility that leads to reduced crop yields, soil erosion and low livestock production (Sanchez, 1976; Brady and Weil, 1999). As a result, the ASAL communities remain vulnerable to scarcity of resources and their future survival is uncertain (Barrow, 1996; Herlocker, 1999).

ASAL ecosystem is endowed with natural vegetation mainly of Acacia woodlands, thorny trees, shrubs and bushy grasslands that are being over exploited for fuel wood, charcoal, medicines and building poles, food and fodder. This has led to loss of biodiversity resources, serious land degradation and negative impact on gum producing trees. Among the most exploited tree species, is the *Acacia senegal*, a multipurpose agro forestry tree belonging to (subfamily *Mimosoideae*, family *Leguminosae*), whose product is the gum arabic exudate, an article of commerce internationally (Green Way *et al.*, 1961).

In Kenya, *Acacia senegal* and other closely related species (*A. tortilis, A. elatior, A. nilotica, A. mellifera* and *A. seyal* var. *fistula, Commiphora* and *Boswellia* species) are found in abundance in the dry land areas of northern districts of Rift Valley and Eastern provinces (Chikamai and Gachathi, 1994). *Acacia senegal* is an important multipurpose tree for gum arabic production in the Sudan gum gardens (Bekele-Tesemma *et al.*, 1993). The tree prefers to grow at altitudes of 100 - 1700 metres above sea level (A.S.L) in coarse texture soils, sandy loams, slightly loamy sands and clay loam soils with annual rainfall mainly between 300- 400 mm. *Acacia senegal* is a drought resistant tree which can tolerate high daily temperatures over 45^oC and plays key role in rehabilitation of degraded areas, restores soil stabilization and nutrient holding capacity and improves the turnover of soil organic matter pools (Bekele-Tesemma *et al.*, 1993). Gum arabic yielding trees are potential species suitable for afforestation of ASAL areas in combating the process of desertification, land degradation and restoring soil fertility.

Gum arabic exudate is a pale orange brown coloured natural product obtained from the stems of *Acacia senegal* var. *senegal* and *A. seyal* var. *fistula* and other closely related Acacia species. Gum arabic is used as a binder, emulsifier, stabilizer and protective agent in food, pharmaceutical and technical industries (FAO, 1995). Gum arabic is also edible and in addition, the tree is an important fodder species as the pods and leaves are palatable to livestock especially goats (Kokwaro, 1976).

The Joint Expert Committee on Food Additives (FAO/WHO) international revised specifications states that the quality parameters of gum arabic must conform to the characteristics of the Sudan gum from Kordofan gum belt region. The specifications of quality of gum arabic must be obtained from the branches and stems of *Acacia senegal* var. *senegal*, with moisture content of 13 to 15 % at 105^oC, total ash of less than 4% after ashing at 550^oC, specific rotation of -26° to -34° , and nitrogen content of 0.26 to 0.39 %. The average cationic composition of total ash is usually 52 ppm Cu, 730 ppm Fe, 69 ppm Mn and 45 ppm Zn (FAO, 1990).

Gum arabic is produced in the dry land districts of northern Kenya, namely; Isiolo, Samburu, Marsabit, Wajir, Mandera and Turkana. In some areas pure stands of *Acacia Senegal* and other closely related Acacia species form extensive vegetations covering several square kilometers (Chikamai and Gachathi, 1994). Production of gum arabic has increased gradually with time from a few tonnes in 1990 to 400 tonnes in 1994 (Chikamai, 1997). Gum arabic is gathered at random by uncoordinated group of pastoralists from regenerated natural stands of *Acacia senegal* and closely related species, over wider areas in different locations at different seasons. The harvested gums are mixed and sold to middle businessmen, usually operating other kinds of trades at local trading centres. These merchants export the gums without standard quality control to world market.

Kenya has emerged as a new supplier of gum arabic in the world market, but the country does not meet the competitiveness and adequate supply of the commodity to the world market (FAO, 1995). The Kenyan gum is of low quality when compared with the Sudan gum in the world market according to JECFA specifications (FAO, 1990). The quality of Kenyan gum has not been investigated adequately to allow its improvement. High quality gum will help to improve production of better-priced gums locally and internationally. It is anticipated that the quality of gum arabic from regenerated natural stands of *Acacia senegal* in a particular locality may be influenced by differences from one variety to the other, variation in soil types and compositions, specific habitats, climate, altitude and social factors. It can be speculated that micronutrient uptake may be limiting in some varieties of *Acacia senegal* and the seasonal factors influence the quality of gums. It is imperative therefore to postulate that if quality of gum is improved the ASAL communities would benefit from higher prices and this could be another source of income to supplement dependence on livestock production.

Gum arabic resources if efficiently, effectively and sustainably exploited in the ASAL areas may become the best alternative source of livelihoods for millions of ASAL people for a long time to come. These resources may be the benchmark of rural industrial revolution for the future of ASAL people. This can assist the ASAL communities in diversifying unsustainable farming practices to sustainable income generating activities where climatic conditions are not favourable for rainfed agriculture. It is hoped that the findings and recommendations of this project may be used to encourage farmers in the ASAL areas to plant, maintain, protect, sustain and manage *Acacia senegal* varieties and other closely related species in their farming systems for high quality gum arabic production.

1.1 Rationale

Efforts to establish tree planting in the ASAL environments are often difficult due to variations in climatic and edaphic factors. The major challenge is that the pastoral and nomadic communities find no immediate reward in tree planting in arid and semi-arid lands. To reduce and reverse the trends of the alarming rate of deforestation and associated problems of environmental degradation, afforestation programmes are being initiated to promote tree planting of more indigenous species to increase stand density of the vegetation cover. Appropriate technologies for raising mass production of *Acacia senegal* seedlings deserve a priority in dry land afforestation programmes (Darkoh, 1991).

For successful afforestation programmes in the ASAL, research areas should include studies on soil and plant nutrition, inter-relationship between and among the macronutrients (e.g. nitrogen (N), phosphorus (P), pofassium (K) and organic carbon (C), as well as micronutrients (e.g. copper (Cu), iron (Fe), manganese (Mn) and zinc (Zn) uptake by tree species. This would supplement the work that has been done on effect of irrigation on soil nutrient status in a forestry plantation at Bura Irrigation Scheme (Mwendwa *et al.*, 1993).

Macronutrients are needed in large quantities while micronutrients are taken in small (trace) quantities by plants. The functions of macro and micronutrients vary but they are very important to plant growth as they facilitate the catalytic enzyme reactions or stimulants in regulating the various physico-chemical processes in plant metabolism. Each element performs definite functions within the plant and no single element can be completely substituted for another. Ideally, all elements are needed in their normal concentrations (i.e., "balanced nutrients level") to play their role in plant growth. Any other quantity may result in deficiency and/or toxicity of that particular element by manifesting visible symptoms such as growth retardation, foliage colour and other irregular plant metabolic functions.

Mineral nutrition of indigenous trees in these fragile ecosystems, especially trace metals such as Cu, Fe, and Mn etc. uptake has not been studied. Presently, there is limited knowledge on the mineral nutrient levels in the soils of ASAL and their uptake by the *Acacia* species. It is postulated that in certain tree species, trace metals may enhance moisture retention, by limiting the rate of trans evaporation in trees growing in the ASAL areas. These are important parameters that need evaluation, since trees require supplies of certain chemical elements from the soils in the form of ions for growth and development (Evans, 1992).

1.2 Problem Statement

Gum arabic is collected from natural stands of *Acacia senegal* and other closely related species by pastoralists involved in herding activities during the dry seasons in northern Kenya. Gums are faced with several constraints such as inadequate seeds for tree propagation technologies for raising mass production of tree nursery seedlings, potting media, quantity, quality, agronomic studies, management, demand, lack of standard harvesting methods and processing, poor prices and marketing. Studies on site characterization of factors such as site-to-site variations, soil types, humidity, temperature, altitude, physiological state of Acacia gum trees and seasonal difference in relation to gum arabic quality and quantity per tree from a specific variety need to be investigated. These factors influence quality of gum arabic. This study therefore attempts to establish the effect of micronutrients uptake by *Acacia senegal*

varieties on gum arabic quality, levels of trace elements uptake by *Acacia senegal* var. *senegal* and *A. senegal* var. *kerensis* seedlings in glass house pot experiment, and factors that may influence the quality of gum arabic production under the ASAL conditions of Marigat division, Baringo district.

1.3 Objectives

The overall objective of this project was to establish the effect of micronutrients uptake by *Acacia senegal* varieties on gum arabic quality under arid and semi - arid conditions.

The specific objectives of this study were to:

- 1. Determine soil characteristics and fertility in the study area.
- 2. Quantify uptake of copper (Cu), iron (Fe), manganese (Mn) and zinc (Zn) by *Acacia senegal* varieties using; pot experiments in glasshouse simulating the arid and semi-arid conditions.
- 3. Evaluate the physiochemical properties of gum arabic between and within sites.
- 4. Elucidate effect of *Acacia senegal* varieties on the quality of gums produced in the study area.

1.4 Organisation of the Work presented

The description and discussion of the project undertaken in this thesis is preceded by review of literature in Chapter Two on the general overview of the physical and chemical properties of soils, and the functions of mineral nutrients in plants. In addition, the description, distribution and uses of gum arabic exudates and Acacia senegal trees are discussed. Harvesting and marketing of gum arabic and status of production in Kenya is reviewed. Chapter Three deals with materials and the experimental methods used throughout the study; including the detailed description of study area, selection and description of sites, soil sampling and pretreatment, and the methods of soil analysis used. Studies on micronutrient uptake by Acacia senegal varieties under glasshouse conditions are preceded by the experimental design used, preparations and applications of nutrient solutions to the potted plants. Data collection, plant sampling and tissue analysis, gum arabic harvesting and analysis and statistical methods of data analysis are also given. Chapter Four describes the results of soil characteristics and fertility status, while Chapter Five reports on studies on micronutrient uptake under the simulated arid and semi-arid conditions. Chapter Six discusses the effect of Acacia senegal variety differences in quality of gum arabic, followed by an Integrating Discussion in Chapter Seven, which uses the results of all previous Chapters to address the objectives of the project. The way forward for the establishment of the quality control of Kenyan gum and recommendations for further research are also given.

CHAPTER TWO

LITERATURE REVIEW

2.1 Physical properties of Soil

The physical properties of soils are texture, bulk density, particle density, porosity; percolation rate and moisture content which have close inter-relationship with the chemical properties (Lal and Greenland, 1979).

Soil Texture

Soil texture estimates the percentage sand, silt and clay contents of the soil and is often based on the proportions of different particle sizes bound into aggregates that require dispersion of the soil into individual particles as sand (2.00-0.05 mm), silt (0.05-0.002 mm) and clay (<0.002 mm) fractions (Page, 1982). Soil texture determines soil workability, water-holding capacity, soil structure and nutrient retention. Clay soils have micro pores that hold more water and retain nutrients than sandy soils and are easily prone to erosion that may lead to loss of soil productivity. Sandy soils have macro pores that retain less water and nutrients and may have low soil organic matter (Landon, 1991).

Bulk Density

Bulk density is used to identify the problems of root penetration, soil aeration in different soil horizons and soil texture of uncultivated and cultivated soils (Landon, 1991). Bulk density values vary considerably with moisture content, and determined by porosity of the given soil. Bulk densities of soils influence soil organic matter levels, and determine the ^{supply} of nutrients for growth of plants. They also play important role in storage of water for root uptake (Lal and Greenland, 1979). For instance, soils with high organic matter have lower bulk densities than soils low in organic matter.

Particle Density

Particle density is used together with the bulk density and water content to calculate the airspace and percentage of water saturation. Particle density is the ratio of the total mass of a given soil to its volume and varies between narrow limits of 2.60 and 2.75 g/cm³ per 100 cm³ (Blake, 1965).

Porosity

Total porosity is the ratio of pore empty space within a soil to the bulk volume (V_b) the soil sample (Lal and Greenland, 1979). Porosity influences aeration, water movement and root penetration in soils. Compacted soils reduce total pore space; hence increases bulk density which may hinder moisture availability to crops.

Percolation

Percolation is the downward movement of water that will drain freely from the soil in response to the force of gravity (Brown, 1987). When all the available pore spaces in the upper part of the soil are filled with water the soil is saturated and the rate of percolation is slow (Lal and Greenland, 1979). The amount of water percolating through the soil profile is determined by the amount of water supplied and the total moisture conductivity of the lower horizon together with the amount of water retained by the profile at its field capacity (Landon, 1991).

Moisture content

Moisture content is the maximum water content that the soil will retain following free drainage after saturation (Brown, 1987). It influences nutrient uptake by plants and the recycling of available nutrients either upwards from the plant roots to the upper leaves or downwards from the leaves to the root system. Moisture content is influenced by many factors such as soil texture, bulk density, porosity and organic matter (Anderson and Ingram, 1993).

2.1.2 Chemical properties of soils

The chemical properties of soils include soil pH, electrical conductivity, organic carbon, nitrogen, phosphorus, exchangeable cations (Ca, K, Mg and Na) and micronutrients (B, Cu, Fe, Mn, Mo and Zn). The properties are useful for determining the soil fertility status, maintenance of soil productivity in terms of physical, chemical and biological functions, and estimating and evaluating the amount of nutrient amendments needed for optimum plant growth, Anderson and Ingram (1993).

Soil pH

Soil pH is a measure of the acidity (pH< 7.05) or alkalinity (pH> 7.05) status of soil solutions. Most plants grow best in the soil pH range 5.5-7.05. The availability of many essential nutrients needed for plant growth is often dependent on the pH of the soil solution, which is closely identical to that of the roots of plants growing in that soil (Parker and Walker, 1986).

Electrical conductivity

Electrical conductivity is the measure of salinity or the level of soluble salts in a soil solution (McBride *et al.*, 1990). The total salt content of a soil can therefore be estimated from this measurement since it reflects the extent to which the soil is suitable for plant growth (Lund *et al.*, 1999). Muhammed (1996) and Brady and Weil (1999) reported that conductivity increases with the salt content but varies with the nature of the ions in soils.

Organic Carbon

Soil organic carbon is a measure of the organic matter content in a soil (Jones, 1991). The organic matter determines the soil fertility status following the decomposition process which leads to release of nutrients to the soil. Soil organic matter functions to promote aeration and soil moisture regimes (Nelson and Sommers, 1996).

Nitrogen

The organic-N fraction is a measure of the soil reserve of N or its capacity to release N required by plants through the process of mineralization (McGill and Figueiredo, 1993). Plants use forms of nitrogen as NH_4^+ , NO_3^- and NO_2^- ions through nitrification process. Thus, methods of N analysis for most soils must take into account the various fractions if present. The Kjeldahl procedure gives a good estimate of total soil N content.

Phosphorus

Phosphorus (P) is a major nutrient in soils that occurs in both inorganic and organic forms and ^{is} mainly taken by plants in form of orthophosphate anions (Bolland, 1992). Inorganic phosphorus exists as the phosphates of metals held by positive charges of clay soil particles in soil solutions (Day and Parker, 1985). Enzymes must first mineralise organic phosphorus before being taken by plants. Since P compounds in soils are highly variable and are related to soil type or parent material, many extraction techniques for available phosphorus are used worldwide for evaluating soil fertility (Brown *et al.*, 1977). Using the method of Olsen *et al.* (1954), the soil is extracted with 0.5 M solution of sodium bicarbonate at pH 8.5 prior to measurement. This procedure is generally accepted as a suitable index of P "availability" a wide range of soil types, especially those with pH above 7.05. In the original procedure carbon black was added to the extraction reagent to eliminate the color (because of soil organic matter) in the extract. The use of carbon black has since been eliminated in subsequent modification by Murphy and Riley, (1962), Watanabe and Olsen, (1965), and Olsen and Sommers, (1982) where, a single solution reagent containing ammonium molybdate, ascorbic acid and a small amount of antimony is used for color development (Stevenson, 1998).

Exchangeable cations in Soils

Exchangeable cations are nutrients ions released from the soil by solutions of neutral salts (Bohn *et al.*, 1985). Inorganic and organic colloids (clay particles and soil organic matter, respectively) are negatively charged and consequently can attract and bind cations such as potassium (K^+), sodium (Na^+), calcium (Ca^{2+}), and magnesium (Mg^{2+}) (Bohn *et al.*, 1985). The availability of exchangeable cations in the soils depends on soil texture and organic matter and leaching effects. Thus, nutrients can be held in the soil and not lost through leaching, and can subsequently be released for plant uptake (Bremner and Mulvaney, 1982).

Micronutrients in Soils

Micronutrients play important metabolic roles in plants, particularly in enzyme systems associated with the synthesis. Solubility of micronutrient cations decreases with an increase

in soil pH and lead to nutritional deficiencies (Mascagni and Cox, 1985). The distribution of micronutrients is used for characterization of soil types and soil fertility. Soil test values serve to predict deficient conditions of soils and to indicate how much is required to improve a given situation.

2.1.3 Functions of Mineral Elements in Plants

2.1.3.1 Macronutrients

(a) Nitrogen

Nitrogen is the most abundant element in plants and plays an essential role as a constituent of proteins, nucleic acids, chlorophyll, vitamins, growth hormones, and ligands for metal complexation in heterocyclic ring of metalloprotein complexes and uptake of other nutrients (Reuter and Robinson 1997). The synthesis of various amino acids and amides needed to make proteins depends on the availability of nitrogen and carbon compounds in the leaves. Nitrogen requirement for optimal growth in plants is in the range between 2 and 5% of the plant dry weight (Stevenson, 1986). Plants suffering from nitrogen deficiency exhibit symptoms such as retarded growth, yellow or pale green leaves with small cells and thick walls, inhibition of root elongation and delay of flowering (Spiers, 1978).

(b) Phosphorus

Phosphorus as phosphate is a constituent of nucleic acids, nucleoproteins and phospholipids, including those of cytoplasmic membranes (Miller and Donahue, 1992). Phosphorus is responsible for strongly acidic nature of nucleic acids (DNA), the carrier of genetic information and ribonucleic acids (RNA), which carries genetic messages from DNA to transport specific amino acids to the ribosomes where they joint together to make proteins (Marschner, 1986). It takes an active part in the synthesis of organic matter and formation of various plant tissues and in the transfer mechanism of genetic characteristics in plants (Munson and Nelson, 1990). Phosphate and other nutrient ions are essential for cell division and for the development of meristem tissue (Wild, 1988). Phosphorus requirement for optimal growth in plants is in the range of 0.3 to 0.5 % of dry weight during the vegetative stages of growth (McKeague, 1978). Plants suffering from phosphorus deficiency show retarded growth, stunted root system, and dying of leaves (Martin and Matocha, 1973). Other metabolic processes include; dull greyish - green leaf colour, red pigmentation as a result of accumulation of anthocyanins during respiration and photosynthesis (Terry and Ulrich, 1973). Phosphorus toxicity in plants has not been reported (Marschner, 1986).

(c) Potassium

Potassium is required in large amounts by plants (Marschner, 1986). It is the most abundant cellular cation and is involved in transport of acidic metabolites, neutralization of anionic groups of macromolecules, activation of many enzymes and the control of cellular osmotic potential (Kresge and Younts, 1962). Potassium also plays a specific role in the mechanism of opening and closing of stomata; it accumulates, with malate, in the guard cells when stomata are open and is released when they are closed, thereby contributing to the osmotic potential on which the turgor of the guard cells depends (Wild, 1988). Potassium is also involved in the transport of photosynthetic products from the leaves and as a result it can have an indirect effect on photosynthesis (Spiers, 1983).

Potassium can be substituted to some degree by sodium in its physiological roles, particularly in the maintenance of cell turgor (Tandon, 1991 and 1993). Nicholaides (1985) reported the symptoms of potassium deficiency are often seen when there is premature death of the older leaves, because of its withdrawal to the young leaves. When nitrogen and potassium are simultaneously in short supply, the plants are stunted; their leaves are small and rather grey in colour and die prematurely, first at the tips and then along the outer edges. Fruits and seeds of K_{-} deficient plants are usually small in quantity and size (Tisdale *et al.*, 1985).

(d) Calcium

Calcium is essential for the growth of meristems, and root tips (Loneragan and Snowball, 1969). It plays a key role in the maintenance of cell membranes, protecting root tips against leakiness due to ion imbalance, low pH and presence of toxic ions such as aluminium (Brown, 1987). A few enzymes, including amylase and some nucleases require calcium (Wild, 1988). It is the main cation associated with the middle lamella of cell walls. In seeds, it is present as the salt of phytic acid (Brown *et al.*, 1977). Calcium deficiency causes the growth of the root system to be stunted, and leaf margins to curl backwards (Loneragan *et al.*, 1968). Degeneration at the apex of the young fruit (blossom end rot) is a common symptom of deficiency in tomatoes.

(e) Magnesium

Magnesium is required in large amounts by plants (Wild, 1988). It plays a major role as a central metal ligand in chlorophyll (magnesium porphyrin) in which one atom of magnesium is bound to four pyrrole rings (Brown, 1987). As a cofactor of most of the enzymes that act on phosphorylated substrates, it is of great importance in the transfer of energy (Martens and Lindsay, 1990). It also activates certain other enzymes, including carboxylases and some dehydrogenases (Tekalign *et al.*, 1991). Because of its role as a cofactor of enzymes that act on phosphorylated substrates, the distribution of magnesium in plants is often similar to that of phosphorus (Hewitt, 1983). The symptoms of magnesium deficiency vary among plant

species, but the first symptom is always interveinal yellowing, or chlorosis, because of diminished levels of chlorophyll (Westerman, 1990). This is accompanied by a decreased rate of photosynthesis. Biosynthetic pathways are disrupted as a result of inhibition of phosphorylation processes and soluble nitrogenous compounds accumulate with concomitant decreases in the proportion of proteins (Stevenson, 1986). In acidic and sandy soils, high levels of potassium or ammonium ions in the root medium similarly inhibit the uptake of magnesium.

(f) Sulphur

Sulphur is absorbed from the soil solution as sulphate ions (Stevenson, 1986), and can also be assimilated from the sulphur dioxide present in the atmosphere (Allaway, 1992). A large proportion of sulphate is reduced to sulphide, which is then incorporated in the amino acids such as cysteine, cystine and methionine (Verma, 1977). Normally these amino acids are present in the free state in very small quantities as they are rapidly converted to proteins (Stevenson, 1986). The sulphide is also present in some coenzymes, including biotin, thiamine and coenzyme A, which are essential for metabolism when attached to appropriate The ferrodoxins, which are non-haem proteins involved in apoenzymes (Wild, 1988). photosynthesis and other electron transfer process, contain sulphide and iron in equivalent amounts (Anderson and Evans, 1956). When the supply of sulphur is ample, the rate of uptake may exceed the rates of reduction and assimilation into proteins and other organic compounds and this leads to accumulation of sulphates in the plant tissues (Stevenson, 1986). Sulphur is essential for the synthesis of the sulphur-containing amino acids and proteins including enzymes (Williams and Steinbergs, 1959). Sulphur deficiency disrupts many biochemical reactions in plants leading to increased levels of amines, amides and since these substrates are not utilized for the synthesis of proteins. The level of carbohydrates is also reduced as a result of diminished photosynthesis (Johnson and Fixen, 1990). Sulphur- deficient plants are stunted and weak with thin stems while the leaves remain small and distorted. A deficiency of either sulphur or nitrogen causes chlorosis, or yellowing, of the leaves as a result of diminished levels of chlorophyll (Wild, 1988).

(g) Sodium

Plants to a varying degree depending on the species absorb sodium (Spiers, 1983). It is not known to be essential except for some salt tolerant *Atripelex* species. Sodium can, however, sometimes have beneficial effects on plant growth (Landon, 1991). It can particularly replace potassium in osmotic roles and the maintenance of cell turgor and this effect is large when the supply of potassium is inadequate (Wild, 1988).

2.1.3.2 Micronutrients (Trace elements)

(a) Copper

Copper occurs in two oxidation states Cu^+ and Cu^{2+} ions. The available species in soils is the Cu^{2+} (Wild, 1988; Reuter and Robinson 1997). Copper is an essential trace element absorbed by plant roots and translocated to the shoots predominantly in ionic form (Marschner, 1986). Copper is a constituent of a prosthetic group of oxidases in which molecular oxygen is used directly in the oxidation of substrate. These include cytochrome oxidase, phenol oxidase, laccase, ascorbic acid oxidase and amine oxidase (Marschner, 1986; Martens and Lindsay, 1990).

The catalytic activity of these enzymes depends on the ability of their copper to undergo reversible change from Cu⁺ to Cu²⁺ (Whitney, 1988). This is an oxidation-reduction process. In addition to the oxidases which cataylse the reduction of molecular oxygen, plants have superoxide dismutase, an enzyme containing copper and zinc, which reacts with superoxide ions (O_2) to produce molecular oxygen and hydrogen peroxide (Stevenson, 1986). As the superoxide ion is toxic and readily produced from molecular oxygen, superoxide dismutase plays an important, protective role in plant growth and development. Copper also plays a role in photosynthesis as an essential constituent of plastocyanin. This protein is located in the chloroplasts and forms part of oxidative phosphorylation process that mediates the electron transport chain between the two photochemical systems of photosynthesis (Stevenson, 1986).

In cytochrome oxidase, copper acts as the terminal recipient of electrons from mitochondrial electron transport chain prior to transfer to oxygen (Reuter and Robinson 1997). The activity of the cytochrome oxidase is attributed to reversible change in oxidation state in photosynthesis (Wahle and Davies, 1977). Copper also plays a role in the catalytic activity of oxidation reactions of plant phenols in two distinct enzymes: phenolase and laccase. Phenolase catalyzes oxidation of monophenols to diphenols such as dihydroxyphenyl-alanine (Dopa) and also involved in the synthesis of lignin and alkaloids which protect wounded tissues from fungal attack. Laccase is found in thylakoid membrane of chloroplast and is required for the synthesis of plastoquinone, a constituent of the photosynthetic electron transport chain (Martens and Lindsay, 1990).

In ascorbic acid oxidase, copper catalyses the oxidation of ascorbic acid to L- dehydroascorbic acid. The enzyme occurs in the cell wall and cytoplasm (Reuter and Robinson 1997). In amine oxidase, copper catalyses oxidative deamination of polyamines which occur abundantly in legumes during the seedling stage (Walker and Webb, 1981). The normal concentrations of copper in soils and plants which are considered as adequate, range between 10 to 18 mg kg⁻¹ and 7 to 30 mg kg⁻¹ Cu, respectively (Mitchell, 1964).

The concentration depends on the type of soil and plant species and may fall outside these ranges. When the Cu concentration in plants is less than 3 ppm in the dry matter, deficiency symptoms are likely to occur (Jones, 1972). These symptoms often depend on plant species or variety, and the stage of growth (Graham *et al.*, 1981). Reuter *et al.* (1981) reported that copper deficiency was responsible for delay in flowering and formation of fruits of *Chrysanthemums*, possibly because of induced phenolase activity and consequent interference in auxin oxidase system. Copper deficiency has no effect on chloroplasts of spinach or maize but affects the chloroplasts of oats even when the white tip symptom is evident (De Kock *et al.*, 1979). In general, many plant species display chlorosis, necrosis, leaf distortion and terminal dieback in young shoot tissues (Lindsay and Norvell, 1978).

Copper toxicity is reported to be rare in field grown plants, except in areas where high copper containing pesticides have been applied. When Cu levels exceed 20 ppm in mature leaves, toxicities may occur (Reuter and Robinson 1997). The common symptoms of copper toxicity in many plants are: reduced shoot vigour, poorly developed and discoloured root systems, and leaf chlorosis. They develop in a manner regarded as characteristic for phloem immobile nutrients (Loneragan, 1968); affecting those meristems which are active when external copper supply is high.

(b) Iron

The functions of iron in plants include chlorophyll formation and activation of some enzymes (Whitney, 1988). Iron is taken by plant roots as Fe^{3+} and plant roots first reduce it to Fe^{2+} , which is then translocated to the shoots where it is again oxidized back to Fe^{3+} (Tiffin, 1970; Tisdale, 1985). Some of the iron can be stored in the leaves as phyto-ferritin, which is essential for developing plastids during the process of photosynthesis (Hewitt, 1983).
The active iron occurs in numerous enzyme systems either as a structural component of prosthetic groups or as a constituent of the protein itself (Wallace, 1980). The best-known prosthetic groups are the iron porphyrins which, when attached to specific proteins, are known as haem proteins (Rains, 1976). These include peroxidase, catalase and some dehydrogenase enzymes, as well as cytochromes that function as electron carriers during photosynthesis and The reduction of nitrate to ammonia requires iron, since the enzyme nitrite respiration. reductase itself comprises a haem protein, called sirohaem and a non-haem component containing iron and sulphur (Anderson and Evans, 1956). Leghaemoglobin, another haemprotein, is required for the fixation of dinitrogen in legume root nodules. Through reversible oxygenation, it regulates the supply of oxygen to nitrogenase that is highly sensitive. The most important non-haem protein is ferredoxin that has a very high negative redox potential and acts as an electron carrier in photophosphorylation, in the photosynthetic reduction of nicotinamide adenine dinucleotide phosphate (NADP), and in the reduction of nitrite. In legume root nodules and other biological nitrogen fixing systems, either ferredin or flavodoxin acts as an electron carrier to nitrogenase, which comprises a non-haem protein with molybdenum as an essential constituent and an iron-sulphur protein (Rains, 1976).

The normal concentrations of iron in soils and plants which are considered as adequate range between 1 and 10% and 25 to 500 mg kg⁻¹ Fe, respectively, (Mitchell, 1964). Fe is an immobile element in plants and the Fe content in plants can vary considerably with a critical level of 20 ppm in grasses and corn (Jones *et al.*, 1991). Fe deficiency is likely to occur when the concentration in leaves is 50 ppm or less and very difficult to correct in some crops (Ponnamperuma *et al.*, 1981). Foliar applications of Fe have been found to be effective in correcting Fe deficiencies in plants such as turf grasses (Mitchell, 1964).

(c) Manganese

Manganese is absorbed in its biologically active form of Mn^{2+} by an energy-linked process that is negatively influenced by the presence of Mg^{2+} and Ca^{2+} (Peverill *et al.*, 1999). Manganese, like magnesium, can act as a cofactor of many enzymes that act on phosphorylated substrates (Mascagni and Cox, 1985). Some of the enzymes in the Kreb's cycle, notably decarboxylases and dehydrogenases, are also activated by manganese, although it can be substituted by magnesium.

Manganese is present in the chloroplasts in a complex that oxidizes water to produce molecular oxygen, hydrogen ions and electrons (photolysis) in photosynthesis. The role of manganese in the photosynthetic evolution of oxygen is reflected in gross changes in structure of the chloroplasts in manganese deficient plants. Manganese also plays a role in regulating the levels of auxin in plant tissues through activation of the auxin oxidase system. Manganese deficiency in plants varies considerably depending on soil types and plant species (Miller and Donahue, 1992).

Manganese differs from other micronutrients in that both deficiency and toxicity are widespread in agricultural practice. Manganese deficiency normally occurs when the leaf tissue concentration is less than 15 ppm. Plants which are sensitive to Mn deficiency are equally sensitive to excessive Mn. Growth of soybeans, which are particularly sensitive to Mn deficiency, is reduced when leaf Mn levels approach 200 ppm (Ohki, 1976).

Morgan *et al.* (1966) found that Mn toxicity is associated with the destruction of indole acetic acid (IAA). Meudt (1971) reported that Mn enhances the sulphate-induced chain reaction during the enzyme auto-oxidation of IAA. Manganese ions are believed to mediate in electron

transfer between sulphite ions and IAA free radicals. Symptoms of manganese toxicity are likely to occur in acid soils (as a result of the reduction of manganese oxides). Induced iron deficiency is another well known symptom (called "Crinkly leaf") induced by manganese toxicity in dicotyledons such as cotton and beans (Foy *et al.*, 1981). Excessive uptake of manganese in acid soils of East Africa has been established to be a principal problem (Chamberlain and Seale, 1963).

(d) Zinc

Zinc is absorbed by plants as Zn^{2+} and translocated to the shoots primarily as the free ion. Zinc is known to be an essential constituent of a number of plant enzymes such as carbonic anhydrase, alcohol dehydrogenase, auxin oxidase, superoxide dismutase, DNA and RNA polymerases, peptidases, proteinases and cytochromes (Stevenson, 1986). All these enzymes play varied, but significant roles (Whitney, 1988). The function of carbonic anhydrase is to catalyze the hydration of CO_2 in plants during photosynthesis. The catalytic action of the enzyme is both in the cytoplasm and the chloroplasts. Reuter and Robinson (1997) reported that alcohol dehydrogenase catalyses the reduction of acetaldehyde to ethanol in meristematic zones of roots in higher plants. Superoxide dismutase (a zinc-copper containing enzyme) acts in the reaction mechanisms involving the detoxification of molecular oxygen during the reduction of superoxide ion in the chloroplasts (Marschner, 1986). Zinc is required for the synthesis of tryptophan, a precursor for the synthesis of I.A.A. in plants.

Zinc deficiency produces leaf malformation, and is often characterised by irregular mottling, with yellow ivory interveinal areas. Extreme resetting of terminal and lateral shoots in woody species and multiple branching are also a common feature. Mitchell (1964)

reported that the normal concentrations of zinc in soils and plants considered as adequate are in the range of 10 to 300 and 21 to 70 mg kg⁻¹ zinc, respectively.

Stunting is a frequent symptom associated with Zn deficiency (Boyle and Smith, 1985). Low levels of zinc that result in deficiencies have been reported in coffee, sisal leaves and tea in various parts of East Africa and in soils of some parts of Kenya (Chamberlain, 1961). The critical Zn value for apple is about 14 ppm with the first symptom of the deficiency being small fruit size. Zinc deficiency in pecans occurs when the Zn leaf level is 30 ppm or less (Whitney, 1988).

Zinc interacts with phosphorus when large amounts of phosphorus fertilizers are applied to soils low in zinc (Martens and Lindsay, 1990; McBride, 1994). The excess phosphorus enhances zinc adsorption, thereby maintaining the availability of zinc (Jones *et al.*, 1991). The phosphorus-zinc interactions in plants minimize the inhibition of zinc uptake and zinc translocation from the roots to the shoots by calcium cations (Loneragan, 1968).

Zinc is relatively non-toxic to many crops and a concentration of 500 mg kg⁻¹ Zn in soil is often needed before zinc toxicity is observed (Jones, 1972; Walsh and Beaton, 1973). Zinc toxicity can readily induce iron deficiency, resulting in chlorosis in young leaves as well as inhibition of root elongation. This may be as a result of the isoelectric nature of the two hydrated Zn^{2*} and Fe^{2+} ions (Woolhouse, 1983).

(e) Boron

Boron (B) is absorbed by plants as undissociated boric acid and follows the flow of water into the roots (Bell, 1997). Boron requirements vary considerably among crops and certain plant species show toxicity symptoms if the level exceeds 200 ppm B (Watson and Brown, 1998). The optimum range in leaf tissue of most crops is 20 to 100 ppm. Some crops are particularly sensitive to B and can be injured when the B level in the leaf is too high. For example, B levels in excess of 50 ppm have been associated with B toxicity in peaches (Johnson and Fixen, 1990). The B critical level for corn is about 4 ppm, while lucerne, cotton, peanut, and soybeans have critical levels of 20 ppm.

Boron in plants is not a very mobile element because uptake from the roots to the shoot is confined to the xylem. Uptake and translocation are closely related not only to the mass flow of water to the root surface but also to the xylem water flow (Gaines and Mitchell, 1979). It acts as a substrate in carbohydrate metabolism for the synthesis of phenolic acids and hemicelulose (cell wall material). Boron is involved in transport of sugars and also regulates the flux of substrate in lignin biosynthesis via the formation of stable phenolic borate complexes, particularly with caffeic acid, and maintains IAA oxidase activities during cell division and root elongation (Hewitt, 1983).

Dicotyledons are more susceptible to boron deficiency than monocotyledons, with the exception of maize and sorghum (Keren and Bingham, 1985). Boron deficiency symptoms in plants occur in the newly emerging tissues and show disorganized meristems leading to early death of stem tips (Shuman *et al.*, 1992). Leaves may become crinkled and misshapen while stems and petioles thicken and crack. Flowering may be totally suppressed while fruit and seed formation, if it occurs, is abnormal (Gupta, 1993). A very rapid effect of boron deficiency is a reduction in the level of RNA and cessation of cell division in root tip meristems, which make the roots become shortened and appear bumpy. The symptoms of boron toxicity on mature leaves are marginal and/or tip chlorosis and necrosis. Corn whose B

requirement is low, is also sensitive to excess B. Toxicities may occur when the B level in young corn leaf tissue exceeds 25 ppm (Watson and Brown, 1998). On young leaves there is vellowing of margins, crumpling, blackening and distortion (Rashid *et al.*, 1997).

(f) Molybdenum

Molybdenum (Mo) is absorbed from soils predominantly as the MoO₄²⁻ ion (Wild, 1988). The content of molybdenum in plants that is considered to be adequate for healthy growth is only 0.1 ppm Mo dry weight. Molybdenum is an essential constituent of two important enzymes involved in biological nitrogen fixation; namely nitrate reductase and nitrogenase (Reuter and Robinson, 1997). Nitrate reductase is a molybdoflavo-protein and is present in the cytoplasm of leaves and roots (Hewitt, 1983). Molybdenum is specifically required for the biological fixation of dinitrogen both in the symbiotic systems of legumes and certain non-legumes, and in free-living organisms (Munson and Nelson, 1990). Nitrogenase, the enzyme responsible for the reduction of dinitrogen, consists of two iron-sulphur proteins; one contains molybdenum while the other contains iron (Martens and Lindsay, 1990).

Molybdenum requirement of legumes is higher than that of other plants since Mo is essential for the fixation of atmospheric N by the symbiotic bacteria (Lowe and Massey, 1965). Molybdenum is essential for the conversion of nitrate to ammonium ions in plants. The symptoms of molybdenum deficiency include proliferation of small nodules on the roots, and reduction in levels of amino acids, amides and amines (Possingham and Brown, 1957). Molybdenum toxicity in young leaves causes malformation and golden yellow discoloration of the shoot tissues. In mature leaves it causes mottling over whole leaf but little pigmentation, cupping and distortion of stems. Molybdenum is quite toxic to animals if the forage being consumed contains more than 15 ppm Mo (Reuter and Robinson, 1997).

2.1.3.3 Soil Fertility Status under Arid and Semi-arid Conditions

Soil fertility may be defined in terms of soil productivity, as the capacity of a soil to supply essential nutrients in adequate and right proportions for optimum growth of plants (Sanchez et al., 1996). Soil degradation is the main cause of low soil fertility and leads to poor vegetation cover and low crop yields in dryland agriculture. According to Lal (2000), soil degradation in the ASALs is often caused by neglect and misuse of soil over a long period of time. Soils in most of the arid and semi-arid areas have low organic matter, low cation exchange capacity, low nutrient reserves and low plant-available water holding capacity. They are highly fragile and easily degraded. Land degradation in the ASALs is likely to intensify in the future due to increase in population density, intensive cultivation of marginal lands, as well as the more escalating arid conditions caused by the effects climate change. Soil fertility replenishment in ASALs is often not feasible because it is too costly, risky and impractical for many farmers as a result of high potential evaporation rates, high temperatures that do not encourage bacterial activity especially in acid soils and leaching effects (Lal, 1996). In addition, there are no adequate fertility replenishment strategies and resources to improve soil productivity for sustained irrigated agriculture (Pereira, 1993).

Water run-off removes considerable quantities of soil organic matter and mineral nutrients. This consequently destabilizes soil structure and decreases the infiltration rate, in addition to reducing the available soil moisture for plant growth (Lal, 1996). Nutrient uptake by plants is influenced by soil moisture, the principal factor limiting the growth of plants in dry land environment. Organic matter influences soil stability, proneness to erosion, the quality of soil structure, water holding capacity, and nutrient storage and turnover. Soil structure regulates movement of water retention and nutrient transformations, faunal activity, species diversity, strength and rigidity of the rooting media (Lal, 1996). A decline in soil organic

matter levels may cause an increase in proneness to compaction, lower infiltration rates, increase run-off and hence enhance the likelihood of erosion.

Sanchez *et al.* (1996) reported that loss of organic matter and soil structure stability leads to severe land degradation with adverse consequences on physical, chemical, biological and nutrient status of soil. Loss of nutrients is often recognized as an increasing problem particularly in sub-tropical and tropical countries (Smaling, 1990; Smaling *et al.*, 1997).

2.1.4 Description and Distribution of Acacia senegal varieties in Kenya

The Acacia senegal tree is a widespread woody species. It is extremely variable in its habitat, from a bush of about 2 m to a tree of 15 m tall, with a flat to rounded crown (Bentje, 1994). The tree has many branches and erect twigs with small, grey-green spreading within the upright part. The bark is yellow/brown and smooth on younger trees, changing to dark grey, gnarled and cracked on older trees. The branchlets have thorns just below the nodes: three thorns with the central one hooked downwards and laterals curved upwards and the median one curved backwards (Gachathi, 1994). The tree produces white or cream colored flowers during the rainy season. *Acacia senegal has* four different varieties namely: *Acacia senegal* var. *senegal*, *A. senegal* var. *kerensis* Schweinf. *A. senegal* var. *rostrata* Brenan, and *A. senegal* var. *leiorhachis* Brenan (Chikamai and Gachathi, 1994).

2.1.4.1 Description, Uses of Gum arabic exudates and Acacia senegal

Gum arabic is water soluble, exudes naturally or in response to wounding of the stems of the Acacia senegal varieties. It is colloidal and insoluble in alcohol and ether. It contains high amount of sugar and is closely allied to the pectins. The gum from other Acacia species (A seyal, etc.) is available commercially as gum tahla and marketed for technological applications (FAO, 1990). Gum arabic has been used for over 4,000 years. Egyptians used gums as early as 2,000 B.C. in food, preparations of human and veterinary medicines, in crafts, and as a cosmetic. Sudan gum has been an article of commerce since 100 A.D (Anderson, 1987).

Gum arabic is used as a stabilizer, emulsifier, thickener, binder and protective agent in the food, pharmaceutical, and technical industries (FAO, 1995). FAO specification for gum arabic intended for food use stipulates that it should come from *Acacia senegal* or closely related species (FAO, 1990). The exudate is suitable for food purposes having been evaluated toxicologically as a safe food additive (Anderson and Morrison, 1990). In Europe, the legislative requirement for the food additive of gum arabic is the quality performance in bakery products. This makes gum arabic the material of choice in food industry (FAO, 1990). As an emulsifier, gum arabic is used in confectionery products, particularly to prevent crystallisation of sugars. In candy products it is also used as a glaze, means of encapsulating flavours (for example, spray-dried flavours and citrus oils), and in a range of dairy and bakery products it is used as a glaze. It is used either as a vehicle for flavouring or as a stabilizer or clouding agent in soft and alcoholic drinks.

In pharmaceutical industry, gum arabic is used in tablet manufacture as a binding agent, as soothing and softening agent for cough syrups, and as an emulsifying agent in cosmetics such as creams and lotions and sometimes in combination with other gums (FAO, 1995).

Gum arabic is used in the manufacture of adhesives and ink, and as a binding medium to suspend the colouring matter. It is also used to increase the viscosity of ink, or to make it now well, to prevent it from feathering. It is also used as a protective agent to treat offset lithographic plates such as a protective coating to prevent oxidation in printing industry and

as a component of solutions to increase hydrophilicity and impart ink repellency to the plates, and as a base for photosensitive chemicals (FAO, 1995). Other technical uses include ceramics, where gum arabic helps to strengthen the clay, certain types of inks, and pyrotechnics. It is also used as a binding agent in textiles, paints and adhesives such as traditional office glue and postage stamps (FAO, 1995).

The tree is used in agro forestry systems in dry land farming for honeybee forage, fuel wood, charcoal making, fibres (ropes) and tool handles. It is a windbreaker and a nitrogen fixing tree, an important attribute in stabilizing soil structure and improving soil fertility (FAO, 1986; NAS, 1979).

2.1.4.2 Gum arabic Production, Harvesting and Marketing in Kenya

Gum arabic is collected by uncoordinated groups of women and herdsmen from pastoralist communities at different seasons over wider areas (Gachathi, 1994). The collected gum is marketed through middle businessmen usually operating other kinds of trades at local trading centres. These businessmen include hotelkeepers and shopkeepers who sell the same at a profit to local companies such as Semi arid lands for livestock (SALTLICK), Nalepo farmers, AfriGums, etc. for export to the world market. In the Sudan however, gum arabic is collected in an orderly manner and systematically from tapped *Acacia senegal* plantations. The gum is therefore uniform in most respects and is considered the best grade internationally (Gachathi, 1994). The grading and quality control yardstick for gum arabic on the international market is based on standards of the Sudan gum.

Gum arabic production in Kenya is faced with several problems and constraints. The major problems facing gum quality are inadequate tools and analytical procedures on timing of happing and harvesting, cleaning, sorting, handling, packaging and storage. These factors affect the grading of Kenyan gum arabic in the world market. When the Kenyan gum is tested in the world market, it is rated of low quality according to the strict regulations of IECFA specifications (FAO, 1990). There is inadequate information on data in terms of harvesting techniques, cleaning, sorting, handling, packaging, storage and transportation all of which affect quality of gum arabic. Acacia senegal varieties as well as seasonal factors may also influence quality of gum arabic. It is also possible that variation in soil type, humidity and rainfall or even vegetative or physiological state of the species influence the quality. It is imperative therefore to postulate that if quality of gum is improved the local communities would benefit from higher prices and this could be another source of income to supplement dependence on livestock production in the ASALs. There is need therefore to investigate factors that influence the quality and production of gum arabic and to determine variability in physicochemical characteristics of Kenyan gum arabic and their relationship with variables such as soil types, altitude and seasons. This would be in addition to establishing a Kenyan system quality control for grading of gum arabic based on physical and chemical properties of soils.

CHAPTER THREE

MATERIALS AND METHODS

3.1 The Study Area

3.1.1 Introduction

Baringo district is situated in the Rift Valley province of Kenya. It covers an area of 8,655 square kilometers. It lies between latitudes $0^{0}15'$ and $1^{0}45'$ N and longitudes $35^{0}30'$ and $36^{0}15'$ E.



Figure 3.1: Location of the Study Sites – Baringo District

It borders Turkana district to the north, Samburu and Laikipia districts to the east, Koibatek to the south, and Keiyo, Marakwet and West Pokot districts to the west (Figure 3.1). The district has three main agro ecological zones: lowlands, medium highlands and the highlands (GOK, 1993). The lowlands comprise the northern plateau, Lake Baringo and Kerio Valley basins. The medium highlands extend from Kimose to Mugurin while the highlands comprise the Tugen hills, which start from Tenges, Lembus forest, Kabarnet through Kabartonjo with an elevation of 1800 - 2300 metres above sea level.

The district experiences two seasons of rainfall. The long rains start from the end of March to July and short rains from September to November. The average annual rainfall varies from 1000-1500 mm in the highlands to less than 500 mm in the lowlands around Nginyang and Lake Baringo basin. The soils of the district east of Kerio river are predominantly saline, sodic calcareous, while those of Tugen hills are developed on undifferentiated volcanic rocks and ashes of older volcanoes, mainly of basic igneous rocks (GOK, 1993).

About 46% of Baringo district comprises the arid and semi-arid lands (ASALs), with low and erratic rainfall, high potential evaporation rates, high temperatures and poor soils (Sombroek *et al.*, 1982). This area is predominantly a lowland dry zone. The only activities are livestock, sorghum and millet farming.

Marigat division, one of the 12 divisions in the district, receives an average annual rainfall of less than 500 mm. Annual temperatures average 30^oC in dry seasons and 26^oC during wet seasons. The annual potential evaporation rate is about 2000 mm. The topography of the area comprises mainly plain, undulating, rolling and hilly features. The soils are

variable, shallow, stony and rocky with loose lava boulders of igneous rock rich in calcium carbonate (Sombroek et al., 1982).

The division is endowed with natural vegetation mainly of Acacia woodlands consisting of *Acacia mellifera, A. tortilis, A. senegal* var. *kerensis, A. senegal* var. *senegal, A. nilotica,* A. *reficiens, A. seyal* var. *fistula* and A. *hockii.* It is the dominant Acacia vegetation that made Marigat division a choice study site for gum quality investigations. The division is faced with serious land degradation problems caused by deforestation of trees and shrub species capable of producing various essential products, such as gums, tannins, herbal medicines, essential oils, fodder, honey, charcoal and food for domestic and commercial uses.

3.1.2 Selection and Description of Study Sites

The study sites were Solit (ST), Kimorok (KK), Kapkun (KN) and Maoi (MI), and selection was based on survey of high stand density, occurrence and wide distribution of *Acacia senegal* and its varieties in their habitats.

(a) Solit

Solit is located about 9 kilometres south of Koriema centre along Marigat-Kabarnet road in Sabor sub location, Kimalel location. It lies at latitude 0^0 25' N and longitude 35⁰ 53' E. The topography is 3% slope with an altitude of 1300 metres above sea level.



Plate 3.1: Solit Site - Acacia senegal var. senegal and other closely related species

The soils are shallow with moderate erosion on the upper part to severe erosion on the lower part of the site by water run off during rainy seasons. The study site covered 0.5 hectare and was dominated by moderate stands of *Acacia senegal* var. *senegal* and other closely related species (Plate 3.1).

(b) Kimorok

Kimorok is located about 5 kilometres from Marigat town along Marigat-Kabarnet road in Kimalel sub-location, Kimalel location (Plate 3.2).



Plate 3.2: Kimorok Site - Acacia senegal var. kerensis and other closely related species

UNIVERSITY OF NAIROBI LIBRARY P. O. Box 30197

It lies at latitude 0^0 28' N and longitude $35^055'$ E. The topography is 1% gradient with an altitude of 1200 meters above sea level. The soils are shallow, stony and rocky and moderately swept by gully erosion during rainy seasons. The site was 0.8 hectare, dominated by high stand density of *Acacia senegal* var. *kerensis* and other closely related species with sparse distribution of grass cover, bushes and shrubs.

(c) Kapkun

Kapkun is located about 12 kilometres to the south of Marigat-Kabarnet road in Sabor sublocation, Kimalel location (Plate 3.3). It lies at latitude 0° 23' N and longitude 35° 54' E.



Plate 3.3: Kapkun Site - Both Acacia senegal variety senegal and kerensis with other closely related species

The topography is moderately plain but slants from east to west with 2% gradient and an altitude of 1300 metres above sea level. The site was 1.0 hectare, dominated by *A. senegal* var. *senegal*, and *A. senegal* var. *kerensis* and other closely related species.

(d) Maoi

Maoi is located about 2 kilometres east of Maoi centre along Marigat-Nakuru road in Kaplelwo sub-location, Kaibosoi location (Plate 3.4). It lies at latitude 0^0 24' N and longitude $35^056'$ E.



Plate 3.4: Maoi Site - Acacia senegal var. kerensis and other closely related species The topography is mainly plain with a gentle slope on the eastern part of about 1% gradient with an elevation of 1200 metres above sea level. The soils are shallow and rocky with moderate erosion by water run-off during rainy seasons. The site was 1.0 hectare, dominated by high stand density of *A. senegal* var. kerensis and other closely related species.

3.2.1 Soil Sampling and Pretreatment

Soil samples were dug at a depth of 15-25 cm with a mattock, from five different holes and mixed in a basin to give one representative sample in all the four study sites. Thirty samples of about 600 g each were collected and put in labelled polythene bags. A soil pit of about 60 to 100 cm deep was dug in each site and horizon boundaries identified, marked

and measured to provide the soil profile. Samples in each pit were taken from each horizon for determination of bulk density, particle density and porosity (Plate 3.5).



Plate 3.5: Soil sampling in the Study sites

Soil samples were air-dried, ground in a mortar and sieved through 2 mm sieve mesh to remove gravel and plant materials for analysis of physical and chemical properties and for glasshouse pot experiments.

3.3 Soil Analysis

3.3.1 Physical properties of Soils

(a) Soil texture

Mechanical analysis was used to determine soil texture based on the difference in velocity of settling particles between coarse and fine fractions within a water column. Soil was shaken in a dilute alkaline solution of 10 % sodium hexametaphosphate (dispersing agent) to disperse the coarse and fine colloids in the soil aggregates. The dispersing agent removes oxides of iron and aluminium, calcium carbonate (CaCO₃) and organic matter that have a cementing action on the particles. The effectiveness of this dispersing agent depends upon the adsorption of sodium in exchange for other cations, and the resultant development of strong electrical repulsion forces between the soil particles.

Soil texture analysis (particle size analysis) was determined by the hydrometer method of Bouyoucos (1962). Air-dried sample (50g) portions were weighed into 500 ml bottles and placed in a mechanical shaker. Aqueous sodium hexametaphosphate (10%) dispersing agent (50 ml) and 300 ml of distilled water were added and the bottles tightly stoppered and shaken over night. The contents were quantitatively transferred into a 1000 ml graduated measuring cylinder and made to the mark with distilled water. The procedure used to treat the blank was similar to the treatment of the sample (suspension mixture). The suspension was mixed thoroughly with a plunger. The hydrometer was lowered gently into it and the reading taken at the upper edge of the meniscus surrounding the stem. The suspension mixture was stirred thoroughly again with a plunger for one minute, and time was recorded when the stirring ceased. The hydrometer was carefully lowered into the suspension again and the reading and temperature recorded 40 seconds after the previous timing. This reading was used to determine silt and clay.

Two hours following the previous reading, the second hydrometer reading and temperature were taken for both the soil suspension and blank. This reading was used for the determination of clay fraction (Hinga *et al.*, 1980).

(b) Bulk density, Particle density, Porosity, Percolation Rate and Soil moisture content

Bulk density, particle density and porosity were determined by methods of Blake (1965) and Hinga *et al.* (1980), while percolation rate and soil moisture content were determined by methods of Anderson and Ingram (1993) and Okalebo *et al.* (2002). For bulk density, coring of cylindrical metal samplers of 5 cm diameter of known volume were driven to 5 cm deep into the soil surface in each field site. The core sampler was filled with the soil and then carefully dug out with its contents so as to obtain a core sample in a natural soil condition. The soil sample was carefully removed, trimmed around the cylinder with a small shovel on both ends and tightly closed with caps on top and bottom of the cylinder. Core samples were oven dried at 105^{0} C and weighed for determination of bulk density.

Particle density was measured by weighing clean, dry pyknometer in air and filled about one third full with 2 mm air-dry soil in duplicate. It was heated gently for a few minutes to remove entrapped air and was cooled with its content to room temperature, carefully stoppered and the outside wiped with a clean, dry cloth. The contents were weighed and temperature of the suspension was measured for particle density, taken as the ratio of the total mass (weight) of solid particles to their volume, excluding pore spaces.

Porosity was determined by core method according to (Blake, 1965) as the bulk density, where the core samples were saturated with water and all the pores filled. Core soil samples were oven dried at 105° C_w to a constant weight. The volume of the water was obtained by converting the weight lost on drying at 105° C on the basis of density of water (D_w = gcm⁻³), the same as the bulk volume of the core sample. Then porosity was calculated from the dry bulk density and particle density given by the expression as a volume percentage equal to the volume percent water content at saturation.

Percolation rate was determined by filling a large and small-diameter steel cylinder with water. Water in the large steel cylinder was allowed to drop to the same level as in the small cylinder and time was recorded at intervals of 5 minutes, and then refilled up to its original level. The height of water in the small diameter cylinder was allowed to fall for about 15

minutes and then refilled up to its original level. The height of water in the outer ring was adjusted throughout the measurements until a constant rate was obtained. Percolation rates were measured from the water level at predetermined time intervals and the rate of inflow dropped with time and the experiment was stopped when the rate remained constant after three hours of percolation.

Soil moisture was determined by weighing air-dried soil sample in a clean dry bottle and placed in an oven at 105^oC for 6 hours, removed and allowed to cool in a desiccator for 30 minutes and then re-weighed to obtain an oven-dry weight.

3. 4 Determination of Chemical properties of Soils

Organic carbon, total nitrogen, soil pH water (1:2), soil pH calcium chloride ($0.01M CaCl_2$), electrical conductivity, available phosphorus, exchangeable cations (potassium, magnesium, calcium and sodium) and micronutrients (copper, iron, manganese and zinc) were determined by the procedures described in the subsequent sections.

3.4.1 Organic carbon

Organic carbon was determined by the method of Okalebo *et al.* (2002). An air-dry soil sample (0.5g) was weighed in a 500 ml conical flask and 5 ml of 1.0 M potassium dichromate solution and 7.5 ml of concentrated sulphuric acid added. The contents were placed in a heated block at 145-155^oC for 30 minutes, removed and cooled to room temperature. The digested sample was transferred to a 100 ml conical flask and 0.3 ml of ferroin indicator solution added before titrating with 0.2 M ferrous ammonium sulphate solution. All the titres were corrected for blank and recorded as a measure of unused potassium dichromate.

3.4.2 Total Nitrogen by Kjeldhal Method

Total nitrogen by Kjeldhal method was determined according to the methods of Anderson and Ingram (1993) and Okalebo *et al.* (2002). A soil sample (0.3 g) was weighed and transferred to labeled clean digestion tubes and 2.5 ml digestion mixture (prepared by dissolving 3.5 g of selenium powder and 72 g of salicylic acid in 1 litre of concentrated sulphuric acid) was added, mixed well and placed in a block digester (Skalar Block Digester System, Model SA 5640).

The digestion tubes were heated to 110° C for 1 hour and removed to cool for 30 minutes before adding three 1 ml aliquots of hydrogen peroxide (30%) in succession. The contents were returned to the block digester and heated at 330° C for two hours to obtain clear, colourless digests. The contents were removed and allowed to cool, mixed with 25 ml of distilled water and then transferred into a 50 ml volumetric flask and made up to the mark with distilled water. An aliquot of 5 ml solution was transferred to the distillation apparatus and 10 ml of 40% sodium hydroxide added. 5 ml of 1% boric acid and 4 drops of mixed indicator (a reagent used to indicate the end point in a titration procedure) were added to the distillate and then distilled till the indicator turned from red to green. The distillate was titrated with 0.05 M HCl till the colour changed from green to grey.

3.5 Soil pH and Electrical conductivity

Soil pH and electrical conductivity were measured according to the procedures of Anderson and Ingram (1993) and Okalebo *et al.* (2002), respectively. A 20g sample was weighed into 100 ml plastic bottles and 50 ml distilled water added. The contents were shaken for 10 minutes, allowed to stand for 30 minutes, and shaken again for 2 minutes. The soil suspension was allowed to settle for one hour. The pH and electrical conductivity were measured using a pH meter (Model 691) and conductivity meter (Model TOA Cm-20S), respectively.

3.5.1 Extractable Phosphorus

The available phosphorus was extracted using the procedure of Olsen *et al.* (1954) and analysed according to the method of Anderson and Ingram (1993). Air-dry soil (5 g) was weighed into a 250-ml shaking bottle and 100 ml of 0.5 M sodium bicarbonate solution added. The contents were shaken on a mechanical shaker for 30 minutes at 200-300 rpm.

The suspension was filtered through Whatman No. 42 filter paper and an aliquot of the filtrate (10 ml) pipetted into a 50-ml volumetric flask. To this was added 5 ml of 0.8 M boric acid and 10 ml of ascorbic acid reagent. The contents were diluted to the mark with distilled water, mixed well and let to stand for 1 hour to allow full colour development. A blank was made by substituting the sample with 5 ml of distilled water and carrying through the procedure. A standard series containing 0.0, 1.0, 2.5, 5.0, 7.5, 10, and 12 ppm P was complexed similarly and **a**bsorbance values of all the solutions were measured at 880 nm (Model UV Spectronic 21-Milton Roy Co.).

3.5.2 Soil Extractable Macronutrients

(a) Potassium and Sodium

Potassium and sodium were determined using a flame photometer (Corning M 410) (Anderson and Ingram, 1993). An air-dry soil sample (< 2-mm) of 5 g was weighed into a 250-ml clean plastic bottle and 100 ml of 1.0 M neutral ammonium acetate solution added and the contents were shaken on a mechanical shaker for 30 minutes (Neutral ammonium acetate was used to extract maximum cations that occupy exchange sites on the soil surface). The contents were filtered (Whatman No. 42 filter paper). To 5 ml of the filtrate contained in a 50 ml volumetric flask was added 1 ml of 26.8 % Lanthanum chloride solution and the contents made to the mark with the ammonium acetate solution. Standards containing potassium and sodium at concentrations 0.0, 2.5, 5.0, 10, 15 and 20 ppm were prepared similarly (to fall within the measurable range of the calibrated flame photometer and atomic absorption spectrophotometer). The flame emission intensities were measured at 766 and 589 nm for potassium and sodium, respectively.

(b) Calcium and Magnesium

An air-dry soil sample (< 2-mm, 5 g) was weighed into a 250-ml clean plastic bottle and 100 ml of 1.0 M ammonium acetate solution (pH 7.0) added. The bottle was stoppered and shaken on a mechanical shaker for 30 minutes. The contents were filtered (Whatman No. 42) and 5 ml of soil extract pipetted into a 50 ml volumetric flask. To the aliquot 1 ml of 26.8 % Lanthanum chloride solution was added before making to volume with the ammonium acetate solution. Standards containing calcium and magnesium at concentrations 0.0, 2.5, 5.0, 10, 15 and 20 ppm Ca and 0.0, 0.4, 1.0, 2.0, 3.0, and 4.0 ppm Mg were prepared similarly (to fall within the measurable range of the calibrated flame photometer and atomic absorption spectrophotometer). The absorbance readings were taken at 422.7 and 285.2 nm, respectively for calcium and magnesium using a Shimadzu Model AA – 6410 F atomic absorption spectrophotometer (Anderson and Ingram, 1993).

3.5.3 Soil Extractable Micronutrients

Extractable micronutrients (iron, zinc, manganese and copper) in soils were determined at wavelengths of 248, 213, 279 and 324.8 nm, respectively, by Shimadzu Model AA - 6410 F atomic absorption spectrophotometer according to the method of Anderson and Ingram (1993). A 5 g air-dry soil sample (< 2-mm) was weighed into a 250 ml clean plastic bottle

and 50 ml of 1% EDTA extracting solution added. (EDTA, a chelating agent was used because of its capability to extract all forms of metal-chelate ionic complexes at low concentrations of available iron, zinc, manganese and copper in soils). The sample was shaken on a mechanical shaker for 1 hour. The suspension was filtered (Whatman No. 42) and 5 ml of the filtrate pipetted into a 50 ml volumetric flask. The contents were filled to the mark with 1% EDTA extracting solution. Standards were prepared containing 0.0, 0.5, 0.75, 1.0, 1.5 and 2.0 ppm of Cu and Zn, and 0.0, 2.0, 4.0, 6.0, 8.0, and 10.0 ppm of Fe and Mn, respectively, using the extracting solution.

In these methods of soil extractions, initial recovery tests in soils were valid for all the nutrients to achieve a significant recovery test. However, dilution effects of nutrients uptake inhibited the recovery tests over time because of soil heterogeneity. Standard methods of Okalebo *et al.* (2002) and Anderson and Ingram (1993) on recovery tests on soil nutrients amelioration were adopted in the study.

3.6 Micronutrients Uptake Studies

3.6.1 Seed Collection

A flowering survey was carried out from January to March 2002 to determine the proper time for seed collection in the study areas. About 200g of *Acacia senegal* seeds were collected randomly by hand from selected healthy trees in each site. The seeds were dried and stored in polythene bags for establishment in glasshouse pot experiments.

3.6.2 Seed Germination in the Controlled Growth Chamber

Seeds were sown in a plastic trough with sterilized pure sand. The germinated young seedlings were placed in the automated controlled growth chamber to prevent fungal attack.

Temperature was set at 28°C, light (photoperiod) and air circulation were automatically controlled (Plate 1).



Plate 1 Controlled Growth Chamber for Seed Germination

The seeds were irrigated with distilled water after every four days to provide adequate moisture for germination. Tree seedlings were germinated and transplanted after 21 days into half-filled pots with soil and vermiculite media under glasshouse conditions that simulated the ASAL environment (Plate 2).



Plate 2 Glasshouse Experiment

The vermiculite, a solid inert aggregate, was added to provide a homogenous medium for raising tree seedlings in the glasshouse experiments (Stout, 1961).

3.6.3 Glasshouse Experimental Design

A Split-plot block design was used for this study. The layout comprised main plots, subplots and sub-subplots. The soil and vermiculite media were assigned to the main plots. Sites and micronutrient treatments were assigned to the subplots and sub-subplots, respectively, in two replicates. Table of random numbers was used to allocate media, sites and treatments randomisation in order to take care of interactions between and within treatments, sites and media. Hydroponics nutrient solution was used for micronutrient uptake by the seedlings in the glasshouse pot experiment (Stout, 1961).

3.6.4 Preparation of Hydroponics Solution Culture (Balanced Nutrient Solution Media)

Hydroponics normal solution culture, a balanced nutrient medium that contained adequate concentrations of all essential elements was prepared from reagent salts of analytical grade or equivalent. For preparation of nutrient solutions: nitrogen, phosphorus and potassium, a concentration of 2000 ppm of each element was made by dissolving 6.9989g (NH₄)₂FeSO₄.6H₂O and 14.0711g potassium nitrate (KNO₃), 8.7872g potassium hydrogen phosphate (KH₂PO₄) and 5.1718g potassium nitrate (KNO₃) respectively, in one litre of distilled water. Calcium and magnesium at 8000 ppm and 1000 ppm, respectively, were prepared by dissolving 29.3440g calcium chloride (CaCl₂.2H₂O) and 10.1406 g magnesium sulphate (MgSO₄.7H₂O), in one litre of distilled water (Stout, 1961). For the micronutrients copper, iron, manganese and zinc, concentrations of 6 ppm, 100 ppm, 50 ppm and 20 ppm were prepared by dissolving 0.0235 g copper sulphate (CuSO₄.5H₂O), 0.4978 g ferrous sulphate (FeSO₄.7H₂O), 0.2030 g manganese sulphate (MnSO₄.4H₂O) and 0.0879 g zinc sulphate (ZnSO₄.7H₂O) in one litre of distilled water (Stout, 1961).

3.6.5 Preparation of Low_Micronutrient Medium

Copper, iron, manganese and zinc concentrations of 2 ppm, 50 ppm, 25 ppm and 10 ppm were prepared by dissolving 0.0079 g CuSO₄.5H₂O, 0.2489 g FeSO₄.7H₂O, 0.1015 g $MnSO_4.7H_2O$ and 0.0440 g ZnSO₄.7H₂O in one litre of distilled water (Stout, 1961).

3.6.6 Preparation of High Micronutrient Medium

Copper, iron, manganese and zinc concentrations of 10 ppm, 150 ppm, 100 ppm and 40 ppm were prepared by dissolving 0.0393 g CuSO₄.5H₂O, 0.7467 g FeSO₄.7H₂O, 0.4060 g $MnSO_{4}.7H_{2}O$ and 0.1759 g of ZnSO₄.7H₂O in one litre of distilled water (Stout, 1961).

3.6.7 Application of Nutrient Solutions, Monitoring and Data Collection Procedure

The hydroponics balanced nutrient solutions were applied, drained and replaced with fresh solutions after every fourteen days to minimise depletion in nutrient levels during the growth of tree seedlings (Plate 3).



Plate 3 Application of dosage treatments to Acacia senegal seedlings

Because of potential fluctuations in pH during the process of nutrient uptake by the *Acacia senegal* seedlings, the concentration of each element in the nutrient solution was buffered with 1 ml of 1 M ammonium acetate solution at pH 7.0 in each media. The dose levels of the micronutrient media were applied in increasing order as low, normal (balanced) and high concentrations. The choice of these levels was based on balanced nutrient levels occurring in higher plants as suggested by Stout (1961).

There were a total of 144 pots. Seventy two were each filled with 500g of soil. The rest were filled with 500g of vermiculite. In each pot was planted one seedling. The experiment was replicated twice and each treatment was randomly allocated in a row (block). The position of blocks by soil or vermiculite per site was randomly allocated for one medium. There were 9 treatments randomized to either vermiculite or soil with micronutrient solutions (Cu, Fe, Mn, and Zn) in Table 3.1).

	Copper (Cu)	Iron (Fe)	Manganese (Mn)	Zinc (Zn)		
Treatment Levels	ppm					
Normal						
concentration	6	100	50	20		
Low concentration	2	50	25	10		
High	10	150	100	40		
concentration						
Footnotos	*					

 Table 3.1 Categorizations of Nutrient Treatments (Stout, 1961)

Nutriant Concentrations

Normal concentrations represent treatment Number 1

Low concentrations represent treatment Numbers 2, 4, 6 and 8

High concentrations represent treatment Numbers 3, 5, 7 and 9

The seedlings were applied with normal treatments of 10 ml three times in a month for two hundred and forty days. This was to allow each seedling to attain stable growth before applications of low and high dosage treatments. Diameter at ground level (Dgl) and heights (ht) in millimetres were recorded once in thirty days during the period of study.

3.8 Gum arabic Analysis

Ninety gum arabic samples of about 200g were collected during the dry seasons from the two varieties of *Acacia senegal*: *senegal* and *kerensis*. The samples were air dried for fourteen days and ground using a pestle and mortar for analysis of moisture content, ash, volatile matter, internal energy, nitrogen and the trace elements; copper, iron, manganese and zinc.

3.8.1 Determination of the Moisture and Ash contents

A portion of sample (5 g) was weighed into a clean dry crucible of known weight and oven dried at 105°C for 6 hours, and cooled in a dessicator for 30 minutes. The crucibles were re-weighed to determine the moisture content as percent ratio of the change in weight to the original sample weight. The dry weight was then placed in a muffle furnace and temperature raised to 550°C. After one hour, the crucibles were removed, cooled in a dessicator for 30 minutes and weighed. The contents were ignited at 900°C for two hours, cooled in a dessicator for 30 minutes and weighed. The contents were ignited at 900°C for two hours, cooled in a dessicator for 30 minutes and weighed. The ash content was taken as the percent loss in weight after ignition that of the original sample.

3.8.2 Determination of Internal energy in gum arabic samples

Gum arabic (5 g) was weighed into a clean dry crucible of known weight and then placed in a muffle furnace and temperature raised to 550° C to obtain optimum internal energy. The furnace was open to cool to 150° C to eliminate moisture, before putting in a dessicator to cool to room temperature for 30 minutes and weighed. The contents were ignited at 850° C for two hours, then cooled in a dessicator for 30 minutes and weighed. Internal energy was determined as the percent loss in weight after ignition of the original sample.

3.8.3 Determination of the Volatile matter in gum arabic samples

A clean dry crucible of known weight was weighed with 5 g of gum arabic sample and oven dried at 105°C for 6 hours and then removed and cooled in a dessicator for 30 minutes. The crucibles were re-weighed to determine the volatile matter as percent ratio of the change in weight to the original sample weight according to the methods of Anderson and Ingram (1993 and Okalebo *et al.* (2002).

3.8.4 Determination of Nitrogen content and Trace elements in gum arabic samples

The methods for determination of nitrogen content and trace elements in gum arabic samples are similar to the methods described earlier in sections 3.4.2 and 3.5.3.

3.9 Statistical Methods of Data Analysis

Statistical analysis of data was carried out using SPSS for windows Release 8.0.0 (1997), and Microsoft Excel (2003) computer software. The statistical method used was Analysis of Variance (ANOVA) using Generalized Linear Models Procedure (GLM). Generalized Linear Modelling equations for soil physical and chemical analysis:

 $Y_{ijk} = \mu + S_i + \varepsilon_{ij}$

Where,

 $\mu = Overall mean,$

 $S_i = effect of i^{th} site, i = 1, 2, 3, 4.$

 $\mathcal{E}_{ij} = random \, error$

3.8.3 Determination of the Volatile matter in gum arabic samples

A clean dry crucible of known weight was weighed with 5 g of gum arabic sample and oven dried at 105°C for 6 hours and then removed and cooled in a dessicator for 30 minutes. The crucibles were re-weighed to determine the volatile matter as percent ratio of the change in weight to the original sample weight according to the methods of Anderson and Ingram (1993 and Okalebo *et al.* (2002).

3.8.4 Determination of Nitrogen content and Trace elements in gum arabic samples

The methods for determination of nitrogen content and trace elements in gum arabic samples are similar to the methods described earlier in sections 3.4.2 and 3.5.3.

3.9 Statistical Methods of Data Analysis

Statistical analysis of data was carried out using SPSS for windows Release 8.0.0 (1997), and Microsoft Excel (2003) computer software. The statistical method used was Analysis of Variance (ANOVA) using Generalized Linear Models Procedure (GLM). Generalized Linear Modelling equations for soil physical and chemical analysis:

 $y_{ijk} = \mu + S_i + \varepsilon_{ij}$

Where,

 μ = Overall mean,

 $S_i = effect of i^{th} site, i = 1, 2, 3, 4.$

 $\varepsilon_{ij} = random \, error$

Analysis of micronutrients uptake:

 $\mathbf{y}_{ijkl} = \mu + \mathbf{c}_i + \mathbf{s}_j + \mathbf{m}_k + (\mathbf{s}^{x} \mathbf{m})_{jk} + \varepsilon_{ijkl}$

Where,

 μ = Overall mean,

c i = Covariate

 $Si = effect of i^{th} site, i = 1, 2, 3, 4.$

 m^{k} = effect of the kth medium, k = 1, 2;

 $S^{x}m$ = interaction effect of kth medium in the ith site, i = 1, 2, 3, 4, k = 1, 2,

 $\varepsilon_{ijklmn} = random error.$

Generalized Linear Modelling equations for Analysis of heights and diameter at ground level (Dgl).

 $y_{ijk1} = \mu + s_i + m_j + t_k + (s^{x}m)_{ij} + (s^{x}t)_{ik} + (m^{x}t)_{jk} + \varepsilon_{ijk1}$

Where,

 μ = Overall mean,

 $S_i = effect of i$ th site, i = 1, 2, 3, 4.

 $m_j = effect of the m^{th} medium, j = 1, 2;$

 t_k = effect of the t th treatment, k = 1, 2, 3, 4, 5, 6, 7, 8, 9

s = m = m = m = 1, 2, 3, 4, j = 1, 2, 3, 4, 5, 6, 7, 8, 9.

 $s^{x}t =$ interaction effect of ith treatment in the kth medium, l = 1, 2, 3, 4, 5, 6, 7, 8, 9, k = 1, 2. m^xt = interaction effect of jth treatment in the kth medium, where j = 1, 2, 3, 4, 5, 6, 7, 8 and 9, k = 1, 2,

 ε_{11k1} = random error.

Analysis of variance was caried to test significant differences among the means of physical and chemical properties of soils, gum arabic, micronutrients concentration, diameter at ground level and heights in media (soils and vermiculite), sites, treatments and variety between and within sites in the experimental unit. Pearson correlation analysis was also used to correlate levels of soil chemical properties, gum chemical properties, low, normal and high dosage treatments of micronutrients uptake by seedlings of *Acacia senegal* variety *kerensis* and *Acacia senegal* variety *senegal* (Meredith and Stehman 1991).

CHAPTER FOUR

CHARACTERIZATION OF SOILS AND SOIL FERTILITY STATUS

4.1 Physical Properties of Soils

(a) Soil Texture

The physical properties of soils are given in Table 4.1. Soil texture in Solit and Maoi comprised 68.75 % sand, 16.69 % silt, and 14.56 % clay and 62.29 % sand, 27.23 % silt, and 10.49 % clay, respectively (Appendix I). Kimorok and Kapkun comprised 55.47 % sand, 27.60 % silt, and 16.27 % clay, and 57.60 % sand, 26.50 % silt, and 15.90 % clay, respectively.

		Solit (n=32)	Kapkun (n=32)	Kimorok(n=32)	Maoi (n=32)
Sand	%±SE	68.75 <u>+</u> 0.72	57.6+0.75	55.47 <u>+</u> 0.75	62.29 <u>+</u> 0.69
Clay	%+SE	14.56 <u>+</u> 0.62	15.9 <u>+</u> 0.64	16.27 <u>+</u> 0.64	10.49 <u>+</u> 0.59
Silt	% <u>+</u> SE	16.69 <u>+</u> 0.59	26.5+0.61	27.6 <u>+</u> 0.61	27.23 <u>+</u> 0.56
Bulk density	g cm ⁻³ +SE	1.24 <u>+</u> 0.04	1.19 <u>+</u> 0.04	1.22 <u>+</u> 0.04	1.17+0.04
Particle density	g cm ⁻³ +SE	1.61 <u>+</u> 0.10	1.67 <u>+</u> 0.10	1.69 <u>+</u> 0.10	1.72 <u>+</u> 0.10
Porosity	% <u>+</u> SE	23+5.78	29 <u>+</u> 5.78	26.75 <u>+</u> 5.78	28.75 <u>+</u> 5.78
Percolation	ml min ⁻¹ ±SE	33 <u>+</u> 11.83	11.5 <u>+</u> 11.83	40 <u>+</u> 11.83	27.5+11.83
Moisture	%±SE	8.50±1.66	3.50±1.66	13.5±1.66	11.5±1.66

Table 4.1: Th	e Physical	properties	of Soils
---------------	------------	------------	----------

Footnote:

n = number of samples analysed

S.E. = Standard error

Sand content in Solit and Maoi of 68.75 % and 62.28 % was significantly higher (P < 0.05) than 55.47 % and 57.60% in Kimorok and Kapkun, respectively. Silt content of 16.69 % in
Solit was significantly lower (P < 0.05) than 27.6%, 26.5 and 27.3% in Kimorok, Kapkun and Maoi, while 10.49 % clay in Maoi was significantly lower (P < 0.05) than 16.27%, 15.90% and 14.56% in Kimorok, Kapkun and Solit, respectively (Figure 4.1).

Soil texture in Solit and Maoi was sandy loam with high sand contents (68.75 % and 62.28 %) due to the fact that weathering of parent material into inorganic constituents of soil minerals formed sandy soils. These soils were low in clay and high sand contents with low soil organic matter of 0.78% and 1.15%, respectively (Table 4.2).

Sandy soils in Solit and Maoi had low soil moisture content (8.5 and 11.5%) which influenced decomposition process of low soil organic matter. Kimorok and Kapkun had sandy clay loam with clay contents (16.27 % and 15.90 %), respectively, which revealed that clay content had higher organic matter content (1.97% and 1.73%) than that of sandy soils.



Figure 4.1 Soil Texture in the Study sites

The results of texture were similar to that of sandy soils reported by Landon (1991) that soils with high clay (20%) generally has high organic matter content (3.0%), because of slow decomposition of organic matter compared with sandy soils. Both texture had good drainage and aeration and may be susceptible to water erosion during wet seasons.

(b) Bulk and Particle Densities

The average bulk and particle densities were 1.21 and 1.66 g cm⁻³ in all the sites. The values were high because of high sand content which indicated that levels of soil organic matter and soil moisture characteristics were low in the study area (Table 4.1).





The average bulk density of 1.2 g cm⁻³ was similar to that of Chemeron Irrigation Scheme in Marigat reported by Van Engelen *et al.* (1983); the range (1.2 g cm⁻³ and 1.3 g cm⁻³) was within that reported by Anon (1991).

(c) Porosity and Percolation

porosity in Kapkun, Kimorok and Maoi of 29%, 26.75% and 28.75% was high because of silt clay texture, while percolation rate in Kimorok, Solit and Maoi of 40, 33 and 27.5 ml/min was high as attested to influence of high levels of sand content, respectively (Table 4.1). Percolation rate of 40 ml/min in Kimorok was high because of influence of high levels of silt and clay contents of 27.60% and 16.27%, respectively.



Figure 4.3 Porosity and Percolation in Sites

Percolation rate in Kapkun was lower than in Solit, Kimorok and Maoi because of high silt (26.50%) and low moisture which influences nutrient uptake and rainfall infiltration in the soil (Landon, 1991).

(c) Soil Moisture Content

Soil moisture (13.5 %) in Kimorok was significantly higher (P < 0.05) than 3.5 % in Kapkun, respectively (Figure 4.4).



Figure 4.4 Soil Moisture content in Sites

Soil moisture of 13.5% in Kimorok was high as a result of influence of high levels of soil organic matter and clay content of 1.97% and 16.27%, respectively. Organic matter permits the infiltration of water and acts as a biological buffer ensuring that a balanced supply of nutrients is available to the plant roots, and clay particles contribute to the stability of soil aggregates.

Soil moisture in Kapkun was low due to the effect of soil texture (sandy clay loam) and low percolation of water (11.5 ml/min), in which clay soils are so finely textured that very little water penetrates to lower levels of soil horizon. The results of moisture content were low that revealed water stress limits the nutrient uptake by plants in all the sites. The effects of low moisture regime affect the physiological growth of plants and conditions of soil microorganisms in arid and semi-arid environments.

4.2 Chemical Properties of Soils

(i) Soil pH

The soil pH values were significantly different from each other in all the sites (P < 0.05, Appendix II). Soil pH 7.05 in Solit was significantly higher than those of Kimorok, Kapkun and Maoi (6.0, 5.9 and 6.19), respectively (P < 0.05, Table 4.2). The characterization criteria for soils are given in Table 4.2.

	Site	Solit (n= 62)	Kimorok (n= 62)	Kapkun (n= 62)	Maoi (n= 62)	
		Mean±S.E.	Mean±S.E.	Mean±S.E.	Mean±S.E.	
рН		7.05 <u>+</u> 0.04	6.05 <u>+</u> 0.04	5.96 <u>+</u> 0.04	6.19 <u>+</u> 0.04	
EC	mS/cm	0.06 <u>+</u> 0.00	0.05 <u>+</u> 0.00	0.04 <u>+</u> 0.00	0.04+0.00	
С	%	0.46 <u>+</u> 0.03	1.15+0.03	1.01 <u>+</u> 0.03	0.67 <u>+</u> 0.03	
C:N		2.99	9.37	4.82	4.25	
SOM	%	0.78+0.06	1.97 <u>+</u> 0.06	1.73 <u>+</u> 0.06	1.15 <u>+</u> 0.06	
N PO ₄ ³ -P	%	0.18+0.02	0.14+0.02	0.30 <u>+</u> 0.02	0.17 <u>+</u> 0.02	
	ppm	11.54 <u>+</u> 1.00	14.60 <u>+</u> 1.01	14.10 <u>+</u> 1.02	13.91 <u>+</u> 0.99	
К	cmol (⁺)/kg	0.45 <u>+</u> 0.05	1.24 <u>+</u> 0.05	1.58 <u>+</u> 0.05	1.16 <u>+</u> 0.05	
Ca	cmol (⁺)/kg	17.33 <u>+</u> 1.27	5.74 <u>+</u> 1.28	6.40 <u>+</u> 1.30	4.97 <u>+</u> 1.26	
Mg	cmol (⁺)/kg	6.08 <u>+</u> 0.24	3.31+0.24	3.99 <u>+</u> 0.24	2.75 <u>+</u> 0.23	
Na	cmol (⁺)/kg	1.00 <u>+</u> 0.10	0.54+0.10	0.99 <u>+</u> 0.11	1.17 <u>+</u> 0.10	
Cu	ррт	1.67 <u>+</u> 0.08	0.73 <u>+</u> 0.08	0.61 <u>+</u> 0.09	0.58 <u>+</u> 0.08	
Fe	ppm	151+11	234 <u>+</u> 11	250 <u>+</u> 11	287 <u>+</u> 11	
Mn	ppm	263 <u>+</u> 14	320 <u>+</u> 14	344 <u>+</u> 14	383 <u>+</u> 14	
Zn	ppm	1.73 <u>+</u> 0.47	7.64 <u>+</u> 0.46	6.00±0.47	5.06 <u>+</u> 0.46	

Table 4.2: Characterization Criteria for soils

Footnote: n= number of samples analysed; S.E. = Standard error

The soils of Solit were neutral because of high levels of calcium and magnesium of 17.33 and 6.08 cmol (⁺)/kg, respectively, while those of Kimorok, Kapkun and Maoi were acidic soils as a result of increased the levels of soil organic matter of 1.97%, 1.73% and 1.15%, respectively. pH values were similar to the values reported by Peverill *et al.*, (1999) on acidic and neutral soils that had a pH range of 5.5-7.05 which indicates adequate availability of calcium, potassium, nitrogen, phosphorus and micronutrients to most plants. The report also states that the soils of many arid and semi-arid areas have favourable pH range of about 6 and 7, but there is not enough precipitation to optimize the availability of nutrients for most plants. Brady and Weil (1999) reported that soil pH range in soils varied and the availability of nutrient uptake by plants is influenced by other factors such as texture, soil organic matter, soil moisture, leaching and erosive effects.

(ii) Electrical conductivity

Electrical conductivity of 0.06 mS/cm in Solit was significantly higher than those of Kimorok, Kapkun and Maoi (0.05, 0.04, and 0.04 mS/cm), respectively (P < 0.05, Table 4.2). This was because of sand content of 68.75%, soil pH 7.05, high levels of calcium and magnesium (17.33 and 6.08 cmol (⁺)/kg) and Ca: Mg cation ratio of 3:1, which may have influenced the level of salinity in the soils and the availability of nutrient uptake by plants (Jaynes, 1996; Lund *et al.*, 1999).

The average level of salinity in all the sites was 0.05 mS/cm. This was low in view of the fact that moisture content averaged 9.3% aggravated by low and erratic rainfall in the study area. The low moisture limited the availability of nitrogen, phosphorus and micronutrients uptake by plants in the soils (Sudduth *et al.*, 1998).

(iii) Carbon and Carbon: Nitrogen ratio

Carbon content (1.15 and 1.01%) in Kapkun and Kimorok were significantly higher than those of Solit and Maoi (0.46% and 0.67%), respectively (P < 0.05, Table 4.2). The high levels of carbon in Kapkun and Kimorok was because of increased the levels of soil organic matter of 1.97% and 1.73%, respectively.

Carbon to nitrogen ratio (9.37:1.00) in Kimorok was significantly higher than the ratios for Kapkun, Maoi and Solit (4.82: 1, 4.25:1 and 2.99:1), respectively (P < 0.05, Table 4.2). This was ascribed to high soil moisture content 13.5%, which aids the oxidation of carbon in the decomposition of organic matter to release carbon dioxide in the soil. This oxidation process may have influenced the low carbon content as plants take in water and carbon dioxide in the presence of sunlight energy during photosynthesis.

Soil organic carbon also influences the decomposition of SOM by microorganisms through nitrification process (Landon, 1991). High C: N ratio of 8.21:1 was ascribed to increase in clay content of 16.27% which may have influenced the immobilization of nitrogen uptake by plants and fixing the ammonium ions in the exchange sites of SOM through mineralization during nitrification process (Jones, 1991)

(iv) Soil organic matter (SOM)

Soil organic matter of 1.73 and 1.97% in Kapkun and Kimorok were significantly higher than those of Solit and Maoi (0.78 and 1.15%), respectively (P < 0.05, Table 4.2 and Figure 4.5).



Figure 4.5 Comparison of SOM

This was attested to high clay content that enhanced the soil organic matter. Soil organic matter acts as store and supplies the plant nutrients and improves the biological functions of the soil. Maoi had higher SOM (1.15%) than 0.78% in Solit because the soils were acidic (pH 6.19), where in acid soils organic matter increases while it decreases in neutral soils (pH 7.05). These findings agree with the work by Donahue and Brann (1984), which reported that organic matter in neutral and basic soils (pH 7.0 – 7.8) decreases while it **increases** in acidic soils (pH 5.8 – 6.5).

Microorganisms influence the decomposition of Soil organic matter through nitrification process (Walker and Woodson, 1987; Donahue and Brann, 1984). SOM plays a key role in maintaining good physical conditions of soils including water holding capacity, provides a balanced supply of nutrients, protect nutrients against leaching by improving cation exchange capacity and greater recycling and supply of micronutrients (Vance *et al.*, 1987).

(v) Nitrogen

Nitrogen content (0.3%) in Kapkun was significantly higher than the contents at Solit, Maoi and Kimorok (0.18%, 0.17% and 0.14%), respectively (P < 0.05, Table 4.2 and Figure 4.6).



Figure 4.6 Comparison of Soil N

This was attributed to sandy clay loam with pH 5.96 which influenced the mineralization of ^{soil} organic matter by microorganisms to release oxidised ammonium ions (NH_4^+) to nitrite

and nitrate during the nitrification process (Jones, 1991). This may also be coupled with release of nitrate ions and phosphate ions that accumulated during the long period of dry season, where suddenly green plants spring up and grow rapidly within a very short time at the onset of the first rains in the drylands (Landon, 1991). Since climatic conditions play a key role in the formation of SOM content, some carbon dioxide that is released by microorganisms in the atmosphere dissolves in rainwater to form weak carbonic acid (H_2CO_3). This may react with soil minerals to form carbonates and hydrogen carbonates of calcium, magnesium and potassium within positive and negative surface charges of colloids in the soils in the wet seasons (Vance *et al.*, 1987).

(vi) Phosphorus

Phosphorus contents (11.54 ppm and 14.6 ppm) in Solit and Kimorok were significantly different from each other (P < 0.05), while Kapkun and Maoi (14.10 and 13.91 ppm), were not significantly different, respectively (Table 4.2 and Figure 4.7).



Figure 4.7 Comparison of Soil P

This was as a result of influence of calcium ions $(17.33 \text{ cmol} (^+)/\text{kg})$ at soil pH 7.05 which interacted with phosphate ions to form precipitation of insoluble calcium phosphates and the solubility of phosphate ions is very dependent on soil pH that influences the availability of phosphate ions in soil solution (Tisdale *et al.*, 1985).

(vii) Potassium

Potassium of 1.57 cmol (⁺)/kg in Kapkun was significantly higher (P < 0.05) than 1.24, 1.16 and 0.45 cmol (⁺)/kg in Kimorok, Maoi and Solit, respectively (Table 4.2 and Figure 4.8).



Figure 4.8 Comparison of Soil K

This was in view of the fact that clay content (15.9%) and SOM (1.73%) influenced the ^{availability} of potassium ions in soil solution (Tisdale *et al.*, 1985).

Potassium content (0.45 cmol (⁺)/kg) in Solit was significantly lower than those of Kimorok, Kapkun and Maoi (1.25, 1.57 and 1.16 cmol (⁺)/kg), respectively (P < 0.05).

Potassium content in Solit was low as a result of sandy loam texture with low soil moisture which K^+ ions were prone to leaching and interaction with other cations (Ca²⁺, Mg²⁺ and Na⁺) that reduced the concentrations of K^+ ions within the soil solution during the wet season. There were also high levels of calcium and magnesium (17.33 and 6.08 cmol (⁺)/kg), respectively, which reduced the availability of K⁺ions in the soil.

Kimorok, Kapkun and Maoi had high levels of potassium owing to sandy soils with high levels of soil organic matter that reduced leaching effects and high phosphorus contents that increased K uptake (Bohn *et al.*, 1985). Rhoades (1982) reported that soil organic matter is negatively charged colloids which attract and retain high levels of cations such as potassium (K^+), sodium (Na⁺), calcium (Ca²⁺) and magnesium (Mg²⁺) within the soil aggregate.

(viii) Calcium

Calcium of 17.33 cmol (⁺)/kg in Solit was significantly higher than those of Kimorok, Kapkun and Maoi (5.74, 6.4 and 4.97 cmol (⁺)/kg), respectively (P < 0.05, Table 4.2 and Figure 4.9).



Figure 4.9 Comparison of Soil Ca

Calcium content (17.33 cmol (⁺)/kg) in Solit was high as a result of soil pH (7.05) which influenced interaction of calcium ions in the soil solutions with phosphate ions to form precipitation of insoluble calcium phosphates. The precipitation of calcium phosphate is promoted at high soil pH which influences the fixation of phosphate ions (11.54 ppm) in the soil solution. The results of calcium levels in sandy soils of the study sites were similar to the results of Bremner and Mulvaney (1982) reported that interaction of Ca²⁺ with other cations (HPO₄²⁻, Mg²⁺, K⁺ and Na⁺) increased and protected the concentrations of Ca²⁺ ions ^{against} leaching by improving cation exchange capacity within the sandy soils.

(ix) Magnesium

Magnesium contents (6.08 and 2.75 cmol (⁺)/kg) in Solit and Maoi were significantly different from each other as well as Kimorok and Kapkun (3.31 and 3.99 cmol (⁺)/kg), respectively (P < 0.05, Table 4.2 and Figure 4.10).





Magnesium contents (2.75 and 3.31 cmol (⁺)/kg)) in Maoi and Kimorok were not significantly different from each other, respectively. In Solit, magnesium content (6.08 cmol (⁺)/kg) was high in view of the fact that the high level of calcium (17.33 cmol (⁺)/kg) may have influenced the adsorption of magnesium ions on negatively charged surfaces of soil colloids in SOM. This effect may have inhibited the leaching of Mg²⁺ under field conditions (Landon, 1991). Kapkun soils were more acidic than that of Maoi which had higher magnesium of 3.99 cmol (⁺)/kg) than 2.75 cmol (⁺)/kg. This was attested to high level of SOM (1.73%) compared with 1.15% of Maoi. This also agrees with the report on the work of Donahue and Brann (1984) that organic matter in neutral and basic soils (pH 7.0 – 7.8) decreases while it increases in acidic soils (pH 5.8 – 6.5). This reason applies to levels of SOM in Kimorok and Solit, respectively. Maoi had low level of magnesium

because of leaching effects in more acid sandy soils with Ca: Mg cation ratio of 3:1 on the exchange sites of soil organic matter. Rhoades (1982) reported that clay particles and soil organic matter are negatively charged colloids that may attract and retain cations such as potassium (K⁺), sodium (Na⁺), calcium (Ca²⁺), and magnesium (Mg²⁺). Therefore, level of magnesium cation with high Ca: Mg cation ratio of greater than 5:1 may have influenced magnesium availability in the soil. The availability of exchangeable magnesium in soils also depends on texture, organic matter and leaching effects (Bremner and Mulvaney, 1982).

(x) Sodium

All the sites were significantly different in sodium content from each other (P < 0.05; Table 4.2). Sodium content (1.17 cmol (⁺)/kg) in Maoi was high because of interaction of soluble salts in the exchange sites of SOM. This may have protected sodium ions against leaching by improving cation exchange capacity within the soil aggregates. Kapkun had low level of sodium as a result of acid sandy soils with low moisture and leaching effects that **decreased** sodium cations on the negative charge of the exchange sites of soil organic matter in the soil solution (Landon, 1991).

(xi) Copper

Copper content (1.67 ppm) in Solit was significantly higher than those of Kimorok, Kapkun and Maoi (0.73, 0.61 and 0.58 ppm), respectively, (P < 0.05, Table 4.2 and Figure 4.11).



Figure 4.11 Comparison of Soil Cu

Available copper (1.67 ppm) in Solit was low because of influence of high calcium content (17.33 cmol (⁺)/kg) which may have formed stable complexes with phosphates (11.4 ppm) in clay soils (14.56%) and (0.45% SOM) that inhibited copper ions by strong covalent bonding with organic matter in the soils (Tisdale *et al.*, 1985).

(xii) Iron

Available iron (287 ppm) in Maoi was significantly higher than those of Solit, Kimorok and Kapkun (151, 234 and 250 ppm), respectively (P < 0.05, Table 4.2 and Figure 4.12).



Figure 4.12 Comparison of Soil Fe

Solit and Maoi (151 and 287 ppm) were significantly different from each other (P < 0.05) as well as Kimorok and Kapkun, respectively while Kapkun and Kimorok (250 and 234 ppm) were not significantly different from each other. Available iron (250 and 234 ppm) in Kapkun and Kimorok were attested to soil pH given that by comparing pH of Kapkun with that of Kimorok, at pH 5.96, iron (250 ppm) increased while iron (234 ppm) at pH 6.05 had decreased. The level of iron (151 ppm) in Solit was low because of high soil pH 7.05; hence at soil pH 7.05 available Fe becomes less soluble in the soil. In view of the fact that the solubility of iron is highly dependent upon soil pH and its availability decreases at high pH in the soil solution (Parker and Walker, 1986). Increased soil pH reduces the availability of Fe⁺² to plants through increased adsorption at negative charged exchange sites of SOM (Woolhouse, 1983).

High level of available iron in Maoi was ascribed to acidic soils (pH 6.19), as the soil becomes acidic the availability of iron increases at low pH. The interaction of iron and manganese with similar oxidation states of Fe^{+2} and Mn^{+2} with ionic radii of 0.73 and 0.75, respectively. The cation ratio of Fe: Mn of 1:1 competes with available iron ions for the same carrier site in the soil organic matter which influence the increase of the availability of iron in the soil (Bohn *et al.*, 1985).

(xiii) Manganese

Available manganese of 383 ppm in Maoi was significantly higher than those of Solit, Kapkun and Kimorok (263, 344 and 320 ppm), respectively (P < 0.05, Table 4.2 and Figure 4.13).



Figure 4.13 Comparison of Soil Mn

Available manganese in Maoi was high due to acid sandy soils (pH 6.19) and high soil moisture of 13.5% which aided the oxidation of manganese ions in the soil solution to enhance its availability at low pH. At soil pH 6.19, as the soil becomes acidic the availability of manganese increases, since the solubility of manganese is highly dependent upon soil pH and soil moisture during the decomposition of organic matter to release other cations (Cu^{2+} , Fe^{2+} and Zn^{2+}) in the soil.

In Solit, available manganese (263 ppm) was low by reason of high soil pH (7.05). As soil pH increases Mn^{+2} tends to precipitate, as MnO_2 thus is less available to plants. In addition, high levels of other cations (Fe²⁺ and Zn²⁺) can reduce plant uptake of Mn^{+2} , probably through competition at binding sites of clay soils. Manganese interacts with other cations, particularly iron citrate when applied in acidic soils (pH 5.9) which increased the absorption of Mn in the roots of avocado tree which corrected Mn deficiency in clay soils (Tisdale *et al.*, 1985).

(xiv) Zinc

Available zinc of 7.64 ppm in Kimorok was significantly higher than those of Solit, Kapkun and Maoi (1.73, 6.00 and 5.06 ppm), respectively (P < 0.05, Table 4.2 and Figure 4.14). High level of available zinc in Kimorok was ascribed to acid sandy clay soils (pH 6.05) which increased Zn²⁺ ions in soil solution that had adsorbed on clay surfaces and the negative charge of soil organic matter to form stable complexes in the soils.





Available zinc in Kapkun and Maoi were 6.00 and 5.06 ppm, this was because of high levels of nitrogen and potassium (0.3% and 1.58 cmol (⁺)/kg) which inhibited the availability of zinc in the soils. The level of available zinc in Solit was low on the basis of soil pH (7.05). As soil pH increases, Zn^{+2} precipitates as $ZnFe_2O_4$ or $ZnSiO_4$ and is unavailable for plant uptake. High level of Mn (263 ppm) may also have interacted with Zn and other cations to form strong complexes on the exchange sites of soil organic matter in the soils (Jones, 1972). Available zinc was low in all the sites, as a result of high soil pH (6.0 – 7.05). At high pH, zinc precipitate to form insoluble complexes of zinc silicates in the soil, since the solubility of zinc is highly dependent upon soil pH and its availability decreases at high pH in the soil solution (Tisdale *et al.*, 1985).

4.2.1 Characterization Criteria for soils

The criteria used to characterise the soils of the study sites were soil types shown on the soil map, soil texture, levels of nutrients in soils and two varieties of *Acacia senegal*.

The soil map on characterization of study sites is given in Figure 4.15.



A Soil map of Study Sites - Baringo District

Figure 4.15: Soil Map on Characterization of Study Sites

The soil map shows the locations, colour representation and delineation boundaries of soil types of Solit, Kimorok, Kapkun and Maoi (Figure 4.15). Soil texture in Kimorok and Kapkun was sandy clay loam with ratio of sandy clay to silt clay of 2:3 and 1:2, while Solit and Maoi was sandy loam with ratio of sandy clay to silt clay of 1:5 and 2:1, respectively.

The soils of Solit were characterized as neutral sandy loam, well drained, shallow, rocky and stony calcic Xerosols with nitrogen, phosphorus, calcium, magnesium and copper of 0.18%, 11.54 ppm, 17.33 cmol (⁺)/kg, 6.08cmol(⁺)/kg and 1.67 ppm, respectively (Table 4.2). Soils in Kimorok were acidic sandy clay loam, well drained, shallow, rocky and stony Lithosols and Xerosols with soil organic matter, carbon, nitrogen, phosphorus, potassium and zinc of 1.97%, 1.15%, 0.14%, 14.6 ppm, 1.24 cmol (⁺)/kg and 7.64 ppm, respectively.

In Kapkun, the soils were acidic sandy clay loam, well drained, shallow, rocky and stony ferralic Cambisols with soil organic matter, nitrogen, phosphorus, potassium, calcium, iron and manganese (1.73%, 0.3%, 14.10 ppm, 1.58 cmol (⁺)/kg, 6.40 cmol (⁺)/kg, 250 ppm, 344 ppm), respectively.

Maoi had acidic sandy loam, well drained, shallow, rocky and stony ando-chromic Cambisols with calcic Xerosols with levels of soil organic matter, nitrogen, phosphorus, sodium, iron and manganese (1.15%, 0.17%, 13.90, 1.17 cmol (⁺)/kg, 287 and 383 ppm), respectively (Table 4.2).

Two varieties of Acacia senegal; Acacia senegal variety senegal and Acacia senegal variety kerensis were also used to characterize the study sites. Acacia senegal variety senegal was dominant in calcic Xerosols and ferralic Cambisols with calcic Xerosols of Solit and Kapkun, while Lithosols and Xerosols and ando-chromic Cambisols with calcic xerosols of Kimorok and Maoi had only Acacia senegal variety kerensis and other closely related ^{species}.

43 Soil Fertility Status in the Study Sites

The nutrient levels were used to rate the soil fertility status of the study sites in relation to occurrence of *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis* according to the method of Landon (1991).

In Solit, *Acacia senegal* variety *senegal* grows well in neutral, sandy loam, calcic Xerosols, low to medium fertile with high levels of nitrogen, calcium, magnesium and copper and low levels of soil organic matter, C: N ratio, carbon, potassium, sodium, iron, manganese and zinc, respectively (Table 4.2).

The soils of Kimorok were acidic, sandy clay loam, Lithosols and Xerosols which sustained the growth of *Acacia senegal* variety *kerensis*, medium to high fertile with high levels of carbon, C: N ratio, soil organic matter, phosphorus, potassium, calcium and zinc, and low levels of nitrogen, magnesium, sodium, copper, iron and manganese.

Acacia senegal variety senegal in Kapkun grows well in acidic, sandy clay loam, ferralic Cambisols with calcic Xerosols, medium to high fertile with high levels of carbon, C: N ratio, soil organic matter, phosphorus, potassium, calcium and zinc, and low levels of nitrogen, magnesium, sodium, copper, iron and manganese, respectively.

In Maoi, the soils were acidic, sandy loam, ando-chromic Cambisols with calcic xerosols which supported the growth of *Acacia senegal* variety *kerensis*, low to medium fertile with ^{high} levels of nitrogen, calcium, magnesium and copper, and low levels of soil organic ^{matter}, C: N ratio, carbon, potassium, sodium, iron, manganese and zinc, respectively.

Gum arabic, a natural exudate obtained from branches and stems of Acacia senegal variety senegal and Acacia senegal variety kerensis is produced and the trees grow in abundance and occupies over 80 % of the vegetation cover in the study area.

4.4 Conclusions

All the soil types of the study sites had high sand content and low soil moisture and high levels of calcium and magnesium. The effect of low soil moisture regimes in sandy soils limits the nutrient uptake by plants and the physiological growth of plants and conditions of soil microorganisms in arid and semi-arid environments.

Acacia senegal variety senegal grows well in neutral, calcic Xerosols and acidic, ferralic Cambisols with high levels of nitrogen, potassium, calcium, magnesium and copper, respectively.

Acacia senegal variety kerensis grows well in acidic Lithosols and Xerosols and andochromic Cambisols with ealcic Xerosols in high levels of carbon, phosphorus, iron, manganese and zinc, respectively.

Gum arabic, a natural exudate obtained from branches and stems of *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis* was produced in the study area.

CHAPTER FIVE

STUDIES ON MICRONUTRIENTS UPTAKE BY ACACIA SENEGAL VARIETIES IN A SIMULATED ARID AND SEMI-ARID ENVIRONMENT

51 Determination of Micronutrients Uptake by Acacia senegal varieties

Studies on micronutrients uptake by two *Acacia senegal* varieties using normal, low and high dosage treatments in soil and vermiculite media are given in Tables 5.1- 5.7 and Figures 5.1 - 5.12, respectively. The uptake of copper, iron, manganese and zinc by *Acacia senegal* seedlings, heights (Ht) and diameter at ground level (Dgl) and their relationships are discussed. The concentrations of micronutrients uptake by tree seedlings in soil and vermiculite media are given in Table 5.1 and 5.2.

(a) Uptake of Micronutrients on Normal and Dosage treatments in Soil Medium

(i) Copper

Uptake of copper (102.7 ppm) on dosage treatments in Solit was significantly higher (P < 0.05) than those of Maoi, Kimorok and Kapkun (41.4, 33.4 and 30.1 ppm), respectively (Table 5.1). This was because of *Acacia senegal* variety *senegal* that seemed to take higher levels of available copper than *Acacia senegal* variety *kerensis* from the soil than other micronutrients.

Indie 5.1: Uptake of Micronutrients by Acacia senegal seedlings, Heights and Diameter at

on Normal and Dosage Treatments in Soil Medium

Site	Variety	Medium	Cu (ppm)	Fe (ppm)	Mn (ppm)	Zn (ppm)				
			n=36	n=36	n=36	n=36				
		1	Normal Treatments							
			means±S.E.	means ± S.E.	means± S.E.	means±S.E.				
Solit	senegal	Soil	31.4 <u>+</u> 13.7	595.5+227.6	264.6 <u>+</u> 24.8	94.4+22.8				
Kapkun	senegal	Soil	155.6 <u>+</u> 13.7	499.2 <u>+</u> 227.6	339.2 <u>+</u> 24.8	132.3 <u>+</u> 22.8				
Kimorok	kerensis	Soil	126.9 <u>+</u> 13.7	388.7+227.6	409.9+24.8	107.1 <u>+</u> 22.8				
Maoi	kerensis	Soil	87.1 <u>+</u> 19.4	399.8 <u>+</u> 321.9	342.1 <u>+</u> 35.1	68.5 <u>+</u> 32.2				
				Dosage Treat	tments					
Solit	senegal	Soil	102.7 <u>+</u> 6.6	552.8 <u>+</u> 46.4	300.5 <u>+</u> 93.9	422.8 <u>+</u> 55.7				
Kapkun	senegal	Soil	30.1 <u>+</u> 6.5	653.5 <u>+</u> 45.7	638.3 <u>+</u> 92.4	274.9+54.8				
Kimorok	kerensis	Soil	33.4+5.5	366.0 <u>+</u> 38.5	307.1 <u>+</u> 78.0	533.2.0 <u>+</u> 46.2				
Maoi	kerensis	Soil	41.4 <u>+</u> 9.0	233.8.6 <u>+</u> 62.7	519.7 <u>+</u> 126.8	200.2 <u>+</u> 75.2				

Footnote: n = number of samples analysed

S.E. = + Standard error

Copper uptake (155.6 ppm) in soil medium on normal treatments in Kapkun was significantly higher than the quantities of Kimorok, Solit and Maoi (126.9, 31.4 and 87.1 ppm), respectively (P < 0.05). This was attributed to acidic sandy clay loam soils (pH 5.95) with high soil organic matter (1.97%) that increased the availability of copper to seedlings of *Acacia senegal* variety *senegal* (Parker and Walker, 1986).

Copper uptake in soil medium on normal treatments in Kimorok and Maoi was 126.9 and 87.1 ppm and was significantly higher than 33.4 and 41.4 ppm in dosage treatments, respectively (P < 0.05). This was as a result of high levels of soil organic matter (1.97 and 1.15%) which enhanced the availability of copper ions to tree seedlings in the soil solution. The copper uptake (33.4 and 41.4 ppm) on dosage treatments was low because of high levels of available iron and manganese (287.20 and 383.23 ppm) that reduced the availability of copper ions in the soils. This indicates that *Acacia senegal* variety *kerensis* seems to take low concentrations of available Cu from the soil.

(ii) Iron

Iron uptake in soil medium on normal treatments (595.5 and 499.2 ppm) in Solit and Kapkun were significantly higher than 388.7 and 399.8 ppm in Kimorok and Maoi, respectively (P < 0.05, Table 5.1). This was ascribed to acidic sandy clay loam soils (pH 5.96) with high soil organic matter (1.73%) that increased the availability of iron to seedlings of *Acacia senegal* variety *senegal*. This shows that *Acacia senegal* variety *senegal* seems to take low concentrations of available Fe from the soil.

The uptake of iron (653.5 ppm) on dosage treatments in Kapkun was significantly higher than those of Maoi, Kimorok and Solit (233.8, 366.0 and 552.8 ppm), respectively (P < P

0.05). This was as a result of sandy clay soils with high level of soil organic matter (1.73%), which increased iron ions that become available for uptake by tree seedlings in the soil solution. The high level of iron ions was also attributed to soil pH (5.96). At this pH iron becomes more available in low soil reaction, since the solubility of manganese is highly dependent upon soil pH (Woolhouse, 1983). Iron uptake (126.9 and 87.1 ppm) in Kimorok and Maoi, in soil medium on dosage treatments was low because of high levels of phosphorus (14.6 and 13.9 ppm, respectively) that interacted in the soil solution with iron ions that reduced its availability for the uptake by tree seedlings.

(iii) Manganese

In Kimorok, uptake of manganese (409.9 ppm) in soil medium on normal treatments was significantly higher than concentrations in Solit, Kapkun and Maoi (264.6, 339.2 and 342.1 ppm) respectively (P < 0.05, Table 5.1). This was as a result of soil pH (6.05). At this pH Mn becomes more available in acidic soils, and since the solubility of manganese in soil solution is highly dependent upon soil pH (Woolhouse, 1983).

In dosage treatments for Kapkun, manganese uptake of 638.3 ppm was significantly higher than those of Solit, Kimorok and Maoi (300.5, 307.1 and 519.7 ppm), respectively (P < 0.05). This was as a result of acid sandy clay soils with high soil organic matter (1.73%) that increased the availability of manganese to tree seedlings. This shows that *Acacia senegal* variety *senegal* tends to take higher concentrations of available Mn from the soil than other micronutrients.

Manganese uptake in soil medium on dosage treatments (300.5 ppm) at Solit was significantly lower than the quantities of Kimorok, Maoi and Kapkun (307.1, 519.7 and

638.3 ppm), respectively (P < 0.05). This was low due to neutral calcic Xerosols (pH 7.05). At soil pH 7.05 available Mn becomes immobile as a result of low solubility in the soil, since the solubility of manganese is highly dependent upon soil pH (Woolhouse, 1983).

Uptake of manganese decreases with high levels of soil pH, available phosphorus and copper (7.05, 11.4 and 1.67 ppm) in soils which made it unavailable to tree seedlings. Mn uptake was considerably affected by competitive interaction with high level of calcium (17.33 cmol (⁺)/kg) which decreased its availability to tree seedlings. Manganese uptake of 519.7 ppm in soil medium on dosage treatments at Maoi was high as a result of high level of available Mn and soil organic matter (383 ppm and 1.15%) in the soils that increased its availability to tree seedlings.

(iv) Zinc

Zinc uptake in soil medium on normal treatments in Kapkun and Kimorok (132.30 and 107.1 ppm) was significantly higher than those of Maoi and Solit (68.5 and 94.4 ppm), respectively (P < 0.05, Table 5.1). This was because of high level of soil organic matter (1.97 and 1.73%) which adforbed zinc ions in the soil solution that released to tree seedlings. Zinc uptake (68.5 ppm) in soil medium on normal treatments at Solit was low on the basis of neutral calcic Xerosols (pH 7.05). At this pH, available Zn becomes immobile because of low solubility in the soil, since the solubility of zinc is highly dependent upon soil pH (Woolhouse, 1983). Zinc uptake decreases with high levels of available copper at high pH in the soils. The uptake of zinc (94.4 ppm) on normal treatments in Maoi was low in a result of high level of available phosphorus (13.9 ppm) in the soils which may have buduced the zinc uptake by the seedlings of *Acacia senegal* variety *kerensis*.

The uptake of zinc (533.2 ppm) in dosage treatments in Kimorok was significantly higher than 200.2, 274.9 and 422.8 ppm in Maoi, Kapkun, and Solit, respectively (P < 0.05). This was as a result of acid sandy clay soils with high soil organic matter (1.73%) that increased the availability of zinc to be more available to tree seedlings. This shows that *Acacia senegal* variety *senegal* tends to take higher concentrations of available Zn from the soil than other micronutrients.

Uptake of zinc by tree seedlings in soil medium on dosage treatments in Maoi and Kapkun (200.2 and 274.9 ppm) was significantly lower (P < 0.05) than 422.8 ppm in Solit, respectively. This was low on the basis of high levels of available phosphorus (13.9 and 14.05 ppm) that interacted with available Zn that made it unavailable to tree seedlings (Woolhouse, 1983). Uptake of zinc (422.8 ppm) in soil medium on dosage treatments in Solit was high as a result of low level of available Fe (151 ppm). The available iron ions interacted with Zn in the soils with soil organic matter (0.78%) that enhanced its availability to tree seedlings.

Table 5.2: Uptake of Micronutrients by Acacia senegal seedlings, Heights and Diameter on Normal and Dosage Treatments in Vermiculite Medium

Site	Variety	Medium	Cu (ppm)	Fe (ppm)	Mn (ppm)	Zn (ppm)	1
			n=36	n=36	n=36	n=36	r
			/	Normal Tre	eatments		_
		27	r means±S.E.	Means ± S.E.	means± S.E.	means±S.E.	ľ
Solit	senegal	Verm	29.8+13.70	383.2+227.60	114.5+24.80	86.6 <u>+</u> 22.80	1
Kapkun	senegal	Verm	39.9 <u>+</u> 13.70	291.8 <u>+</u> 227.60	95.3 <u>+</u> 24.80	112.5 <u>+</u> 22.80	1
Kimorok	kerensis	Verm	33.8 <u>+</u> 14.10	376.2+234.20	161.0 <u>+</u> 25.60	81.8 <u>+</u> 23.50	1
Maoi	kerensis	Verm	87.7 <u>+</u> 13.70	672.1±59.20	157.6 <u>+</u> 242.80	153.4 <u>+</u> 22.80	1
				Dosage Trea	atments		
Solit	senegal	Verm	49.0 <u>+</u> 6.30	547.1+44.30	468.9 <u>+</u> 89.70	284.3 <u>+</u> 53.20	1
Kapkun	senegal	Verm	46.0 <u>+</u> 6.30	353.3 <u>+</u> 44.30	422.9 <u>+</u> 89.70	356.0 <u>+</u> 53.20	1
Kimorok	kerensis	Verm	55.9 <u>+</u> 7.30	485.1 <u>+</u> 50.70	499.60 <u>+</u> 97.32	621.6 <u>+</u> 60.80	1
Maoi	kerensis	Verm	65.0 <u>+</u> 9.00	405.7 <u>+</u> 62.70	264.1+126.80	594.9 <u>+</u> 75.20	1
Footn	ote: n	= num	ber of sample	es analysed; S. 87	E. = Stand	ard error; Vo	eri

(b) Uptake of Micronutrients on Normal and Dosage treatments in Vermiculite Medium

(i) Copper

In vermiculite medium, copper uptake (87.7 ppm) on normal treatments in Maoi was significantly higher than for Kapkun, Solit and Kimorok (39.9, 29.8 and 33.8 ppm), respectively (P < 0.05, Table 5.2). In Kimorok, Kapkun and Solit, copper uptake of 33.8, 39.9 and 29.8 ppm were not significantly different from each other. This was as a result of *Acacia senegal* variety *kerensis* that had an affinity for copper uptake in the medium. The effect of vermiculite medium showed that copper uptake increased growth of tree seedlings of *Acacia senegal* variety *kerensis*. This effect confirmed that vermiculite is a good medium for determining the levels of micronutrient uptake by each variety of *Acacia senegal*. Growth of seedlings in vermiculite medium indicates that *Acacia senegal* variety *kerensis* in Maoi tends to take high concentrations of copper in the nutrient solutions. In Maoi, copper uptake (65.0 ppm) in vermiculite medium on dosage treatments was significantly higher (P < 0.05) than those of Kimorok, Solit and Kapkun (39.9, 29.8 and 33.8 ppm), respectively. This reason was similar to that of normal treatments.

(ii) Iron

In Maoi, iron uptake (672.08 ppm) in vermiculite on normal treatments was significantly higher than those of Solit, Kapkun and Kimorok (383.2, 291.8 and 376.18 ppm), respectively (P < 0.05, Table 5.2). This was as a result of the effect of vermiculite medium, a homogenous inert medium which indicated that *Acacia senegal* variety *kerensis* had high affinity for iron uptake in the nutrient solutions. The level of iron uptake (547.1 ppm) in Vermiculite medium on dosage treatments at Solit was higher than those of Kimorok, Maoi and Kapkun (485.1, 405.7 and 353.3 ppm), respectively. This was attested to *Acacia senegal* variety *senegal* that had high tendency for iron uptake in the nutrient solutions. Vermiculite medium also revealed the interaction effects of high levels of Cu (102.7 ppm) in the soil medium on dosage treatments through competitive interactions with cations of iron and copper. This inhibited the uptake of iron that led to decrease in the growth of tree seedlings from 169.7 mm to 168.3 mm and also concentrations of copper from 595.5 to 552.8 ppm (Table 5.2). This also reveals that *Acacia senegal* variety *senegal* take higher levels of available iron than *Acacia senegal* variety *kerensis* from the soil than other micronutrients.

(iii) Manganese

In Kimorok, manganese uptake (161.0 ppm) in vermiculite on normal treatments was significantly higher than those of Kapkun Solit, and Maoi (95.3, 114.5 and 157.6 ppm), respectively (P < 0.05, Table 5.2). This was on the basis of *Acacia senegal* variety *kerensis* that had high affinity for manganese in the nutrient solutions.

Uptake of Mn (468.9 ppm) in vermiculite medium on dosage treatments at Solit was significantly higher than those of Maoi, Kimorok and Kapkun (64.1, 400.6 and 422.9 ppm), respectively (P < 0.05, Table 5.2). This was as a result of similar cation ratio Cu: Mn (1:1) which interacted with Mn ions because of similar ionic radii with Cu (in the oxidation state of Cu²⁺ and Mn²⁺) that increased the uptake of manganese by tree seedlings. Manganese uptake (400.6 and 422.9 ppm) in vermiculite at Kimorok and Kapkun was also high because of high levels of iron (485.1 and 353.3 ppm) which may have enhanced the availability of manganese uptake by seedlings of *Acacia senegal* variety *kerensis* and *Acacia senegal* variety *senegal*.

Vermiculite medium also showed consistent growth pattern on the concentrations of uptake of micronutrients by seedlings of *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis*.

(iv) Zinc

Zinc uptake (153.4 ppm) in vermiculite medium on normal treatments in Maoi was significantly higher (P < .05) than those of Kapkun, Kimorok and Solit (112.5, 81.8 and 86.6 ppm), respectively (Table 5.2). This was because of high level of iron and manganese that enhanced the availability of zinc which indicates *Acacia senegal* variety *kerensis* to tend to take high level of iron in the nutrient solutions.

The uptake of zinc (621.6 ppm) in dosage treatments in Kimorok was significantly higher (P < 0.05) than those in Maoi, Kapkun, and Solit (594.9, 356.0 and 284.3 ppm), respectively. This was as a result of high level of magnesium that enhanced zinc availability and the effect of vermiculite medium, a homogenous inert medium with no organic matter apart from the applied nutrients. This revealed that *Acacia senegal* variety *kerensis* had high affinity for zinc uptake.

(b) Heights and diameter at ground level (Dgl) on Normal and Dosage treatments in Soil and Vermiculite media

(i) Heights of Tree seedlings on Normal treatments in Soil and Vermiculite medium

The heights and diameter at ground levels (Dgl) of *Acacia senegal* seedlings on normal and dosage treatments in soil and vermiculite media are given in Tables 5.1 and 5.2. Comparisons of heights on normal treatments in soil and vermiculite media are showed in Figure 5.1.



Figure 5.1 Comparisons of Heights on Normal treatments in Soil and Vermiculite media

The height of *Acacia senegal* seedlings in soil medium of 232.4 mm in Kimorok was higher than those of Kapkun, Solit and Maoi (158.0, 169.7 and 217.8 mm), respectively (P < 0.05). This was as a result of high level of soil organic matter (1.97%) that increased the growth of tree seedlings. In vermiculite medium, Kapkun had 121.8 mm higher than those In Maoi, Kimorok and Solit (106.1, 109.8 and 111.4 mm), respectively. This was attributed to high levels of Fe and Zn (291.8 and 112.5 ppm) that enhanced the growth of seedlings.

The heights in soil medium in all the sites were higher than those in vermiculite medium. This was because soil, a heterogeneous medium had inherent nutrients while vermiculite was homogeneous inert medium.

(ii) Heights of Tree seedlings on Dosage treatments in Soil and Vermiculite medium

Figure 5.2 shows the comparisons of heights of *Acacia senegal* seedlings on dosage treatments in soil and vermiculite media in the sites.



Figure 5.2 Comparisons of Heights on Dosage treatments Soil and Vermiculite media

The height of seedlings in Maoi (201.1 mm) on dosage treatments in soil medium was higher (P < 0.05) than those of Kapkun, Solit and Kimorok (170.1, 168.3 and 159.0 mm), respectively. This was as a result of high level of soil organic matter (1.15%) that increased the growth of tree seedlings. In vermiculite medium, the height of tree seedlings in Kimorok (180.1 mm) was higher (P < 0.05) than 116.8, 169.2 and 139.2 mm in Maoi, Kapkun and Solit, respectively. This was attested to *Acacia senegal* variety *kerensis* that
tends to take high level of Zn (621.6 ppm) in the nutrient solutions that enhanced the growth of seedlings. The heights of tree seedlings in soil medium in Maoi, Kapkun and Solit (except in Kimorok) were higher than in vermiculite medium. This is because soils (heterogeneous medium) have inherent nutrients while vermiculite has homogeneous inert medium.

(c) Comparisons of Heights on Normal treatments with Low Dosage treatments

The comparisons micronutrients uptake by tree seedlings for heights on copper normal treatments with low dosage treatments is given in Table 5.3.

Site	Cu (ppm) n=36	Fe (ppm) n=36	Mn (ppm) n=36	Zn(ppm) n=36
	ľ	ormal Treatment	s (Heights (mm))	
	Means \pm S.E.	Means ± S.E.	Means ± S.E.	Means \pm S.E.
Solit	144.94±10.58	144.94±10.58	144.94±10.58	144.94±10.58
Kapkun	140.71±9.86	140.71±9.86	140.71±9.86	140.71±9.86
Kimorok	184.63±13 =1 8	184.63±13.18	184.63±13.18	184.63±13.18
Maoi	162.61±10.58	162.61±10.58	162.61±10.58	162.61±10.58
	Ι	Low Dosage Treat	tments	
Solit	206.69±10.58	172.85±11.59	170.42±11.59	172.20±10.58
Kapkun	149.20±11.60	159.20±10.01	172.88±9.86	154.71±9.86
Kimorok	162.11±13.22	177.91±11.61	160.67±13.20	179.15±11.60
Maoi	155.74±10.58	168.41±11.60	161.54±10.58	186.32±11.59

Table 5.3 (Comparisons	of Heights	on Normal	treatments wit	h Low	Dosage	treatments
-------------	-------------	------------	-----------	----------------	-------	--------	------------

Footnote: n= number of samples analysed

S.E. = Standard error

(i) Comparisons of Heights on Cu Normal treatments with Low Cu Dosage

Heights of *Acacia senegal* seedlings in low dosage of Cu (206.69 mm) in Solit was significantly higher than heights of seedlings with copper in normal treatments (144.94 mm) for Kapkun, Kimorok and Maoi (149.2, 162.11 and 155.74 mm), respectively (P < 0.05, Table 5.3, Figure 5.3 and Appendix II).



Figure 5.3 Comparisons of Heights on Cu Normal treatments with Low Cu Dosage This was because of high level of iron (653.51 ppm) which interacted with copper ions in the soil solution. This interactive effect may have been suppressed by the high levels of calcium and magnesium (17.33 and 6.08 cmol (+)/kg) in the nutrient solutions. This enhanced the growth of the tree seedlings to height of 206.69 mm (Table 5.3). The copper and iron interaction in Solit was higher than in Kapkun, Kimorok and Maoi with Cu/Fe cation ratios (Solit: Kapkun 15:1, Solit: Kimorok 4:1, and Solit: Maoi 6:1). High levels of ^{Iron} ions (Fe³⁺) in the oxidation state of +3 in the soil solution may have reduced high levels of available calcium and magnesium to release copper ions for uptake by plants (Tisdale *et al.*, 1985).

(ii) Comparisons of Heights on Iron Normal with Low Fe treatments

The heights of *Acacia senegal* seedlings in low Fe dosage (177.91 mm) in Kimorok was significantly higher than heights of seedlings in Solit, Kapkun and Maoi (172.85, 159.2 and 168.41 mm), respectively, compared with iron normal treatment of 140.71 mm (P < 0.05, Figure 5.4).



Figure 5.4 Comparisons of Heights on Fe Normal treatments with low Fe Dosage

The height of tree seedlings in Kimorok attained 177.91 mm on the basis of high level of available Zn (7.50 ppm) in the soils that interacted with iron ions of similar ionic radii $(Fe^{3+} = 0.73 \text{ and } Zn^{2+} = 0.83 \text{ nm})$ in the soil solution that enhanced Fe uptake by tree seedlings.

(iii) Comparisons of Heights on Manganese Normal with Low Mn treatments

The heights of tree seedlings in Kapkun on low Mn dosage treatments attained 172.88 mm com pared with 140.71 mm on normal treatments. This was higher than those of Solit, Kimor ok and Maoi (170.42, 160.67 and 161.54 mm), respectively ((P < 0.05, Figure 5.5 and Appen dix II).



Figure 5.5 Comparisons of Heights on Mn Normal treatments with low Mn Dosage

This was as a result of acid sandy clay soils (pH 5.96) with high soil organic matter (1.73%) that increased the availability of manganese to be more available for uptake by tree seedlings.

(iv) Comparisons of Heights on Zinc Normal with Low Zn treatments

Heights of *Acacia senegal* seedlings on low Zn dosage (186.3 mm) in Maoi compared with 152.6 mm on normal treatments was higher than those of Kimorok, Kapkun and Solit (179.2, 154.7 and 172.2 mm), respectively (P < 0.05, Figure 5.6).



Figure 5.6 Comparisons of Heights on Zn Normal treatments with low Zn Dosage

Heights of tree seedlings (186.3 and 179.2mm) in Maoi and Kimorok were high on the basis of available Zn at low soil pH (6.2 and 6.0). At this pH Zn becomes more available to meedlings of *Acacia senegal* variety *kerensis*. These results agree with work by Parker and Walker (1986) which states that the solubility of zinc is highly dependent upon soil pH and its availability decreases at high pH in the soil solution. At Kapkun, the heights of *Acacia senegal* variety *senegal* seedlings were low as a result of high level of nitrogen (0.3%) in the soils which reduced the availability of Zn uptake. *Acacia senegal* variety *kerensis* in Kimorok and Maoi (186.3 and 179.2mm) attained higher heights in soil medium than *Acacia senegal* variety *senegal* in Solit and Kapkun (172.2 and 154.71 mm), respedtively. This was attested to *Acacia senegal* variety *kerensis* that seemed to take higher levels of ^{available} zinc than *Acacia senegal* variety *senegal* from the soil than other micronutrients. (d) Comparisons of Heights on Normal treatments with High Dosage treatments The comparisons micronutrients uptake by tree seedlings for heights on copper normal treatments with high dosage treatments is given in Table 5.4 (See Appendix II).

Site	Cu (ppm) n=36	Fe (ppm) n=36	Mn (ppm) n=36	Zn (ppm) n=36
	P	ormal Treatment	s (mm)	
	Means ± S.E.	Means ± S.E.	Means ± S.E.	Means \pm S.E.
Solit	144.94±10.58	144.94±10.58	144.94±10.58	144.94±10.58
Kapkun	140.71±9.86	140.71±9.86	140.71±9.86	140.71±9.86
Kimorok	184.63±13.18	184.63±13.18	184.63±13.18	184.63±13.18
Maoi	162.61±10.58	162.61±10.58	162.61±10.58	162.61±10.58
	Ĭ	High Dosage Trea	tments	
Solit	160.63±10.58	189.37±11.60	155.41±10.58	145.98±10.57
Kapkun	163.58±9.860	168.21±9.86	160.25±9.86	148.92±9.86
Kimorok	172.03±11.642	161.52±13.23	154.05±13.18	178.40±11.60
Maoi	167.61±13.75	130.16±11.61	145.48±10.58	158.46±10.58

Table 5.4	Com	parisons	of Heights	s on Norma	l treatments with	High I)osage (treatments
			or recently					

Footnote:

n= number of samples analysed

S.E. = Standard error

(i) Comparisons of Heights on Copper Normal with High Cu treatments

The height of tree seedlings on high dosage of Cu in Kimorok had 172.03 mm compared with 144.94 mm of copper at normal treatment. This was significantly higher than those of Kapkun, Solit and Maoi (163.58, 160.63 and 167.61 mm), respectively (P < 0.05, Figure 5.7 and Appendix II).



Figure 5.7 Comparisons of Heights on Cu Normal treatments with High Cu Dosage This was because of high level of soil organic matter (1.97%) mineralized through microbial decomposition process to release available Cu ions in the soil solution which increased the height of tree seedlings to 172.03 mm. The height of tree seedlings in Maoi on high copper dosage (167.61 mm) was significantly higher than those of Kapkun and Solit (163.58 and 160.63 mm), respectively (P < 0.05). This was attributed to 6.2 pH. At this pH, copper ions in the soil solution were oxidized to Cu⁺² and becomes more available for uptake by the seedlings.

(ii) Comparisons of Heights on Iron Normal with High Fe treatments

The height of tree seedlings on high Fe dosage in Solit had 189.37 mm compared with 140.71 mm on normal treatments. This was significantly higher than those of Kapkun, Kimorok and Maoi (168.21, 161.52 and 130.16 mm), respectively (P < 0.05, Figure 5.8).



Figure 5.8 Comparisons of Heights on Fe Normal treatments with high Fe Dosage

This was as a result of *Acacia senegal* variety *senegal* that showed a tendency to take higher levels of available iron than *Acacia senegal* variety *kerensis* from the soils. Tree seedlings ^{on} high iron dosage (130.16 mm) in Maoi were significantly lower than those of Solit, Kapkun and Kimorok (189.37, 168.21 and 161.52 mm), respectively (P < 0.05).

This was ascribed to acidic soils (pH 6.2) with high levels of nitrogen and phosphorus (0.17% and 13.9 ppm) which reacted with iron, forming a precipitate of iron ammonium phosphate (Fe (NH₄) PO₄.H₂O)) in the soil solution. This made iron less available for uptake by tree seedlings.

(iii) Comparisons of Heights on Manganese Normal with high Mn treatments

The height of Acacia senegal seedlings on high Mn dosage (160.25 mm) in Kapkun was significantly higher than those of Solit, Kimorok and Maoi (155.41, 154.05 and 145.48 mm), respectively (P < 0.05, Figure 5.9).





This was on the basis of soil pH 5.96. As soil pH decreases, Mn^{+2} become more available for uptake by tree seedlings. High levels of calcium and magnesium (6.38 and 3.96 cmol (+)/kg) may have interacted with manganese ions in the soil solution which made Mn less available for uptake by tree seedlings.

(iv) Comparisons of Heights on Zinc Normal with High Zn treatments

At Kimorok, the height of seedlings on high Zn dosage was 178.4 mm compared with zinc normal treatments of 152.6 mm. This was significantly higher than those of Solit, Kapkun and Maoi (146.0, 148.9 and 158.5 mm), respectively (P < 0.05, Figure 5.10).





The heights of tree seedlings on high Zn dosage treatments in Kimorok and Maoi were 178.4 and 158.5 mm, respectively. This was as a result of available zinc that hydrolysed as Zn^{2+} ions in the soil solution at low soil pH (6.19 and 6.05), since the solubility of zinc is highly dependent upon soil pH. At low pH, Zn becomes more available to seedlings of *Acacia senegal* variety *kerensis*.

At Solit and Kapkun, the heights of the seedling were low as a result of the presence of high levels of calcium and phosphorus (17.33 cmol ($^+$)/kg and 14.4 ppm) in the soils which decreased the availability of zinc for uptake by seedlings of *Acacia senegal* variety *senegal*.

(e) Diameter at ground level (Dgl) on Uptake of Micronutrients on Normal and Dosage treatments

(i) Diameter at ground level of Tree seedlings on Normal treatments in Soil and Vermiculite media

The diameter at ground level of tree seedlings is a horizontal growth that increases the basal diameter during uptake of nutrients. Diameter at ground levels (Dgl) of *Acacia senegal* seedlings on normal and dosage treatments in soil and vermiculite media are given in Tables 5.1 and 5.2. Comparisons of Dgl on normal and dosage treatments in soil and vermiculite media are showed in Figures 5.11 and 12.



Figure 5.11 Comparisons of Diameter at ground level on Normal treatments in Soil and Vermiculite media

The diameter at ground levels (Dgl) of *Acacia senegal* seedlings on normal treatments in soil and vermiculite media in Solit, Kimorok, Maoi and Kapkun were 6.0, 5.9, 5.7 and 5.4 mm and 4.6, 4.1, 4.3 and 4.2 mm, respectively. These were not significantly different from each other at P > 0.05. This was as a result of slow growth rate of *Acacia senegal* variety *kerensis* and *Acacia senegal* variety *senegal* as adaptation mechanism to survive in low soil moisture regimes under dryland conditions. The diameter at ground level of seedlings in soil medium in all the sites were higher than in vermiculite medium as a result of soil that had inherent ^{nutrients} while vermiculite was an inert medium. (ii) Diameters at ground level of Tree seedlings on Dosage treatments in Soil and Vermiculite media

Comparisons of diameter at ground level on dosage treatments in soil and vermiculite media are given in Figure 5.12.





Vermiculite media

The diameter at ground level of seedlings in Maoi of 7.3 mm on dosage treatments in soil medium was higher than 6.0, 6.1 and 5.5 mm in Kapkun, Solit and Kimorok, respectively (P < 0.05). This was as a result of high level of soil organic matter (1.15%) that increased the growth.

There were no significant differences in Dgl of tree seedlings in Solit, Kimorok and Kapkun (5.8, 5.8 and 5.7 mm), respectively, except in Maoi (5.2 mm) in vermiculite medium. These results indicate that diameter at ground level of seedlings is not a good indicator for measuring the growth performance of the *Acacia senegal* varieties.

The diameters at ground level of tree seedlings in Maoi, Kapkun and Solit in soil medium were higher than those of vermiculite medium, but low in Kimorok as a result of soils, a heterogeneous medium, that had inherent nutrients while vermiculite was homogeneous inert medium. (f) Comparisons of Diameter at ground level on Copper Normal treatments with Low Cu treatments

The diameters at ground level of *Acacia senegal* seedlings for normal and dosage treatments of micronutrients uptake are given in Table 5.5.

	Diameter at Gro	ound level (Dgl)		
Site	Cu (ppm) n=36	Fe (ppm) n=36	Mn (ppm) n=36	Zn (ppm) n=36
	Normal Treatm	ents (mm)		
	mm±S.E.	mm±S.E.	mm±S.E.	mm±S.E.
Solit	6.14±0.53	5.91± 0.49	6.64±0.66	5.61±0.53
Kapkun	6.14±0.53	5.91± 0.49	6.64±0.66	5.61±0.53
Kimorok	6.14±0.53	5.91± 0.49	6.64±0.66	5.61±0.53
Maoi	6.14±0.53	5.91± 0.49	6.64±0.66	5.61±0.53
	Low Dosage Tr	eatments		• • • • • • • • • • • • • • • • • • •
Solit	6.74±0.53	6.60± 0.58	4.88± 0.58	5.71±0.53
Kapkun	4.85±0.58	5.21±0.51	5.20±0.49	5.65±0.49
Kimorok	5.33±0.66	5.78±0.58	6.48±0.66	5.05±0.58
Maoi	5.71±0.53	5.45±0.58	5.80±0.53	5.25±0.58
	High Dosage Tr	reatments		
Solit	5.95±0.53	5.86±0.58	6.31±0.53	5.60 ±0.53
Kapkun	5.80±0.49	5.75±0.49	8.10±0.49	5.16±0.49
Kimorok	5.89±0.58	5.86±0.66	5.95±0.66	5.65±0.58
Maoi	4.91±0.69	5.18±0.58	5.63±0.53	5.35±0.53

	Table 5.5:]	Diameter at G	Fround level of	Tree seedlings	for Normal an	d Dosage 🕽	Freatments
--	--------------	---------------	-----------------	----------------	---------------	------------	-------------------

Footnote: n= number of samples analysed

S.E. = Standard error

The comparisons of diameters at ground level of tree seedlings in Solit on low Cu dosage of 6.74 mm with copper normal treatments of 6.14 mm was higher than those of Kapkun, Kimorok and Maoi (4.85, 5.33 and 5.71 mm), respectively (P < 0.05, Table 5.5 and Appendix 1).

Copper uptake in both treatments at Solit was high because of high levels of iron (552.8 ppm) that interacted with copper ions in soils and enhanced copper uptake by tree seedlings. Diameter at ground level on iron uptake in Solit with low and high dosages was 6.60 and 5.86 mm, respectively. This was as a result of nitrogen (0.18%), in the soils which enhanced the iron uptake by seedlings of *Acacia senegal* variety *senegal*.

Diameter at ground level of 8.10 mm in Kapkun for manganese uptake on high dosage was higher than 6.48 mm on low dosage in Kimorok, respectively. This was ascribed to acid sandy soils (6.0 and 5.96) in both sites. At this pH, manganese becomes more available to seedlings of *Acacia senegal* variety *kerensis* and *Acacia senegal* variety *senegal*.

There were no significant differences in Dgl of tree seedlings (5.71, 5.65, 5.25 and 5.05 mm) on low Zn dosage in Solit, Kapkun, Kimorok and Maoi, respectively. The seedlings with diameter at ground level of 5.71 mm in Solit on low Zn dosage (6.74 mm) were higher (P < 0.05) than 5.25 and 5.05 mm in Kimorok and Maoi, respectively. This was attributed to high level of available copper (1.67 ppm) in the soils which enhanced the uptake of zinc by promoting the growth of the seedlings. There were no significant differences in Dgl of tree **needlings** (5.60, 5.16, 5.65 and 5.35 mm) on high Zn dosage treatment in Solit, Kapkun, Kimorok and Maoi, respectively. In Kimorok, the diameter at ground level of tree seedlings on high Zn dosage treatment was 5.65 mm higher than 5.60, 5.16 and 5.35 mm in Solit, Kapkun and Maoi, respectively.

This was attributed to high clay (16.27%) and soil organic matter (1.97%) that released Zn uptake as required by the tree seedlings. In Solit, zinc uptake was low because of soil pH of 7.05. At this pH, Zn becomes less soluble in the soil, since the solubility of zinc is highly dependent upon soil pH (Woolhouse, 1983). *Acacia senegal* variety *kerensis* in Kimorok and Maoi seems to take higher level of zinc from the soils than other micronutrients, respectively.

The average heights and diameter at ground level (Dgl) of seedlings on dosage treatments of 197.2 and 6.3 mm in soil medium were higher than 130.2 and 5.3 mm in vermiculite medium in all the sites. This was attributed to interactive effects in the soil, a heterogeneous medium with factors affecting the soil characteristics such as pH, nutrient changes, soil organic matter, sites, treatments (nutrients) and other cations within soil aggregates that influence the physiological growth of plants. These results revealed that *Acacia senegal* variety *senegal* tends to take high levels of copper, iron and manganese while *Acacia senegal* variety *kerensis* takes high levels of zinc from the soils than other micronutrients.

These results indicate that variations in micronutrient uptake and growth pattern are on the basis of soil medium, a heterogeneous medium with unequal competitions of cation and different levels of interactions with other cations in the soils. Vermiculite, a homogenous inert medium, with equal competitions and interactions of cations, differentiated the **concentrations** of Cu, Fe, Mn and Zn uptake for the two varieties of *Acacia senegal*. The medium also showed the effect of micronutrients uptake on levels of tolerance and response of **seed**lings of *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis*.

5.2 Response of Acacia senegal varieties to Micronutrient uptake

Copper uptake of 102.7 ppm in Solit indicated that *Acacia senegal* variety *senegal* absorbed high levels of Cu, while *Acacia senegal* variety *kerensis* in Kimorok and Maoi absorbed low levels of copper uptake, respectively. This was ascribed to the ability of *Acacia senegal* variety *senegal* which absorbed higher concentrations of available Cu than that of *Acacia senegal* variety *kerensis*. *Acacia senegal* variety *senegal* was more tolerant to high level of copper, while *Acacia senegal* variety *kerensis* was tolerant to low levels of Cu. These findings do not agree with report by Mitchell (1964) on ranges of concentration of copper in plants (7 to 30 mg kg⁻¹). However, the result agrees with the suggestion that the concentration depends on the type of soil and plant species and may fall outside these ranges. The high levels of iron uptake of 751.71 ppm in vermiculite medium in Solit was due to the efficiency of *Acacia senegal* variety *senegal* which absorbed high levels of Fe, while *Acacia senegal* variety *kerensis* in Maoi and Kimorok had low efficiency of iron uptake at low levels. This implies that *Acacia senegal* variety *senegal* is less susceptible to iron deficiency while *Acacia senegal* variety *kerensis* is highly sensitive to iron toxicity at high levels of Fe.

The uptake of iron and manganese of 653.5 and 638.25 ppm in soil medium in Kapkun showed that *Acacia senegal* variety *senegal* has higher affinity to absorb available Mn than *Acacia senegal* variety *kerensis*. This showed that *Acacia senegal* variety *senegal* was more tolerant to manganese toxicity than *Acacia senegal* variety *kerensis*.

The levels of zinc uptake of 533.20 and 882.68 ppm in soil and vermiculite media in Kimorok ¹⁸ attributed to the ability of *Acacia senegal* variety *kerensis* in Kimorok to take up high ^{levels} of Zn. Acacia senegal variety senegal in Solit and Kapkun absorbed low levels of zinc. This showed that Acacia senegal variety kerensis tolerated high levels of Zn while Acacia senegal variety senegal tolerated low levels of Zn. Acacia senegal variety senegal tends to take high levels of copper, iron and manganese uptake from the soils than other micronutrients. Acacia senegal variety kerensis takes high levels of zinc from the soils than other micronutrients. Acacia senegal variety kerensis was less susceptible to high levels of iron and copper.

5.2.1 Concentrations of Micronutrient uptake by Acacia senegal varieties

Micronutrient concentrations in *Acacia senegal* seedlings was based on low, normal and high range, according to ratings of Mitchell (1964), Stout (1961), Jones (1972) and Landon (1991). The concentrations of micronutrients uptake by plants as deficient, sufficient, excessive or toxic are given in Table 5.6. These ranges were from mature leaves of different species and varieties of crops.

	Microfutrient	S	
	Deficient	Sufficient (Adequate)	Excessive or toxic
	Ppm		
Copper	< 4	5 - 20	> 20
Iron	< 50	50 - 250	Not known
Manganese	< 20	20 - 500	> 500
Zinc	< 20	25 - 150	> 400

 Table 5.6 Concentration of micronutrients uptake by plants

Kef: Stout (1961); Mitchell (1964), Jones (1972) and Landon (1991)

Copper concentrations in plants range between 5 -20 ppm and deficiencies and toxicity occur at less than 4 ppm and above 20 ppm, respectively (Jones, 1972). Sufficient iron concentrations range from 50 to 250 ppm, and young plants have normal concentrations of 300 to 400 ppm, while deficiencies at levels below 50 ppm and toxicity are rare under field conditions. Manganese range from 20 to 500 ppm and deficiencies at levels below 50 ppm and toxicity levels below 50 ppm and toxicity levels exceed 500 ppm. Zinc concentrations range from 25 to 150 ppm and deficiencies and toxicity occur at less than 20 ppm and above 400 ppm, respectively. These are generalized ranges but there are considerable different between species and even between varieties of the same species (Jones, 1972). The categorized concentration range of micronutrients uptake by *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis* on low and high levels of tolerance are given in Table 5.7.

	Micronutrients		
	Low range	Normal range	High range
		Ppm	
Copper	39:4 - 48.3	53.5 - 66.6	68.5 - 100.1
lron	252.8 - 271.1	396.9 - 510.8	535.0 - 654.7
Manganese	235.5 - 280.1	340.7 - 346.7	366.3 - 400.0
Zinc	182.5 - 217.3	313.3 - 341.3	340.8 - 440.9

Table 5.7 Concentration Ranges of Micronutrients uptake by Acacia senegal varieties

The concentration of copper in Acacia senegal variety senegal and Acacia senegal variety kerensis in normal range was 53.5 - 66.6 ppm while low and high ranges were 39.4 - 48.3 and 68.5 - 100.1 ppm, respectively.

Iron uptake ranged in low, normal and high was 252.8 - 271.1, 396.9 - 510.8 and 535.0 - 654.7 ppm while manganese and zinc 235.5 - 280.1, 340.7 - 346.7 and 366.3 - 400.0 ppm, and 182.5 - 217.3, 182.5 - 217.3 and 340.8 - 440.9 ppm, respectively.

Copper, iron and zinc uptake by Acacia senegal variety senegal and Acacia senegal variety kerensis in low, normal and high ranges were higher than the ratings while manganese in low, normal and high range were within the range in plants by Mitchell (1964), Stout (1961), Jones (1972) and Landon (1991).

5.3 Conclusions

The heights and diameter at ground level of tree seedlings of seedlings in soil medium in all the sites were higher than in vermiculite medium as a result of soils as heterogeneous medium that had inherent nutrients while vermiculite was homogeneous inert medium.

Vermiculite, a homogenous inert medium, with equal competitions and interactions of cations, differentiated and revealed the concentrations of Cu, Fe, Mn and Zn uptake for the two varieties of *Acacia senegal*. The medium also showed the effect of micronutrients uptake on levels of tolerance and response of seedlings of *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis*.

The heights were good measure of growth performance while diameters at ground level of seedlings were not a good indicator for measuring the growth performance of the varieties. Acacia senegal variety senegal tends to take high levels of copper, iron and manganese while Acacia senegal variety kerensis takes high levels of zinc from the soils than other micronutrients. Acacia senegal variety senegal variety senegal less tolerant to high levels of zinc, while Acacia senegal variety kerensis was less tolerant to high level of iron and copper.

Copper, iron and zinc uptake by seedlings of *Acacia senegal* varieties were higher than those ranges of concentrations of uptake of Cu, Fe and Zn for agricultural crops (Jones, 1972). The results of micronutrients uptake and effect of *Acacia senegal* varieties may help to determine quality parameters and the factors influencing the quality of gum arabic under arid and semi-arid environments.

CHAPTER SIX

INFLUENCE OF ACACIA SENEGAL VARIETIES ON QUALITY OF GUM ARABIC

6.0 Introduction

This chapter examines factors that may influence the quality of gum arabic exudate harvested from the natural stands of two *Acacia senegal* varieties in the four study sites. Physical and chemical properties of soils and gum arabic were used to determine the quality of the Kenyan gum.

6.1 Quality of Gum Arabic

The international specifications of quality parameters of gum arabic are given in Table 6.1.

Table 6.1 International Specifications of Quality parameters of Gum Arabic *

Source of Gum arabic: Kordofan gum belt region, Sudan

Species: Acacia senegal var	. <i>senegal</i> and	its varie	ties	
Moisture content (105 [°] C)	13	-	15 %	
Ash content (550°C)	2	-	4 %	
Volatile matter (105 ⁰ C)	51		65 %	
Internal energy (850 ⁰ C)	30	-	39 %	
Optical rotation	-26 ⁰	-	-34 ⁰	
Nitrogen content	0.26		0.39 %	
Cationic compositions of tot	al ash contei	nt (550°C))	
Copper	Iron		Manganese	Zinc
52 – 66 ppm	730 – 2490 ppm		69 – 117 ppm	45 – 111 ppm

6.2 Physical properties of Gum arabic

The physical properties of gum arabic, established as quality parameters include moisture, total ash, volatile matter, and internal energy. Gum arabic is a natural product complex mixture of hydrophilic carbohydrate and hydrophobic protein components (FAO, 1990). Hydrophobic protein component functions as an emulsifier which adsorbs onto surface of oil droplets while hydrophilic carbohydrate component inhibits flocculation and coalescence of molecules through electrostatic and steric repulsions in food additives (Anderson and Weiping, 1990).

Moisture content facilitates the solubility of hydrophilic carbohydrates and hydrophobic proteins in gum arabic. Total ash content is used to determine the critical levels of foreign matter, acid insoluble matter, salts of calcium, potassium and magnesium. The cationic compositions of ash content are used to determine the specific levels of heavy metals in quality of gum arabic (FAO, 1990).

Volatile matter of gum arabic determines the number of hydrocarbons contained in sugar compositions (arabinose, gafactose and rhamnose); these function as binders in the making of cough syrups in pharmaceutical industry.

Internal energy of gum arabic is the actual energy required to produce the amount of carbon when the gum is heated to 500^oC to release carbon dioxide gas. Carbon dioxide plays an ^{Important} role as a stabilizer, thickener, binder and protective agent in food, pharmaceutical, ^{and} technical industries (FAO, 1995).

Optical rotation is used to determine the nature of sugars in gum arabic obtained from *Acacia senegal* variety *senegal*. The specifications state that the best quality of gum arabic must have negative optical rotation with the range of -26° to -34° (Table 6.1). Nitrogen content in gum arabic determines the number of amino acid compositions with the range of 0.26 % to 0.39%.

6.3 Physical Properties of Gum arabic in the four Study sites

The physical properties of gum arabic from the four study sites are given in Table 6.2. Acacia senegal variety senegal was dominant in Solit and Kapkun while Acacia senegal variety kerensis was dominant in Kimorok and Maoi, respectively.

		Moisture content	Ash content	Volatile matter	Internal energy
Sites	Variety	(%)	(%)	(%)	(%)
		Mean± S.E	Mean± S.E	Mean± S.E	Mean± S.E
Solit	senegal	15.00 ±0.50	2.94±0.20	63.72±0.40	32.96±0.30
Kapkun	senegal	14.90±1.80	3.16±0.20	64.24±0.30	33.00 ± 0.30
Kimorok	kerensis	17.50 <u>+</u> 1.00	2.88±0.20	63.80±0.20	33.40±0.30
Maoi	kerensis	15.40±0.40	2.72±0.20	63.60±0.50	33.76±0.60

Table 6.2: Physical properties of Gum arabic from the study sites

Footnote: n = 5

S.E. = Standard error

(a) Moisture content

Moisture content in gum arabic obtained from variety *kerensis* in Kimorok and Maoi (17.5 and 15.4%) were higher (P< 0.05) than those of variety *senegal* in Solit and Kapkun (15.0 and 14.9%), respectively.

Moisture content in gum arabic from variety *senegal* in Solit and Kapkun (15.0% and 14.9%) fell within international specifications (13% to 15%), while variety *kerensis* in Kimorok and Maoi (17.5% and 15.4%) fell outside the specifications. Gum arabic from *Acacia senegal* variety *senegal* in Solit and Kapkun was of better quality than that of variety *kerensis* in Kimorok and Maoi (Anderson and Weiping, 1990).

(b) Ash content

Ash content in gum arabic from variety *senegal* in Solit and Kapkun (2.94 and 3.16%) was higher (P< 0.05) than those of variety *kerensis* found in Kimorok and Maoi (2.88% and 2.72%), respectively (Table 6.2). Ash content in gum arabic from *Acacia senegal* varieties in the study sites fell within the international specifications. The ash content of the Kenyan *Acacia senegal* varieties was better than that found in Uganda (4.5%) which falls outside the international specifications (Anderson and Weiping, 1991).

(c) Volatile matter

In Kapkun, volatile matter in gum arabic from variety *senegal* (64.2%) was higher (P < 0.05) than the quantities of variety *kerensis* found in Kimorok, Solit and Maoi (63.8%, 63.7% and 63.6%), respectively (Table 6.2). Volatile matter contents of the gum arabic from both varieties (64.2%, 63.7%, 63.8 % and 63.6%) were within the international specifications range of 51% to 65% (Table 6.1).

(d) Internal energy

Internal energy in gum arabic obtained from *Acacia senegal* variety *kerensis* in Maoi and Kimorok (33.76% and 33.4%), were not significantly different (P> 0.05) from those of variety senegal found in Kapkun and Solit (33.0% and 32.96%), respectively (Table 6.2).

The internal energy in gum arabic from *Acacia senegal* varieties meets the international specifications (30 % to 39%) FAO (1990).

6.4 Influence of Soil Composition on the Quality of Gum arabic Varieties

The chemical composition of soils and gum arabic are given in Table 6.3. The correlations between chemical composition of soils and gum arabic in relation to sites and varieties were used to establish which variety of *Acacia senegal* produces better quality of gum arabic in the study sites.

	Site	Solit	Kimorok	Kapkun	Maoi
			Gum arabic		
		Mean± S.E.	Mean± S.E.	Mean± S.E.	Mean± S.E.
N	%	0.30 ± 0.02	0.31 ± 0.02	0.28 ± 0.02	0.34 ± 0.02
Cu	ppm	40±5.49	38.5±5.49	45.2±5.34	32.0±5.49
Fe	ppm	973±20	1243 ± 20	1415±19	861 ±20
Mn	ppm	93±8.8	72 ±8.8	109±8.5	93±8.8
Zn	ppm	78±13.7	124±13	83±13.7	44±13.7
			Soils		
N	%	0.18 ± 0.02	0.14 ± 0.02	0.30 ± 0.02	0.17 ± 0.02
Cu	ppm	1.67 <u>+</u> 0.08	0.73±0.08	0.61+0.08	0.58+0.08
Fe	ppm	151 <u>+</u> 11	234+11	250+11	287+11
Mn	ppm	263 <u>+</u> 14	320+14	344+14	383+14
Zn	ppm	1.73+0.47	7.64+0.46	6.00+0.47	5.06+0.46

Table 6.3 Chemical compositions of Soils and Gum Arabic

Footnote: n = 17

S.E. = Standard error

6.4.1 Nitrogen

The mean nitrogen contents of gum arabic from Acacia senegal variety kerensis in Maoi (0.34%) and Kimorok (0.31%) were higher (P < 0.05) than those of variety senegal found in Kapkun and Solit (0.28% and 0.30%) (Table 6.3, Figure 6.1).



Figure 6.1 Comparison of Soil Nitrogen and Gum arabic Nitrogen

Nitrogen content in gum arabic from variety kerensis in Kimorok (0.31%) was significantly correlated with soil nitrogen (0.14 % with (r = 0.26, P < 0.05) (Appendix III). The positive correlation may be attributed to high concentration of zinc (7.64 ppm) in the soils, which may have interacted with soil nitrogen (0.14%) resulting to an increase in available nitrogen for uptake by the variety, which resulted in increased content of gum nitrogen (0.31%). This observation agrees with the work of Jones (1972) on nitrogen uptake by subterranean clover. He added high concentration of zinc in the soil with low percentage of nitrogen. He found an increase in growth of subterranean clover due to high Zn concentration by applying nitrogen and zinc fertilizers together but not to zinc concentration alone. Jones (1972) reported that UNIVERSITY OF NAIROBI LIBRARY ²n concentration in the roots correlated with high level of nitrogen in the leaves.

120

The report by Jones (1972) on interaction between high concentration of Zn and nitrogen in the leaves of subterranean clover can also be applied on the process of gum arabic exudates from the varieties. Soil nitrogen (0.17 %) was negatively correlated with gum nitrogen arabic (0.34%) from variety *kerensis* in Maoi with r = -0.12. The negative correlation may have been due to the influence of soil organic matter (1.15%) in the soils which supplied more available nitrogen for uptake by the variety which resulted in high content of gum nitrogen (0.34%).

The content of nitrogen in gum arabic (0.28% from variety *senegal* in Kapkun was significantly correlated with soil nitrogen (0.30%) with r = 0.13 (P <0.05). This was due to high level of soil organic matter (1.73%) which may have enhanced soil nitrogen (0.3%) to produce gum nitrogen of 0.28%. Gum nitrogen (0.3%) was negatively correlated with soil nitrogen (0.18%) in Solit with r = -0.52 (P <0.05). The negative correlation may be attributed to the influence of soil pH (7.05) on availability of soil nitrogen (0.18%). At soil pH 7.05, nitrogen becomes less available in the soil because the best soil pH for microorganism to break nitrogen is between 5.8 and 6.6 (Watson and Brown, 1998).

The results of gum nitrogen from the two varieties in the four study sites do not agree with the observations of Chikamai and Banks (1993) on levels of nitrogen in gum arabic from *Acacia senegal* variety *kerensis* found in Kargi, Isiolo and Ngurunit (0.5, 0.4 and 0.43%, **respectively**) in Marsabit and Isiolo districts. The levels of nitrogen in gum arabic (0.28% - 0.34%) from variety *kerensis* and *senegal* were within the international specifications (0.26% - 0.39% N) while those of Chikamai and Banks (1993) fell outside the specifications given in Table 6.1.

6.4.2 Copper

The concentration of copper in gum arabic from Kapkun (45.2 ppm) was higher (P < 0.05) than those found in Solit, Kimorok and Maoi (40, 38.5 and 32.0 ppm, respectively) (Table 6.3, Figure 6.2).



Figure 6.2 Comparison of Soil Cu and Cu concentrations in Gum arabic

Copper content in gum arabic (45.2 ppm) from variety *senegal* in Kapkun was positively correlated with soil copper (0.61 ppm) with r = 0.41 (P <0.05) (Appendix III). This may be ascribed to ferralic Cambisols with soil organic matter (1.73%) and carbon: nitrogen ratio (4.82:1) which may have enhanced copper uptake by the variety resulting to high copper content in gum arabic (45.2 ppm). The concentration of copper in gum arabic (40 ppm) from variety *senegal* in Solit was correlated negatively with soil copper (1.67 ppm) with r = -0.31 (P <0.05). The negative correlation between soil and gum copper in Solit may be due to high levels of calcium (17.33 cmol (⁺)/kg) and phosphorus (11.54 ppm), which can ^{reduce} copper uptake resulting to decreased copper content in gum arabic (40 ppm).

The content of copper in gum arabic (ppm) from variety *kerensis* in Kimorok correlated positively with soil copper (0.73 ppm) with r = 0.49 (P <0.05) (Appendix III). This was most likely due to high levels of phosphorus (14.6 ppm) in the soils which may have reduced the availability copper uptake by the variety which produced copper content in gum arabic (38.48 ppm) (Marschner, 1986).

Copper in gum arabic (32.0 ppm) from variety *kerensis* in Maoi correlated positively with soil copper (0.58 ppm) with r = 0.29 (P <0.05) (Appendix III). This is attributed to high levels of available iron and manganese (287 and 383 ppm) in the soil solution which interacts with copper ions by competing for surface adsorption sites (Jones, 1972).

Copper content in gum arabic (32.0 - 45.2 ppm) from variety *senegal* and *kerensis* in the study sites were below the range of (52 - 66 ppm) from *Acacia senegal* varieties cited in the international specifications (FAO, 1990).

6.4.3 Iron

The concentration of iron in gum arabic from *Acacia senegal* variety *senegal* found in Kapkun and Solit (1415 and 973 ppm) were significantly higher (P < 0.05) than those of *Acacia senegal* variety *kerensis* in Kimorok and Maoi (1243 and 861 ppm) (Table 6.3, Figure 6.3).



Figure 6.3 Comparison of Soil Fe and Fe concentrations in Gum arabic

Iron content in gum arabic (1415 ppm) from variety *senegal* in Kapkun was correlated negatively with soil iron (250 ppm) with r = -0.04 (P <0.05) (Appendix III). The concentration of iron in gum arabic (973 ppm) from variety *senegal* in Solit was correlated with soil iron (151 ppm) with r = -0.16 (P <0.05). These weak negative correlations between soil and gum iron in Solit and Kapkun were due to soil pH (5.96 - 7.05), which may have influenced the decrease of Fe in the soils (Table 4.2). Fe becomes more available the lower the pH (5.96) and it decreases at high pH (7.05), because the solubility of iron is highly dependent upon soil pH (Woolhouse, 1983; Mclean, 1982).

The concentration of iron in gum arabic (1243) from variety *kerensis* found in Kimorok was ^{correlated} negatively with soil iron (234 ppm) with r = -0.52 (Appendix III). The strong ^{negative} correlation between soil and gum iron in Kimorok may be ascribed to interaction ^{with} high levels of zinc (7.64 ppm) in the soils (Table 4.2). The interaction between Zn and Fe in the soils is due to their similar ionic radii of 0.83 and 0.73 nm, respectively, which may have reduced the availability of iron uptake by variety *kerensis* from the soils. Iron content in gum arabic (861 ppm) from variety *kerensis* in Maoi was correlated with soil iron (287 ppm) with r = -0.12 (P <0.05. This is due to the interaction of high levels of manganese (383 ppm) with iron in the soils (Table 4.2). The content of iron in gum arabic (861 - 1415 ppm) from the varieties were within range (730 - 2490 ppm) of the international specifications (FAO, 1990).

6.4.4 Manganese

Manganese content in gum arabic from *Acacia senegal* variety *senegal* in Kapkun (109 ppm) was higher (P < 0.05) than the quantities of *Acacia senegal* variety *kerensis* found in Kimorok, Solit and Maoi (93, 93 and 72 ppm) (Table 6.3, Figure 6.4).



Figure 6.4 Comparison of Soil Mn and Mn concentrations in Gum arabic

The content of manganese in gum arabic (109 ppm) from variety *senegal* in Kapkun was ^{correlated} negatively with soil manganese (344 ppm) with r = -0.28 (P <0.05) (Appendix ^{III}). The negative correlation may be attributed to soil organic matter (1.73%) which may

have increased the level of manganese in the soil resulting in high manganese content in gum arabic (109 ppm). Manganese in gum arabic (93 ppm) from variety *senegal* in Solit was correlated negatively with soil manganese (263 ppm) with r = -0.23 (P <0.05). The negative correlation may be due to high level of soil pH (7.05) which increases the availability of manganese uptake by variety *senegal* from the soils. At pH (7.05), Mn becomes less available in the soil (263 ppm), since the solubility of manganese is highly dependent upon soil pH (Woolhouse, 1983).

The level of manganese in gum arabic (93 ppm) from variety *kerensis* found in Kimorok was significantly correlated (r = 0.40, P <0.05) with soil iron (320 ppm) (Appendix III). Manganese content in gum arabic (72 ppm) from variety *kerensis* in Maoi was correlated positively with soil manganese (383 ppm) with r = 0.25 (P <0.05). The positive correlations may be due to high levels of soil organic matter (1.73 and 1.15%, respectively) which enhances the availability of manganese uptake by *variety kerensis*.

Manganese content in gum arabic (72 - 109 ppm) from variety *senegal* and *kerensis* in the study sites were within the range (69 - 117 ppm) thus falling within the international specifications (FAO, 1990).

6.4.5 Zinc

⁷inc in gum arabic from *Acacia senegal* variety *kerensis* in Kimorok was 124 ppm and was ⁸ignificantly (P < 0.05) higher than the quantities of zinc from Maoi, Solit and Kapkun (44, ⁷⁸ and 83 ppm), respectively (Table 6.3, Figure 6.5).



Figure 6.5 Comparison of Soil Zn and Gum arabic Zn concentrations

The concentration of zinc in gum arabic (78) from variety *senegal* in Solit was correlated positively with soil zinc (1.73 ppm) (r = 0.48, P <0.05) (Appendix III). The positive correlation was due to high soil pH (7.05), which reduces the availability of zinc in the soils (Watson and Brown, 1998). Zinc content in gum arabic (83 ppm) from variety *senegal* in Kapkun was correlated negatively with soil zinc (6.00 ppm) with r = -0.37 (P <0.05) (Appendix III). The negative correlation may be due to high level of phosphorus (14.10 ppm) which reduces the availability of zinc uptake by variety *senegal* from the soils (Jones, 1991).

The concentration of zinc in gum arabic (124 ppm) from variety *kerensis* in Kimorok was correlated with soil zinc (7.64 ppm) with r = 0.36 (P <0.05) (Appendix III). Zinc in gum arabic (44) from variety *kerensis* in Maoi significantly correlated (r = 0.63, P <0.05) with soil the (5.06 ppm). Positive correlations between soil and gum zinc in Kimorok and Maoi may due to the high levels of soil organic matter (1.97% and 1.15%, respectively), which enhances the availability of zinc in the soils (Jones, 1991). Zinc concentrations in gum arabic in Solit and Kapkun (78 and 83 ppm) from variety *senegal* were within the range of (45 - 111 ppm) in the international specifications while those of gum arabic in Maoi and Kimorok (44 and 124 ppm) from variety *kerensis* fell outside the specifications (FAO, 1990).

6.5 Conclusions

Moisture content in gum arabic from *Acacia senegal* variety *senegal* fell within international specifications while *Acacia senegal* variety *kerensis* fell outside the specifications, while ash, volatile matter, zinc and internal energy contents in gum arabic from *Acacia senegal* variety *kerensis* and variety *senegal* fell within the specifications.

By contrast, nitrogen, iron and manganese contents in gum arabic from *Acacia senegal* variety *kerensis* and *Acacia senegal* variety *senegal* fell within the international specifications while copper and zinc did not.

Gum arabic obtained from Acacia senegal variety senegal had high levels of copper, iron and manganese, contrary to Acacia senegal variety kerensis which instead had high levels of zinc and nitrogen.

The quality of gum arabic parameters depends on the variety, the soil conditions and soil ^{types} as well as availability of the nutrients in the study sites. The gum arabic from *Acacia* ^{tene}gal variety senegal in Solit and Kapkun was of better quality than that of *Acacia* ^{tene}gal variety kerensis in Kimorok and Maoi.
CHAPTER SEVEN

INTEGRATING DISCUSSION AND RECOMMENDATIONS

Most gums are produced by *Acacia senegal* and other closely related *Acacia* species which grow in abundance in the vast areas of arid and semi-arid ecosystems in Kenya. The Acacia species are also very useful in soil stabilization and nitrogen fixation, which improves soil fertility by increasing soil organic matter and availability of both macro and micronutrients. The canopies of Acacia gum trees control wind velocity, maintain water balance of plants, and protect soils against rainfall torrents that lead to soil erosion. The canopies also lessen solar radiation and reduce rates of evaporation.

Gum arabic from Sudan is tapped from *Acacia senegal* variety *senegal* trees, which are cultivated as a cash crop in agroforestry system known as the bush-fallow system of shifting cultivation. The main gum producing regions are the sandy clay plains situated in east and central Sudan covering most of Kordofan and Darfur and parts of White Nile state. The quality parameters of gum arabic depend on the source of origin of gum exudates, soil characteristics and climate of the area.

The international specifications used to assess the quality of gum arabic in the world market are based on the Sudan gum obtained from *Acacia senegal variety senegal*. There are no ^{Ruitable} alternative synthetic additives to substitute the use of gum arabic, owing to its ^{Excellent} shelf life stability to oil-in-water emulsions and long lasting fresh taste in foodstuff.

^{the} main factors affecting quality of Kenyan gums are the botanical sources, poor tapping ^{thods} and harvesting period. The harvesting period is particularly important since mixing

of wet and immature gums with mature products can lead to adulteration. Kenya's gum is not able to attract premium prices because of problems relating to quality caused by adulteration and impurities. These impurities can result from transportation of the gum on the dust roads, making it unsuitable for use in the pharmaceutical industry. As a result, exporters have to spend a lot of money (about 20% of their costs) cleaning the gum, thereby reducing profits.

Results presented and discussed in this thesis have shown Acacia senegal variety senegal and Acacia senegal variety kerensis can be good sources of gum arabic in the study area. Attention was also focused on quality of gum arabic from specific botanical source (Acacia senegal varieties), sites and soil types.

Results presented in Chapter Four have shown that physico-chemical properties on characterization of soils based on and soil fertility were factors influencing the quality of gum arabic. The effect of low moisture regimes in sandy soils limits the nutrient uptake, physiological growth of plants and conditions of soil microorganisms' survival in arid and semi-arid environments. The low moisture characteristics affect the solubility of hydrophilic carbohydrates and hydrophobic proteins in gum arabic which determine the emulsification and stabilization properties in food industry.

Acacia senegal variety senegal grows well in neutral calcic Xerosols and acidic ferralic Cambisols which prevail in Solit and Kapkun. The soils are sandy loam to clay loam, of low ¹⁰ medium fertility with respect to nitrogen, potassium, calcium, magnesium and copper.

Acacia senegal variety kerensis grows well in shallow, rocky and stony acidic Lithosols and Xerosols, and ando-chromic Cambisols which prevail in Kimorok and Maoi. The soils are well drained, sandy clay loam to sandy loam and have medium to high fertility especially with respect to carbon, phosphorus, iron, manganese and zinc.

The mean nitrogen content of gum arabic from *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis* (0.28 % - 0.34%) indicate high functionalities of amino acids which provide stable and strong emulsification properties. The levels of nitrogen fall within international specifications range of 0.26% - 0.39% N (FAO, 1990). The Cu content of gum arabic ash (550^oC) from *Acacia senegal* variety *senegal* and *Acacia senegal* variety *kerensis* (32 - 45 ppm) is below the specifications range (52 - 66 ppm), while the levels of Fe, Mn and Zn of (861 - 1415 ppm, 72 - 109 ppm and 78 - 83 ppm) are within the international specifications (730 - 2490 ppm, 69 - 117 ppm and 45 - 111 ppm).

The established standard quality control parameters and specific levels of cationic compositions in total ash content at 550° C are as follows: 52 - 66 ppm Cu, 730 -2490 ppm Fe, 69 -117 ppm Mn and 45 = 111 ppm Zn, respectively. The results of this study indicated that the quality of gum arabic from Kenya is not of low quality as stated by JECFA specifications (FAO, 1990).

The results presented in Chapter Five on uptake of micronutrients have shown that Acacia senegal variety senegal tends to absorb high levels of copper, iron and manganese while Acacia senegal variety kerensis tends to absorb high levels of zinc from the soils than other micronutrients. The uptake results showed that Acacia senegal variety senegal was more leterant to copper, iron and manganese toxicities than Acacia senegal variety kerensis, which by contrast was more tolerant to zinc toxicity. This may imply that gum quality parameters depend on the variety, the soil conditions and availability of the nutrients for uptake by the varieties.

Variations in soil types, soil fertility status and micronutrients uptake and the *Acacia senegal* variety are important in assessing the of type of agronomic practices required for establishment and identifying the factors influencing the quality of gums.

Further research is required to compare gums from natural stands of *Acacia senegal* and closely related species considering the geographical locations, sites, soil types, single botanical sources and climate in the arid and semi-arid regions of Kenya. The research should also include boron and molybdenum uptake by the varieties and their possible effect on gum quality.

The JECFA specifications (FAO, 1990) on gum quality also need to be revised to cover gums unique character from natural stands of *Acacia senegal* and closely related species for new untries which are yearning to join the world market like Kenya. The parameters should be on gum arabic specific rotations, nitrogen contents, boron and molybdenum composition, ic viscosity, proteins and carbohydrates compositions as well as heavy metals and amino acid composition.

REFERENCES

- Allaway, W.H.: (1992). Soil-plant-animal and human inter-relationships in trace element nutrition. In: Trace Elements in Human and Animal Nutrition. 2:465-488, Academic Press.
- Alley, M. M., D. C. Martens, M. G. Schnappinger, Jr., and G. W. Hawkins. (1972). Field calibration of soil tests for available zinc. Soil Sci. Soc. Am. Proc. 36:621-624.
- Anderson, D. M. W. (1987). Food Hydrocolloids, 4: 327.
- Anderson, D. M. W. and Morrison, N. A. (1990). Food Additives and Contaminants 7(2), 175-180.

Anderson, D. M. W. and Wang Weiping. (1991). Food Hydrocolloids 5 (3), 297.

Anderson, J. M. and Ingram, J. S. I. (1993). Tropical soil biology and fertility: A handbook of methods. CAB International, Wallingford, UK.

Anderson, O.E., and R.M. Harrison. (1970). Micronutrient variation within cotton leaf

Tissue as related to variety and soil location. Commun. Soil Sci. Plant Anal.

1:163-172.

- Anderson. A.J. and Evans, H.J. (1956). "Sulphur, manganese, iron and other elements which may decrease the uptake of molybdenum by plants". Adv. Agron. 8,194.
- Anon (1991). Australian Standards Methods of Testing Soils for Engineering Purposes (AS1289.0-1991).

Austin, Max E., (1984). An observation of nutrient levels in old, unfertilized rabbiteye

Blueberry plants. Hort. Science 19(3): 417-418.

Baker, J.M., and B.B. Tucker. (1973). Critical N, P, and K levels in winter wheat. Comm.Soil

Sci. Plant Anal. 4(5): 347-358.

- Ballal, M.E., El Siddig, E.A., Elfadl, M.A. & Luukkanen, O. (2005). Gum arabic yield in differently managed Acacia senegal stands in western Sudan. Agroforestry Forum 63: 237-245.
- **Barrow, E.G.C.** (1996). The Dry lands of Africa: Local participation in tree management Initiative Publishers, Nairobi, Kenya. 268pp.
- Beentje, H.J. (1994). Kenya Trees Shrubs and Lianas. National Museums of Kenya, Nairobi. 722p.
- Bekele-Tesemma A., Birnie A. and Tengno BO. (1993). "Useful Trees and Shrubs for Ethiopia. Identification Propagation and Management for Agricultural and Pastoral Communities". Regional Soil Conservation Unit, RSCU SIDA, Nairobi.
- Bell, R. W. (1997). Diagnosis and prediction of boron deficiency forplant production. Plant Soil 193: 149-168.
- Bennett, J.H., Lee, E.H., Krizek, D.T., Olsen, R.A. and Brown, J.C. (1982).

Photochemical reduction of iron II. Plant related factors". J. Plant Nutr. 5, 335-344.

Bhatt K.C., Vaishnav, P.P., Singh, Y.D., Chinoy, J.J. (1976). "Reversal of gibberellic and induced inhibition of root growth by manganese". Biochem. Physiol, pflanz. <u>170</u>, 453-55.

Bingham, F.T. (1963). Relation between phosphorus and micronutrients in plants. Soil Sci.

Soc. Amer. Proc. 27: 389-391.

Bingham, F.T. (1982). Boron, P. 431-448. In A. L. Page (ed.), Methods of soil analysis, Part 2: Chemical and mineralogical properties. Amer: Soc. Agron. Madison, WI, USA.

- Blake, G.R; (1965). "Bulk density". In "Method of soil analysis". Part I. American Society of Agronomy, No.9. Inc. Publisher, Madison, WI. PP.374-377.
- Blake, G.R; (1965). "Particle density". In "Method of soil analysis". Part I. American Society of Agronomy, No.9. Inc. Publisher, Madison, WI. PP.371-73.
- Bohn, H. L., B. L. McNeal, and G.A. O'Connor. (1985). Soil Chemistry, 2nd ed. John Wiley and Sons, New York.
- Bolland, M. D. A. (1992). Effect of sampling depth on bicarbonate soil phosphorus test values. Fertilizer research 32, 121 – 4
- Bouyoucos, G.J. (1962). "Hydrometer method improved for making particle size analyses of Soils". Agronomy Journal, <u>53</u>; 464-465.
- Boyle, John F., and Cyril B. Smith. (1985). Growth and leaf elemental composition of snap beans as affected by applied zinc and interacting fertilizers. Comm. Soil Sci. Plant Anal 16(5): 501-507.
- Brady, N.C., and R.R. Weil. (1999). The nature and properties of soils, 12th ed. Prentice Hall, Upper Saddle River, New Jersey, USA.
- Bremmer, J.M., and C.S. Mulvaney. (1982). Nitrogen total. P. 595-624. In A.L. Page (ed.), Methods of soil analysis. Agron. No. 9, part 2: Chemical and microbiological Properties, 2nd ed., Am. Soc. Agron. Madison, WI, USA.
- BremnerJ.M. (1965). "Total nitrogen". In Black, C.A, (ed). "Method of Soil Analysis". Agronomy 9. Am. Soc. Agron. Madison
- Bremner, J.M., and D.R. Keeney. (1965). Steam distillation methods for determination of ammonium, nitrate and nitrite. Anal. Chem. Acta. 32: 215- 163.

Brown, J. R. (1987). Soil testing: sampling, correlation, calibration, and interpretation.
Soil Sci. Soc. Amer: Spec. Publ. 21. Soil Sci. Soc. Am. Madison, WI, USA.

Brown, J.C., Clark, R.B., and Jones, W.E. (1977). "Efficient and inefficient use of phosphorus by sorghum". Soil Sci. Soc. Am. J., 41, 747-750

Buresh, R. J., E.R. Austin, and E.T. Craswell. (1982). Analytical methods in N-15 research Fert. Res. <u>3</u>: 37-62.

 Caldwell, T.H. (1971). In "Trace element in soils and crops". PP. 62-72. Ministry of Agriculture, Fisheries and Food. Tech. Bull., 21. HMSO London. California Fertilizer
 Association, Soil Improvement Committee, 1980. Western Fertilizer Handbook 6th (ed.), Interstate Printers and Publishers. Danville, ILL, USA.

Calton, W.E. Vail, J.W., and Padhye, V.P. (1961). "Micronutrients problems in Tanganyika"-Proceedings VIth International Congress of soil science, Paris. East Afri. Agric. For J., <u>27</u>, 13.

Chamberlain G. T. and Searle A.J. (1963). "Trace elements in some East African soils and plants in Relation to Manganese". East Africa Agric. For. J., <u>29</u>; 114-119.

Chamberlain, G.T. (1961). "Trace elements in some East African soils and plants on Relation to cobalt, beryllium, lead, Nickel and zinc". East African Agric. For. J., <u>29</u>; 121-125.

Chapman, H.D. (1966). Diagnostic criteria for plants and soils. Univ. of Calif., Div. of Agric.

Sci., Riverside, Calif. p. 793.

Chapman, H.D. (1967). Plant analysis values suggestive of nutrient status of selected crop p.77-92. In G.W. Hardy (ed.). Soil testing and plant analysis, part II. SSSA Spe Pub. No. 2. Soil Sci. Soc. of Amer., Madison, Wis.

^{pman}, H.D., and P.F. Pratt. (1961). Methods of analysis for soils, plants and water. Univ. California, Berkeley, CA, USA. Chikamai, B. N.and Gachathi, N. (1994). "Gum and Resin Resources in Isiolo District, Kenya. Ethno botanical and Reconnaissance survey". E. Afric. For. J.59 (4), 345-351.

- Chikamai, B. N. (1997). "Production, Markets and quality control of Gum arabic in Africa: Findings and Recommendation for FAO Project. In J.O.Mugah, B.N. Chikamai and E. Casadei (eds.)" Conservation, Management and Utilization of Plant Gums, resins and essential oils." Proceedings of a Regional conference for Africa held in Nairobi, Kenya.
- Chikamai, B. N.and Banks, W.B. (1993). "Gum arabic from Acacia senegal (L) Wild. in Kenya. Food Hydrocolloids <u>7</u> no. 6 pp.521-534.
- Clark, R.B. (1970). Effect of mineral nutrient levels on the inorganic composition and growth of corn (Zea mays L.). Ohio Agri. Res. & Dev. Center Res. Cir. 181. 21 p.

Criley, R.A., and W.H. Carlson. (1970). Tissue analysis standards for various

Floricultural crops. Florist's Rev. 146:19-20, 70-73.

Dale, I.R. and Green Way, P.J. (1961). "Kenya Trees and Shrubs". Buchanan's Kenya Estates Limited in association with Harchards, London, pp.654.

Darkoh, M.B.K. (1991). "Irrigation and development in Kenya's ASAL". Paper presented at the Conference on Environment Stress in African Dry lands and its impacts on River Basins, Addis-Ababa, and April 22-24, 1991.

Day, J.L., and M.B. Parker. (1985). Fertilizer effects on crop removal of P and K in

Coastal Bermuda grass forage. Agron. J. 77: 110-114.

Day, P.R. (1965). Particle fractionation and particle size analysis. P. 546-566. In C.A. Black (ed.), Methods of soil analysis, Agron No.9, Part I: Physical and mineralogical properties. Am. Soc. Agron., Madison, WI, USA.

De Kock., P.C., Hall, A. and Inkson, R.H.E. (1979). "Active iron in plant leaves". Ann. Bot. (London) N.S., <u>43</u>, 737 - 740.

- **Donahue, S.J., and D.E. Brann**. (1984). Optimum N concentration in winter wheat grown ir the Coastal Plain region of Virginia. Comm. Soil Sci. Plant Anal. 15(6): 651 661.
- **Doolittle, J.A., K.A. Sudduth, N.R. Kitchen and S.J. Indorante.** (1994). Estimating depths to clay pans using electromagnetic induction methods. J. Soil and Water Cons. 49:572-575.

Engel, R.E., and J.C. Zubriski. (1982). Nitrogen concentrations in spring wheat at

several growth stages. Comm. Soil Sci. Plant Anal. 13(7): 531-544.

- Evans J. (1982). "Plantation Forestry in the Tropics". Oxford University Press, Nairobi, pp.472.
- Fales, S.L., and K. Ohki. (1982). Manganese deficiency and toxicity in wheat: Influence on growth and forage quality of herbage. Agron. J. 74(6): 1070-1073.
- ^{FAO} (1997). Food and Nutrition Paper No. 52: 5

¹⁴0 (1995). Gums, resins and latexes of plant origin. Non wood forest products Vol.6.

⁴⁰, (1995). The Role of Acacia species in the rural economy of the dry Africa and the ¹¹East.

^{Rome} (1990). Food and Nutrition paper, No.49: 23

FAO. (1990). Management of gypsiferous soils. Soils Bull. No. 62, Food and Agriculture Organization, Rome, Italy.

- **FAO.** (1980). Soil testing and plant analysis. Bull. No. 38/1, Food and Agriculture Organization, Rome, Italy.
- FAO (1986). "African Agriculture: the next 25 years, Annex IV, Irrigation and Water Control". Food and Agriculture Organization, Rome.
- FAO. (1970). Physical and chemical methods of soil and water analysis. Soils Bull. No 10.Food and Agriculture Organization, Rome, Italy.
- FAO. (1974). The Euphrates Pilot Irrigation Project. Methods of soil analysis, Gadeb Soil Laboratory (A laboratory manual). Food and Agriculture Organization, Rome, Italy.
- Fick,K.R., McDowell,LR., Miles,P.H.,Wilkinson,N.S., Funk,J.D. and Conrad, J.H., (1979). Methods of Mineral Analysis for plant and animal tissues (2nd.).Anima Science Department, University of Florida; Gainesville F
- Foy, C.D., Chaney, R.L. White, M.C. (1978)."The physiology of metal toxicity in plants' Ann Rev. Plant physiol., 29, 511-566.
- Foy, C.D., Webb, H.W. and Jones, J.E. (1991). "Adaptation of cotton genotypes to an acid, manganese toxic soil". Agron. J., <u>73</u>, 107 -111.

Gachathi, F.N. (1994). Variation in Acacia senegal and its relationship with A. circummaginat and A. thomasii in Kenya. Msc. Thesis, University of Reading.

Gaines, T. P., and G. A. Mitchell. (1979). Boron determination in plant tissue by the azomethine-H method. Commun. Soil Sci.Plant Ana. 10: 1099 – 1108.

^{vovernment} of Kenya, (1993). National Development Plan, Government Printers, Nairobi.

- Graham, Robin D., and E.K. Sandanandan Nambiar. (1981). Advances in research on copper deficiency in cereals. Aust. J. Agri. Res. 32: 1009-1037.
- Graves, C. J. Adams, P. and Winsor, G.W. (1977). "Some effects of indol- 3- ylacetic acid on the rate of initiation and development of flowerbuds of chrysanthemum moriflorum". Ann. Bot. (London) N.S., <u>41</u>, 747 - 753.
- Guerrero, M.G., Vega, J.M. and Losada, M. (1981). The assimilatory nitrate reducing systems and its regulation. Annu. Rev. Plant physiol., 32, 169-204.
- Gupta, U.C. (1993). Boron, molybdenum, and selenium. Pages 91-99 IN M.R. Carter(ed.) Soil sampling and methods of analysis. Lewis Publ., Boca Raton, Florida.
- Halvorson, A.D. and J.D. Rhoades. (1974). Assessing soil salinity and identifying potential saline seep areas with field soil resistance measurements. Soil Sci. Soc. Amer.Proc. 38:576-581.
- Herlocker, D. (Ed.). (1999). Rangeland Resource in Eastern Africa: Their Ecology and Development. Published by GTZ, Nairobi, Kenya.
- Hesse, P. R. (1971). A textbook of Soil Chemical Analysis. John Murray, London.
- Hewitt, E.J. (1963). "The essential nutrient element requirements and interactions". In F.C. Steward (Ed.). Plant physiology, 1.3, 137-360. Academic press.
- Hewitt, E.J. (1983). In "Diagnosis of mineral Disorders in plants" (Eds. C. Bould, E.J.Hewitt and P. Needham), HMSO, London.
- Hinga, G., F. N. Muchena and C. M. Njihia. (1980). Physical and Chemical methods of soil analysis. Internal Publication, National Agricultural Laboratories, Nairobi.

Horst, W.J. and Marschner, H. (1978). "Effects of excessive manganese supply on uptake and translocation of calcium in bean plants". (Phaseolus vulgaris L.). Z.

Pflanzenphysiol., <u>87</u>, 137-148.In H.D. Chapman (ed.). Diagnostic criteria for plants and soils. Univ. of Calif., Div. of Agric. Sci., Riverside, Calif.

- Jaynes, D.B. (1996). Improved soil mapping using electromagnetic induction surveys. Precision Agriculture, Proceedings of the 3rd International Conference, Ed. P.C. Robert, R.H. Rust and W.E. Larson. p. 169-179.
- Johnson, G. V., and P. E. Fixen. (1990). Testing soils for sulphur, boron, molybdenum, and chlorine. P. 265- 273 In R. L. Westerman (ed.), Soil testing and plant analysis, 3rd ed., Soil Sci. Soc. Am. Madison. WI, USA.
- Jones, J.B. Jr. (1972). "Plant tissue analysis for micronutrients." In Micronutrients in agriculture Eds. J.J. Mortvedt, P.M. Giordano and W.L. Lindsay. Soil Sci. Soc. Am. Inc., Madison, WI. pp. 265-288.
- Jones, J.B., Jr. (1967). Interpretation of plant analysis for several agronomic crops. p. 49-58. In G.W. Hardy (ed.). Soil testing and plant analysis, part II. SSSA Spec. Pub. No. 2, Madison, Wis.
- Jones, J.B., Jr., and H.J. Mederski. (1964). Effect of lime and soil moisture level on the mineral composition of field grown soybean plants. Agron. Abstr. p. 32. Amer. Soc. Agron., Madison, Wis.
- Jones, Jr., J. B. (1991). Kjeldahl method for nitrogen determination. Micro-Macro Publishing Inc., Athens, GA, USA.
- Jones, Jr., J. B., B. Wolf, and H. A. Mills. (1991). Plant analysis handbook. Micro-Macro Publishing, Inc., Athens, GA, USA.

Kalra, Y. P., and D. G. Maynard. (1991). Methods Manual for Forest Soil and Plant Analysis. Forestry Canada, Northwest Region, Edmonton, Alberta, Canada. Inf. Rep. NOR-X-319.

- Kamphake, L. J., S. A. Hannah, and J. M. Cohen. (1967). Automated analysis for nitrate by hydrazine reduction. Water Research 1: 205 – 216.
- Katyal, J. C., and B. D. Sharma. (1980). A new technique of plant analysis to resolve iron chlorosis. Plant Soil 55: 105 – 119.
- Kausar, M. A., M. Tahir, and A. Hamid. (1990). Comparison of three methods for the estimation of soil available boron for maize. Pakistan J. Sci. Ind. Res. 33: 221-224.
- Keeney, D. R., and D. W. Nelson. (1982). Nitrogen-inorganic forms. P. 643-698. In A. L. Page (ed.), Methods of soil analysis. Agron. 9, Part 2: Chemical and microbiological properties, 2nd ed., Am. Soc. Agron., Madison, WI, USA.
- **KEFRI** (1992). "A Dry land Forestry Handbook for Kenya". Kenya Forestry Research Institute, Nairobi, pp.95.
- Keisling, T.C., D.A. Lauer, M.E. Walker, and R.J. Henning. (1977). Visual, tissue, and soil factors associated with Zn toxicity of peanuts. Agron. J. 69: 765-769.
- Keisling, T.C., L.F. Thompson, and W.R. Slabaugh. (1984). Visual symptoms and tissue manganese concentrations associated with manganese toxicity in wheat. Comm. Soil Sci. Plant Anal. 15: 537-540.

Kenworthy, A.L. (1961). Interpreting the balance of nutrient-element in leaves of fruit

trees. p. 28-43. In Walter Reuther (ed.). Plant analysis and fertilizer problems. Amer. Inst. Biol. Sci., Pub. No. 8, Washington, D.C.

- Keren, R., and F. T. Bingham. (1985). Boron in water, soils, and plants. Adv. Soil Sci. 1: 229-276.
- Klute, A.. (1986). Methods of soil analysis, Agron. 9, Part 1: Physical and mineralogical methods. Am. Soc. Agron., Madison, WI, USA.
- Kokwaro, J.O. (1976). "Medicinal plants of East Africa". East African Literature Bureau, Nairobi-KENYA pp.123-126.
- Kresge, C.B., and S.E. Younts. (1962). Effect of various rates and frequencies of potassium application on yield and chemical composition of alfalfa and alfalfa-orchard grass. Agron. J. 54: 313-316.
- Kubota, J., and Allaway, W.H. (1972). In "Micronutrients in Agriculture." (J.J.) Mortvedt, P.M. Giordano and W.L. Lindsay eds.) Soil Sci.Soc. Amer. Inc., Madison, pp 525 -554.
- Lal, R. (1996). "Deforestation and land use effects on soil degradation and rehabilitation in Western Nigeria. 111. Run off, soil erosion and nutrient loss. Land Degrad. Dev. 7: 87-98.

Lal, R. (2000). "Soil management in the developing countries". Soil Sci. 165: 57-72

- Lal, R. and Greenland, D.J. (1979). "Soil physical properties and crop production in the Tropics". John Wiley and Sons, New York, pp.7.
- Landon, J. R. (1991). Booker Tropical Soil Manual. A handbook for soil survey and agricultural land evaluation in the tropics and sub tropics. Longman Scientific & Technical. Hong Kong.
- Lindsay, W. L, and W. A. Norvell. (1978). Development of a DTPA soil test for zinc, iron, manganese and copper. Soil Sci. Soc. Am. J. 42: 421-428.

Loneragan, J.F. (1968). "Nutrient requirements of plants". Nature, 220, 1307-8.

Loneragan, J.F. and Snowball, K. (1969). "Calcium requirements of plants". Aust. J. Agr. Res., 20,465.

- Loneragan, J.F., Snowball, K. and Simmons, W.J. (1968). "Response of plants to calcium concentration in solution culture". Aust.J. Agric., <u>19</u>, 845-857.
- Lowe, R. H., and Massey, H. F. (1965). Hot water extraction for available soil molybdenum. Soil Science 100, 238-43.
- Lucas, R.E., and Knezek, B.D. (1972). In "Micronutrients in Agriculture". (J.J. Mortvedt, P.M. Giordano and W.L. Lindsay, eds.) Soil Sci. Soc. Amer. Inc., Madison, pp. 265-88.
- Ludwick, A.E. (1995). Western Fertilizer Handbook, 8th ed. Soil Improvement

Committee, California Fertilizer Association. Interstate Publ., Dancille, IL, USA.

- Luke, Q. (1994). Conservation of the Coastal forests: the case of the Kayas. In: Second Kenya National Workshop on Plant Genetic Resources, 15-17 February, 1994 National Museums of Kenya, Nairobi.
- Lusigi, W. J. (1984), Integrated project in Arid lands. (IPAL), Technical Report Number A- 6.
- Lund, E.D., C.D. Christy and P.E. Drummond. (1999). Practical applications of soil electrical conductivity mapping. In: Precision Agriculture '99, Proc. of the 2nd European Conf. on Precision Agriculture, Ed. J.V. Stafford. p. 771-779.
- Mahler, R. L., D. V. Naylor, and M. K. Fredrickson. (1984). Hot water extraction of boron from soils using sealed plastic pouches. Comm. Soil Sci. Plant Anal. 15:479-492.
- Marschner, J. (1986). "Mineral Nutrition of Higher Plants". pp. 289-300, Academic Press, London.
- Martens, D.C, and W. L. Lindsay. (1990). Testing soils for copper, iron, manganese, and zinc.
 P. 229-264. In R. L. Westerman (ed.), Soil testing and plant analysis, 3rd ed. Soil Sci. Soc. Am., Madison, WI, USA.

Martin, W.E., and J.E. Matocha. (1973). Plant analysis as an aid in the fertilization of forage Crops. p. 393-426. In L.M. Walsh and J.D.Beaton (eds.). Soil testing and plant analysis. Soil Sci. Soc. Amer., Inc., Madison, WI

Mascagni, J.J., Jr., and F.R. Cox. (1984). Diagnosis and correction of manganese deficiency in corn. Comm. Soil Sci Plant Anal. 15(11): 1323-1333.

Mascagni, H. J., Jr. and F. R. Cox. (1985). Calibration of a Manganese availability index for soybean soil test data. Soil Sci. Soc. Am. J. 49:382-386.

Mastalerz, J.W. (1977). The greenhouse environment. John Wiley & Sons, New York, N.Y. p.510-516.

Matar, A., P. N. Soltanpourand A. Chouinard (ed.). (1988). Soil Test Calibration in West Asia and North Africa. Proc. Second Regional Workshop. Ankara, Turkey, Sept 1 – 7, 1987. ICARDA, Aleppo, Syria.

McBride, M. (1994). Environmental chemistry of soils. Oxford University Press, New York.

McBride, R.A., A.M. Gordon and S.C. Shrive. (1990). Estimating forest soil quality from terrain measurements of apparent electrical conductivity. Soil Sci. Soc. Amer. J. 54:290-293.

McGill, D, and Figueiredo. (1993). Total Nitrogen, p. 201 – 211. In M. R. Carter (ed.), Soil sampling and methods of analysis. Lewis Publ., Boca Raton, FL, USA.

McKeague, J. A. (1978). Manual on soil sampling methods of analysis. Canadian Society of Soil Science: 66-68. McLean, E. O. (1982). Soil pH and lime requirement. P. 199 – 224, In A.L. Page (ed.), Methods of soil analysis, Part 2: chemical and microbiological properties. Am. Soc. Agron, Madison, WI, USA.

- Mengel, K. and Kirkby, E. A. (1982). Principles of Plant Nutrition. 3rd ed. Int. Potash Inst. Bern, Switzerland.
- Meudt, W.J. (1971)." Interactions of sulphite and manganeous ion with peroxidase oxidation products of Indole-3- acetic acid". Phytochemistry, <u>10</u>, 2103-9.
- Meredith, M.P. and S.V. Stehman. (1991). Repeated measures experiments in Forestry. CJFR, 21: pp 957-965.
- Miller, R. W., and R. L. Donahue. (1992). Soils: An introduction to soils and plant growth. 6th ed. Prentice Hall of India, New Delhi.

Mitchell, R. L. (1964). "Trace elements in soils". In chemistry of soil. Ed. F.E. Bear. ACS monograph No. 160. Reinhold Publishing crop. New York.

- Morgan, P.W. Joham, H.E. and Amin (1966). "Effect of Manganese toxicity on Indole acetic and oxidase system of cotton". Plant physiol., 41,718-724.
- Morgan, P.W. Taylor, D.M., Joham, H.E. (1976). "Manipulations of I.A.A-Oxidase activity and auxin deficiency"symptoms in intact cotton plants with manganese nutrition". Physiol. Plant, 37, 149-56.

Muhammed, S. (1996). Soil salinity, sodicity and water logging. P. 472-506. In A. Rushid and K.S. Memon (Managing Authors). Soil Science. National Book Foundation, Islamabad, Pakistan.

Munson, R. D, and W. L. Nelson. (1990). Principles and practices in plant analysis. P. 223 – 248. In R. L. Westerman (ed.), Soil testing and plant analysis, 3rd ed. Soil Sci. Soc. Am., Madison, WI, USA. Munson, R.D. (1969). Plant analysis: Varietal and other considerations. p. 85-104. In F. Greer

(ed.). Proceeding symposium on plant analysis, International Min. & Chem. Corp., Skokie, Ill.

Murphy, J., and J. P. Riley. (1962). A modified single solution method for determination of phosphate in natural water. Analytic Chimica Acta. 27: 31-36.

- Mwendwa, A.K., Otsamo, A. and Otsamo, R. (1993). "Effect of Irrigation on soil nutrient status on a 14 month old Prosopis juliflora and Eucalyptus microtheca plantation in Bura Irrigation Scheme, eastern Kenya". E. Africa Agric. For. J., <u>58</u>(Special Issue), 101-105.
- National Academy of Science (NAS). (1979). Tropical Legumes, resource for the future Washington DC 331 pp.
- Nelson, D.W. and L.E. Sommers. (1996). Total Carbon, Organic Carbon, and Organic Matter. p. 961-1010. In D.L. Sparks (ed.) Methods of Soil Analysis. Part 3. Chemical Methods. Soil Science Society of America Book Ser. 5. SSSA and ASA, Madison, Wis.
- Nicholaides, J.J., H.R. Chancy, L.H. Nilson, and J.E. Shelton. (1985). Snap bean grade and yield response to N rate and time of application and P and K rate. Comm. Soil

Sci. Plant Analy. 16(7): 741-757.

⁰duori, S.M. (1991). "Land use in Keiyo Marakwet district". Department of Resource Surveys and Remote Sensing, Ministry of Planning and National Department, Nairobi.

- Ohki, K. (1976). Manganese deficiency and toxicity levels for 'Bragg' soybeans. Agron. J. 68: 861-864.
- Okalebo, J. R., (1985). "A simple wet ashing technique of P, K, Ca and Mg analysis of plant tissue in a single digest". Kenya J. Sci. and Technol., <u>B6</u>, 129-133.
- Okalebo, J. R., Gathua, K. W. and Woomer, P. L. (2002). Laboratory methods of soil and plant analysis: A working manual. Second Edition. TSBF-CIAT and SACRED Africa, Nairobi, Kenya.
- Olsen, S. R., and L. W. Sommers. (1982). Phosphorus. p. 403-430. In A. L. page (ed.), Methods of soil analysis, Agron. No. 9 Part 2: Chemical and microbiological properties, 2nd ed., Am. Soc. Agron., Madison, WI, USA.
- Olsen, S. R., C. V. Cole, F. S. Watanabe, and L. A. Dean. (1954). Estimation of available phosphorus in soils by extraction with sodium bicarbonate. U. S. Dep. Agric. Circ. 939, USA.
- Ominde, S.H. (1988). "Kenya's population growth and development to the year 2000 AD". Heinemann Kenya, Nairobi.
- Page, A. L. (1982). Methods of soil analysis, Agron. 9 Part 2: Chemical and mineralogical properties 2nd ed., am. Soc. Agron., Madison WI, USA.
- Parker, M.B., and M.E. Walker, (1986). Soil pH and manganese effects on manganese nutrition of peanut. Agron. J. 78(4): 614-620.
- Parker, M.B., F.C. Boswell, K. Ohki, L.M. Shuman, and D.O. Wilson. (1981). Manganese effects on yield and nutrient concentration in leaves and seed of soybean cultivars. Agron. J. 73: 643-646.

- Peverill, K. I., Sparrow, L. A. and Reuter, D. J. (1999). Soil analysis: an Interpretational Manual.SCIRO Australia.SCIRO publishers.
- Pombo, German I., and April Smith. (1986). Growth and nutrient interrelationships of three vegetable crops with different sensitivities to soil pH as affected by lime and fertilizer treatments. Comm. Soil Sci. Plant Anal. 17(3): 353-368.
- Ponnamperuma, F. N., M. T. Caytan, and R. S. Lantin. (1981). Dilute hydrochloric acid as extractant for available zinc, copper and boron in rice soils. Plant Soil. 61: 297-310.
- Rains, D.W. (1976). In Plant Biotechemistry (eds. J. Bonner) and J.E. Varner), Academic press, pp.561-567.
- Rashid, A., E. Rafique, and N. Bughio. (1997). Micronutrient deficiencies in calcareous soils of Pakistan. III. Boron nutrition of sorghum. Commun. Soil Sci. Plant Anal. 28: 441-453.
- Rashid, A., E. Rafique, and N. Bughio. (1994). Diagnosing boron deficiency in repressed and mustard by plant analysis and soil testing. Commun. Soil Sci. Plant Anal. 25: 2883-2897.

Republic of Kenya (1989a). Development Plan 1989-1993. Government Printer, Nairobi.

Republic of Kenya (1989b). Statistical abstract, Central Bureau of Statistics, Ministry of

Planning and National Development Government Printer, Nairobi.

- Reuter, D. J., and J. B. Robinson (ed.). (1986). Plant analysis: An Interpretation manual. Inmate Press, Melbourne, Australia.
- Reuter, D. J., and J. B. Robinson. (1997). Plant analysis: An Interpretation manual. 2nd ed. CSIRO Publ., Australia.

Reuter, D.J. (1975). In "Trace Elements in soil plant Animal systems". (D.J.D. Nicholas, A.R. Egan, eds.), Academic press, New York, pp.291-323.

- Rhoades, J. D. (1982). Cation exchange capacity. P. 149- 157. In A. L. Page (ed.), Methods of soil analysis, Agron. No. 9, Part 2: Chemical and mineralogical properties. Am. Soc. Agron, Madison, WI, USA.
- Rhoades, J. D., and Polemio, M. (1977). Determining cation exchange capacity: A new procedure for calcareous and gypsiferous soils. Soil Sci. Soc. Am. J.41: 524 300.
- Rhoades, J.D. (1982). "Methods of Soil Analysis, Part 2". Second Edition (A.L. Page, R.H. Miller and D.R. Keeney, Eds.) American Society of Agronomy. Inc. Madison.
- Rhoades, J.D. and D.L. Corwin. (1981). Determining soil electrical conductivitydepthrelations using an inductive electromagnetic soil conductivity meter. Soil Sci. Soc. Amer. J. 45:255-260.
- Rhue, R.D., and P.H. Everett. (1987). Response of tomatoes to lime and phosphorus on a sandy soil. Agron. J. 79: 71-77.
- Richards, L. A. (1954). Diagnosis and improvement of saline and alkali soils. USDA Agric. Handbook 60. Washington, D.C. potential. Commun. Soil Sci. Plant Anal. 31(11-14): 2147-2154.
- Ryan, J., and A. Matar. (1990). Soil Test Calibration Workshop in West Asia North Africa. Proc. 3rd Regional Workshop. Amman., Jordan, Sept. 3 –9, 1988. ICARDA, Aleppo, Syria.

Ryan, J., and A. Matar. (1992) Fertilizer use efficiency under rain-fed agriculture in West Asia and North Africa. Proc. 4th Regional Soil Test Calibration Workshop in West Asia- North Africa Region. Agadir, Morocco, 5 – 10, 1991. ICARDA, Aleppo, Syria.

- Ryan, J., and S. Garabet. (1994). Soil test standardization in West Asia North Africa region. Commun. Soil Sci. Plant Anal. 25 (9 &10): 1641-1653.
- Ryan. J, S. Garabet, A. Rashid, and M. El-Gharous. (1999). Assessment of soil and plant analysis laboratories in the West Asia – North Africa region. Commun. Soil Sci. Plant Analysis. 30: 885 – 894.
- Sanchez, P. A. and C. A. Palm. (1996). Nutrient cycling and agro forestry in Africa. Unasylva 185 (47): 24 - 28.
- Sanchez, P. A., R. J. Buresh and R. R. B. Leakey. (1997). Trees, soils and food security. Philes. Trans. R. Soc. London B 352: 949 - 961.
- Sanchez, P.A. (1976). "Properties and management of soils in the Tropics". John Wiley and Sons, New York, pp 96-97.

SAS (2003). Inc. SAS Users Guide Basics. SAS Institute, Inc. Cary, North Carolina, USA

- Sayegh, A. H., N. A. Khan, P. Khan, and J. Ryan. (1978). Factors affecting gypsum and cation exchange capacity determinations in gypsiferoussoils. Soil Sci. 125: 294-300.
- Shuman, L. M., V. A. Bandel, S. J. Donahue, R. A. Isaac, R. M. Lippert, J.T. Sims, and M. R. Tucker. (1992). Comparison of Mehlich 1 and Mehlich 3 extractable soil boron with hot-water extractable boron. Commun. Soil Sci. Plan Anal. 23:1-4.
- Sims, J. R., and G. D. Jackson. (1971). Rapid analysis of soil nitrate with chromotropic acid. Soil Sci. Am. Proc. 35: 603 606.

Smaling, E. M. A. (1993). Soil nutrient depletion in sub-Saharan Africa. P. 53 - 67. In: Van Reuter, H. and Prims, W.H. (ed). The role of plant nutrients and sustainable food production in sub-Saharan Africa. Plonsen & Looijen, Wageningen, The Netherlands.

Smith, P.F. (1962). Mineral analysis of plant tissues. Ann. Rev. Plant Physiol. 13:81-108.

Smith, P.F. and Specht, A.W. (1953). Plant physiol., 28, 371-82.

- Soltanpour, P. N. (1985). Use of ammonium bicarbonate-DTPA soil test to evaluate elemental availability and toxicity. Commun. Soil Sci. Plant Anal. 16: 323-338.
- Soltanpour, P. N., and A. P. Schwab (ed.). (1977). A new soil test for simultaneous extraction of macro and micronutrients in alkaline soils. Commun. Soil Sci. Plant Anal. 8: 195-207.
- Sombroek, W.G., Brawn, H.M.H. and Van der Pouw B.J.A. (1982). "The exploratory soil Map and agro-climatic zone map of Kenya". Exploratory Soil Survey Report E1. Kenya Soil Survey, Nairobi.
- Sparks, D. (1978). Nutrient concentrations of pecan leaves with deficiency symptoms and normal growth. Hort. Science 13(3): 256-257.

Sparks, D. (1986). Nitrogen effects on pecan yield and nut growth - A re-evaluation. Proc. Ga. Pecan Growers Assoc. 17:18-25.

Spiers, James M. (1978). Effects of pH level and nitrogen source on elemental leaf content of `Tiftblue' rabbiteye blueberry. J. Amer. Soc. Hort. Sci. 103(6): 705-708. Spiers, James M. (1983). Influence of N, K, and Na concentration on growth and leaf element content of "Tiftblue' rabbiteye blueberry. Hort Science 18(2): 223-224.

Spiers, James M., (1982). Seasonal variation of leaf nutrient composition in 'Tiftblue' rabbiteye blueberry. J. Amer. Soc. Hort. Sci. 107(2): 255-257.

Stevenson J (1998). Humus Chemistry in Soil Chemistry p148 Wiley Pub. NY

Stevenson F.J. (1986). "Cycles of soil carbon, Nitrogen, phosphorus, Sulphur and Micronutrients". John Willey and Sons, New York pp. 380.

Stout, P.F. (1961). Proc. 9th Annu. Calif. Fert. Conf., pp.21-23.

- Sudduth, K.A., N.R. Kitchen and S.T. Drummond. (1998). Soil conductivity sensing on clay pan soils: Comparison of electromagnetic induction and direct methods. Proc. of the 4th International Conference on Precision Agriculture. Ed. P.C. Robert. P. 979-990.
- Tandon, H. L. S. (1991). Sulphur research and agricultural production in India. 3rd ed., The Sulphur Institute, Washington, D.C. USA.

Tandon, H. L. S. (1993). Methods of analysis of soils, plants, waters and fertilizers.
Fertilizer Development and Consultation Organization, New Delhi, India.

Tekalign, T., I. Haque, and E. A. Aduayi. (1991). Soil, Plant, water, fertilizer, animal manure, and compost analysis manual. Plant Science Division Working Document 13: ILCA, Addis Ababa, Ethiopia.

Terry, N. and Ulrich, A. (1973). "Effects of phosphorus deficiency on photosynthesis and respiration of leaves in sugar beet". Plant physiol., 54, 379- 381.

- Tiffin, L.O. (1970). "Translocation of iron citrate and phosphorus in xylem exudates of soybean." Plant physiol., 45 280-283.
- Tisdale, S.L., Werner, L. N. and Beaton, D.J. (1985). "Soil Fertility and Fertilizers", Macmillan, New York.

Ulrich, A. (1961). Plant analysis in sugarbeet nutrition p. 190-211. In WalterReuther(ed.).

Plant analysis and fertilizer problems. Amer. Inst. Biol. Sci. Pub. No. 8, Washington, D.C.

Van Engelen, V.W.P., (1983). Detailed Soil Survey of Chemeron Irrigation Scheme, Soil Survey Report NO. D32, Baringo district. Kenya Soil Survey, Nairobi.

- Vance, E. D., Brooks, P. C., and Jenskinson, D. S. (1987). An extraction method for measuring soil microbial biomass Carbon. Soil Biol. and Biochem. 19:703-707.
- Verma, B. C. (1977). An improved turbidimetric procedure for the determination of sulphate in plants and soils. Talanta 24: 49-50.
- Waas, P. (1995). Kenya's Indigenous forest status, management and conservation, ICUN Forest Conservation Programme.

Wahle, K.W.J. and Davies, N.T. (1977). Involvement of copper in microsomal mixed-function oxidases reactions: A review. J. Sci. Food Agric. 28, 93-97.

Walker, C.D. and Webb, J. (1981). Copper on plants, forms and behaviour. In "Copper in Soils and Plants". (J.F. Loneragan, A.D. Robson and R.D. Graham, eds.), Academic Press, London and Orlando pp.189-212.

- Walker, D.W., and W.R. Woodson. (1987). Nitrogen rate and cultivar effects on nitrogen and nitrate concentration of sweet potato leaf tissue. Comm. Soil Sci Plant Anal. 18(5): 529-541.
- Walkley A. (1947), "A critical examination of a rapid method for determining organic carbon in soils-effect of variations on digestion conditions and of inorganic soil constituents", Soil Sci, <u>63</u>, 251-254.
- Wallace A. (1980). "Effect of excess chelating agent on micronutrient concentrations in bush beans grown in solution culture." J. Plant Nutr., <u>2</u>, 163-170.
- Walsh, A. (1955). Determination of Trace elements by Atomic Absorption Spectrometry. Spectrochim, Acta, 7, 108.
- Walsh, L.M. and J.D. Beaton. (1973). "Soil Testing and plant Analysis". Soil Sci. Soc. Am. Inc., Madison, Wisconsin. pp.491.
- Walsh, L.M., W.H. Erhardt, and H.D. Seibel. (1972). Copper toxicity in snap beans (Phaseolus vulgaris L.). J. Environ. Qual. 1: 197-200.
- Watanabe, F. S., and S. R. Olsen. (1965). Test of an ascorbic acid method for determining phosphorus in water and NaHCO₃ extracts from soil. Soil Sci. Soc. Am. Proc. 29: 677- 678.
- Watson, M. E. and J. R. Brown. (1998). pH and Lime Requirement. Ch. 4. In J. R. Brown (ed.). Recommended Chemical Soil Test Procedures for the North Central Region.N.C. Reg. Pub. 221 (Revised) (Mo. Agric. Exp. Stn. SB 1001).
- Weinstein, L.H. and Robbins, W.R. (1955). "The effect of different iron and manganese nutrient levels on the catalase and cytochrome oxidase activities on green and albino sunflower leaf tissue". Plant physiology, <u>30</u>, 27-32.

Welz .B. (1985). Atomic Absorption Spectrometry, 2nd Ed., VCH Publishers

Westerman, R. L. (1990). Soil testing and plant analysis. 3rd ed. Soil Sci. Soc. Am. Madison, WI, USA.

 Whitney, D. A. (1988). Micronutrient soil tests for zinc, iron, manganese and copper. p. 20-22.
 In W. C. Dahnke (ed.) Recommended Chemical Soil Test Procedures for the North Central Region. North Dakota Agric. Expt. Stn. Bull. No. 499.

Wild, A. (1988). "Russell's Soil Conditions and plant growth". Eleventh Edition, Longman group UK LTD, London pp.797.

- Williams, C. H., and A. Steinbergs. (1959). Soil sulphur fractions as chemical indices of available sulphur in some Australian soils. Aust. J. Agric. Res. 10: 340-352.
- Wilson, D.O. (1977). "Influence of added manganese on nitrification in a low Mn soil". Plant soil. <u>46</u>, and 678-70.

Wilson, D.O., F.C. Boswell, K. Ohki, M. B. Parker, and L.M. Shuman (1981). Soil distribution and soybean plant accumulation of manganese in manganese deficient and manganese-fertilized field plots. Agron. J. 45: 549-552.

Woolhouse, H.W. (1983). Toxicity and tolerance in the responses of plants to metals. In: Lange OC, Nobel PS, Osmond CB, Ziegler H, eds. Encyclopedia of plant physiology II. <u>12B</u>. New York, NY, Springer-Verlag, pp. 245– 300.

Worley, R.E. (1985). Use of leaf analysis for basing N application for Stuart pecans. Proc. S.E. Pecan Growers Assoc.78: 79-83.

APPENDICES

SOIL ANALYSIS

APPENDIX I: Raw data on Soil chemical analysis during dry and wet seasons in sites

R.

Codes used soil sampling in the four study sites:

ST – Solit

KK – Kimorok

KN – Kapkun

MI – Maoi

SS – Surface samples at depth between 0 - 25 cm

M1–Soil sample number one from medium elevation

L1-Soil sample number one from low elevation

SOLIT DRY SEASON – Soil samples

<u>Senders ref.</u>	H ₂ O	CaCl ₂	Ec	С	Ν	C:N	SOM	Р	K	Ca	Mg	Na	Soil type
	pН	pН	mS/cm	%	%	Ratio	%	ppm	ΕB	ases C m	ol (†)/Kg	5	
SSM1/STM1	7.9	7.3	0.09	0.2	0.15	1.60	0.41	12	0.2	1.29	4.18	0.79	Caclic xerosols
SSM2/STM2	8.1	7.6	0.07	0.4	0.15	2.33	0.60	12	0.46	1.26	4.14	0.61	Caclic xerosols
SSM3/STM3	8.2	7.1	0.07	0.4	0.11	3.64	0.69	15	0.36	1.33	4.14	0.61	Caclic xerosols
SSM4/STM4	7.9	7.2	0.08	0.5	0.15	3.20	0.83	12	0.41	1.3	4.28	0.53	Caclic xerosols
SSM5/STM5	8.0	7.7	0.05	0.4	0.11	3.45	0.66	12	0.25	1.28	6.26	0.48	Caclic xerosols
SSM6/STM6	8.0	7.7	0.05	0.3	0.08	3.88	0.53	11	0.41	1.33	6.32	0.7	Caclic xerosols

SSM7/STM7	8.1	7.1	0.06	0.3	0.11	2.45	0.47	16	0.23	1.23	6.78	0.75	Caclic xerosols
SSM8/STM8	7.9	7.3	0.05	0.6	0.19	2.95	0.97	12	0.46	1.01	3.74	0.75	Caclic xerosols
SSM9/STM9	8.1	7.2	0.06	0.4	0.11	3.64	0.69	13	0.36	1.07	4.24	0.66	Caclic xerosols
SSM10/STM10	8.1	7.2	0.07	0.5	0.11	4.09	0.78	13	0.25	1.09	3.67	0.79	Caclic xerosols
SSM11/STM11	7.5	7.2	0.08	0.5	0.19	2.37	0.78	16	0.41	1.31	4.01	0.83	Caclic xerosols
SSM12/STM12	8.1	7.2	0.05	0.4	0.23	1.57	0.62	14	0.5	1.3	4.1	0.7	Caclic xerosols
SSM13/STM13	8.4	7.3	0.05	0.4	0.15	2.80	0.72	14	0.41	1.35	4.36	1.21	Caclic xerosols
SSM14/STM14	8.0	7.1	0.04	0.6	0.19	3.16	1.03	14	1.22	0.63	4.32	0.7	Caclic xerosols
SSM15/STM15	8.1	7.3	0.06	0.6	0.15	3.80	0.98	14	1.27	0.6	4.82	0.18	Caclic xerosols
SSL1/STL1	7.6	7.1	0.06	0.6	0.11	5.45	1.03	14	0.41	1.02	3.95	0.13	Caclic xerosols
SSL2/STL2	7.7	7.05	0.09	0.5	, 0.11	4.45	0.84	13	0.31	0.97	3.76	0.31	Caclic xerosols
SSL3/STL3	7.1	7.1	0.05	0.4	0.11	3.82	0.72	13	0.41	1.1	3.8	0.13	Caclic xerosols
SSL4/STL4	8.0	7.1	0.08	0.3	0.11	2.82	0.53	12	0.25	0.67	2.85	0.13	Caclic xerosols
SSL5/STL5	7.7	7.3	0.06	0.3	0.34	0.94	0.55	24	0.31	0.89	6.87	0.18	Caclic xerosols
SSL6/STL6	7.7	7.05	0.06	0.6	0.11	5.27	1.00	22	0.46	0.92	3.75	0.44	Caclic xerosols
SSL7/STL7	7.8	7.05	0.06	0.4	0.11	3.27	0.62	12	0.36	1.35	4.38	0.18	Caclic xerosols
SSL8/STL8	7.2	7.1	0.05	0.4	0.26	1.42	0.64	11	0.15	1.14	4.38	0.35	Caclic xerosols
SSL9/STL9	8.0	7.1	0.06	0.4	0.11	3.64	0.69	14	0.2	1.39	3.73	0.09	Caclic xerosols
SSL10/STL10	7.8	7.1	0.06	0.4	0.19	2.00	0.66	9	0.2	1.38	3.53	0.26	Caclic xerosols
SSL11/STL11	7.9	7.1	0.06	0.7	0.26	2.50	1.12	13	0.25	1.15	1.58	0.22	Caclic xerosols
SSL12/STL12	7.9	7.05	0.05	0.5	0.26	2.00	0.90	11	0.25	1.46	2.63	0.31	Caclic xerosols
SSL13/STL13	8.1	7.05	0.05	0.6	0.19	3.05	1.00	16	0.25	1.52	3.19	0.31	Caclic xerosols
SSL14/STL14	8.2	7.1	0.05	0.6	0.11	5.55	1.05	13	0.25	1.51	3.97	0.22	Caclic xerosols
SSL15/STL15	8.0	7.5	0.06	0.2	0.19	1.21	0.40	13	0.2	1.51	5.82	0.22	Caclic xerosols
SSL16/STL16	8.1	7.4	0.05	0.6	0.49	1.16	0.98	11	0.36	1.35	3.15	0.31	Caclic xerosols
SSL17/STL17	8.2	7.1	0.06	0.3	0.34	0.79	0.47	11	0.25	1.12	2.48	0.22	Caclic xerosols

STATIST.	-		P.C		N	CH19	ARKIN	*	P.	Ca	Mg	Letw	Soll type
	pH	рН	µs/cm	%	%	Ratio	%	ppm	EI	Bases C	mol (`) /	Ҡg	
ST01/SS	7.2	7.05	50	0.6	0.19	3.16	1.03	19	0.5	40.1	5.9	1.8	Caclic xerosols
ST02/SS	7.1	6.6	100	0.5	0.34	1.47	0.86	3	0.4	46.9	5.4	1.7	Caclic xerosols
ST03/SS	7.4	7.1	121	0.8	0.34	2.35	1.38	22	1.0	34.8	6.0	1.9	Caclic xerosols
ST04/SS	7.1	6.7	68	0.5	0.11	4.09	0.78	4	1.1	21.5	5.8	1.7	Caclic xerosols
ST05/SS	7.4	7.1	85	0.9	0.19	4.84	1.59	4	0.4	47.05	9.9	1.7	Caclic xerosols
SS/ST/1	7.3	7.1	120	0.3	0.19	1.79	0.59	4	0.7	37.7	9.6	1.8	Caclic xerosols
SS/ST/2	7.2	7.05	112	0.1	0.34	0.35	0.21	6	0.5	39.1	10.2	1.8	Caclic xerosols
SS/ST/3	7.2	6.9	67	0.4	0.26	1.58	0.71	31	0.8	50.9	7.05	2.5	Caclic xerosols
SS/ST/4	7.3	6.9	76	0.2	0.11	2.00	0.38	3	0.5	30.5	9.9	1.8	Caclic xerosols
SS/ST/5	7.3	6.9	37	0.1	0.19	0.26	0.09	4	0.2	17.8	9.3	1.9	Caclic xerosols
SS/ST/6	7.3	7.05	41	0.8	0.19	4.05	1.33	5	1.5	28.3	6.2	2.2	Caclic xerosols
SS/ST/7	7.1	6.6	59	0.8	0.19	4.26	1.40	43	0.9	16.3	7.05	1.8	Caclic xerosols
SS/ST/8	7.1	6.8	68	0.2	0.34	0.62	0.36	2	0.3	29.6	6.2	1.8	Caclic xerosols
SS/ST/9	6.9	6.7	72	0.1	0.19	0.26	0.09	2	0.1	64.3	6.5	1.9	Caclic xerosols
SS/ST/10	7.1	6.9	90	0.1	0.19	0.63	0.21	3	0.8	30.0	7.6	1.8	Caclic xerosols
ST/01/SS	7.6	6.7	61	0.5	0.08	5.63	0.78	19	0.3	68.8	14.7	2.2	Caclic xerosols
ST/02/SS	7.4	7.1	6	0.3	0.15	2.07	0.53	3	0.5	24.9	16.0	3.0	Caclic xerosols
ST/03/SS	7.3	6.5	34	0.6	0.23	2.44	0.95	3	0.5	14.3	5.9	1.9	Caclic xerosols
ST/04/SS	8.2	7.3	154	0.4	0.08	5.47	0.71	14	0.4	19.9	10.6	6.6	Caclic xerosols
ST/05/SS	7.1	6.8	70	0.6	0.15	4.00	1.03	32	0.6	17.6	7.2	0.3	Caclic xerosols
ST/05/SS	7.1	6.8	50	0.8	0.15	5.53	1.43	3	0.9	37.8	8.9	2.0	Caclic xerosols
ST/06/SS	7.05	7.05	75	1.0	0.15	6.93	1.79	2	0.4	41.4	2.0	2.0	Caclic xerosols
ST/07/SS	7.3	6.7	16	0.4	0.15	2.93	0.76	2	0.4	15.9	7.3	0.3	Caclic xerosols
ST/08/SS	7.1	7.05	33	0.6	0.08	7.73	1.00	3	0.5	13.7	7.3	0.2	Caclic xerosols
ST/09/SS	7.4	6.9	50	0.4	0.15	2.73	0.71	10	0.7	18.5	7.1	0.5	Caclic xerosols
ST/10/SS	7.5	6.9	40	0.5	0.15	3.20	0.83	13	0.3	19.2	5.2	0.3	Caclic xerosols
ST/11SS	7.8	6.9	47	0.3	0.15	2.00	0.52	5	0.2	40.5	8.6	0.1	Caclic xerosols
ST/12/SS	7.6	7.2	86	0.3	0.15	1.80	0.47	4	0.3	50.2	6.0	0.1	Caclic xerosols
ST/13/SS	8.0	6.8	53	0.8	0.15	5.20	1.34	5	0.5	55.8	8.9	0.5	Caclic xerosols
ST/14/SS	7.3	6.9	68	0.5	0.15	3.53	0.91	17	0.2	37.1	9.5	0.2	Caclic xerosols
ST/15/SS	7.2	6.9	62	0.2	0.15	1.00	0.26	5	0.1	43.5	11.9	0.3	Caclic xerosols

SEABORN	10,00	-	liuc -	C	N	43324	BOM	1-	к	Ca	Mg	Tri sa	Soll type
Senders ref.	рН	рН	mS/cm	%	%	Ratio	%	ppm	EE	Bases c t	nol (`)/	Kg	T 1 1 / 1
KK1SS	6.8	6.2	0.07	0.56	0.04	14.00	0.97	12	1.38	5.87	1.85	0.87	Lithosols/xerosols
KK2SS	6.5	6.1	0.05	1.71	0.19	9.00	2.95	17	1.83	5.99	2.12	0.87	Lithosols/xerosols
KK3SS	6.5	5.9	0.05	1.47	0.19	7.74	2.53	16	3.16	14.5	2.1	1.29	Lithosols/xerosols
KK4SS	5.9	5.7	0.04	1.44	0.19	7.58	2.48	4	1.48	4.83	2.13	0.82	Lithosols/xerosols
KK5SS	6.5	5.8	0.05	1.76	0.11	16.00	3.03	15	1.53	5.36	2.42	0.82	Lithosols/xerosols
KK6SS	6.0	5.7	0.06	2.03	0.26	7.81	3.50	23	1.68	5.32	2.57	0.77	Lithosols/xerosols
KK7SS	5.9	5.8	0.04	1.09	0.19	5.74	1.88	26	1.73	6.55	2.08	0.87	Lithosols/xerosols
KK8SS	5.6	5.6	0.03	1	0.19	5.26	1.72	26	1.58	5.93	2.27	0.87	Lithosols/xerosols
KK9SS	6.4	6.1	0.03	1.21	0.11	11.00	2.09	8	1.14	5.73	2.1	0.82	Lithosols/xerosols
KK10SS	6.0	5.9	0.06	2.44	0.26	9.38	4.21	6	1.43	5.72	2.1	0.67	Lithosols/xerosols
KK11SS	5.8	5.7	0.04	0.94	0.11	8.55	1.62	7	1.23	5.65	2.42	0.82	Lithosols/xerosols
KK12SS	6.2	5.9	0.03	0.97	0.26	3.73	1.67	10	1.43	6.36	2.36	0.98	Lithosols/xerosols
KK13SS	6.0	5.8	0.04	0.76	0.11	6.91	1.31	7	1.58	5.48	2.08	0.87	Lithosols/xerosols
KK14SS	6.1	5.7	0.03	1.03	0.15	6.87	1.78	3	1.33	6.95	2.24	0.62	Lithosols/xerosols
KK15SS	6.2	5.7	0.03	1.23	0.08	15.38	2.12	3	1.32	5.7	2.65	0.82	Lithosols/xerosols
KK16SS	5.6	5.5	0.03	1.06	0.19	5.58	1.83	14	1.28	5.73	2.54	0.77	Lithosols/xerosols
KK17SS	6.2	5.8	0.04	1.32	0.19	6.95	2.28	12	1.33	6.28	2.37	0.62	Lithosols/xerosols
KK18SS	6.8	6.0	0.05	1	0.11	9.09	1.72	15	1.38	9.75	2.54	0.82	Lithosols/xerosols
KK19SS	6.5	5.8	0.03	1.12	0.15	7.47	1.93	12	1.48	7.53	2.27	0.77	Lithosols/xerosols
KK20SS	6.4	5.9	0.03	0.97	0.15	6.47	1.67	15	1.58	6.73	2.16	0.77	Lithosols/xerosols
KK21SS	6.3	5.9	0.04	1.3	0.11	11.64	2.21	2	1.63	7.057	2.16	0.87	Lithosols/xerosols
KK22SS	6.4	5.9	0.05	1.7	0.26	6.54	2.93	3	1.88	7.86	2.02	0.98	Lithosols/xerosols
KK23SS	6.4	5.7	0.03	1.1	0.19	5.79	1.90	2	1.23	6.55	2.22	0.77	Lithosols/xerosols
KK24SS	6.0	5.7	0.05	1.5	0.11	13.36	2.53	3	1.48	6.82	2.25	0.87	Lithosols/xerosols
KK25SS	6.4	6.0	0.03	1.0	0.19	5.05	1.66	3	2.12	5.88	2.06	0.87	Lithosols/xerosols
KK26SS	6.2	5.7	0.03	0.9	0.19	4.89	1.60	2	1.14	5.59	2.11	0.57	Lithosols/xerosols
KK27SS	6.6	5.5	0.03	1.2	0.19	6.47	2.12	2	1.38	6.26	2.23	0.67	Lithosols/xerosols
KK28SS	6.3	5.9	0.05	1.3	0.11	12.00	2.28	3	2.52	7.15	2.22	0.82	Lithosols/xerosols
KK29SS	6.3	5.7	0.04	0.9	0.19	4.63	1.52	13	2.02	6.08	2	0.82	Lithosols/xerosols
KK30SS	5.8	5.8	0.03	0.7	0.11	6.00	1.14	2	1.63	6.48	2.11	0.77	Lithosols/xerosols

KIMOROK	WET S	EASON	– Soil san	nples									
KK01/SS	6.5	6.3	41	1.1	0.11	9.55	1.81	10	0.9	4.9	4.5	0.2	Lithosols/xerosols
KK02/SS	6.4	6.2	55	0.9	0.11	8.36	1.59	18	0.7	4.5	4.6	0.2	Lithosols/xerosols
KK03/SS	6.6	6.4	85	1.0	0.09	10.89	1.69	20	-1	5.2	3.9	0.2	Lithosols/xerosols
KK04/SS	6.8	6.3	35	0.9	0.11	8.36	1.59	19	0.9	5.6	4.3	0.5	Lithosols/xerosols
KK05/SS	6.3	6.1	42	0.7	0.11	6.64	1.26	17	0.6	4.9	5	0.4	Lithosols/xerosols
KK06/SS	6.5	6.3	66	0.8	0.11	7.18	1.36	9	0.9	4.9	4.6	0.3	Lithosols/xerosols
KK07/SS	6.5	6.2	65	1.2	0.11	10.55	2.00	18	1.1	5.9	4.7	0.2	Lithosols/xerosols
KK08/SS	6.3	6.1	46	1.2	0.11	10.64	2.02	9	0.7	4.1	4.3	0.3	Lithosols/xerosols
KK09/SS	6.5	6.4	55	1.2	0.11	10.55	2.00	18	1	4.1	4.5	0.3	Lithosols/xerosols
KK10SS	6.1	6.0	61	1.2	0.09	13.33	2.07	16	1	3.5	5.5	0.2	Lithosols/xerosols
KK11/SS	6.3	6.1	60	1.6	0.11	14.64	2.78	35	0.9	3.3	4.2	0.2	Lithosols/xerosols
KK12/SS	6.4	6.1	58	1.6	0.11	14.91	2.83	26	0.7	4.9	4.3	0.4	Lithosols/xerosols
KK13/SS	6.7	6.3	40	1.4	0.11	12.91	2.45	29	1.2	4.1	4	0.3	Lithosols/xerosols
KK14/SS	6.6	6.3	39	1.2	0.11	10.91	2.07	24	1.1	4.3	4.3	0.2	Lithosols/xerosols
KK15/SS	6.7	6.2	38	0.6	0.19	3.11	1.02	15	0.7	4.7	4	0.4	Lithosols/xerosols
KK16/SS	6.8	6.3	42	0.9	0.11	8.27	1.57	8	0.6	5	4.7	0.2	Lithosols/xerosols
KK17/SS	6.9	6.3	50	1.1	0.11	9.73	1.84	24	1.4	5.8	3.7	0.2	Lithosols/xerosols
KK18/SS	6.6	6.3	40	0.8	0.04	19.50	1.34	14	0.7	5.4	4.5	0.3	Lithosols/xerosols
KK19/SS	6.9	6.5	54	0.9	0.11	8.27	1.57	34	1.1	5.8	4	0.3	Lithosols/xerosols
KK20/SS	6.5	6.4	57	1.1	0.15	7.60	1.97	16	0.9	4.8	4.2	0.3	Lithosols/xerosols
KK21/SS	6.9	6.4	56	1.3	0.09	14.33	2.22	21	0.8	4.8	3.8	0.3	Lithosols/xerosols
KK22/SS	6.9	6.3	41	1.1	0.11	9.73	1.84	8	0.8	4.3	4.4	0.3	Lithosols/xerosols
KK23/SS	6.7	6.2	35	0.8	0.09	9.11	1.41	11	0.5	3.4	4.3	0.2	Lithosols/xerosols
KK24/SS	6.8	6.2	38	0.9	0.09	9.56	1.48	12	0.7	3.7	4	0.4	Lithosols/xerosols
KK25/SS	6.9	6.3	55	1.2	0.11	10.91	2.07	21	0.8	5.4	4.1	0.3	Lithosols/xerosols
KK26/SS	6.8	6.5	47	1.0	0.04	25.00	1.72	27	1.2	6.7	4.2	0.3	Lithosols/xerosols
KK27/SS	6.5	6.3	83	1.7	0.19	9.11	2.98	28	1.2	6.1	4.1	0.3	Lithosols/xerosols
KK28/SS	6.5	6.3	39	1.2	0.11	11.27	2.14	20	0.9	5.1	3.4	0.5	Lithosols/xerosols
KK29/SS	6.6	6.3	48	1.0	0.11	8.91	1.69	31	0.9	4.7	3.7	0.3	Lithosols/xerosols
KK32/SS	6.8	6.2	41	0.8	0.26	3.00	1.34	13	0.8	7	5.3	0.4	Lithosols/xerosols
KK/04/SS	7.6	6.6	26	0.9	0.11	8.09	1.53	14	0.9	5.7	4.5	0.1	Lithosols/xerosols

KAPKUN DRY SEASON – Soil samples

Senders ref.	H ₂ O	CaCl ₂	Ec	С	Ν	C:N	SOM	Ρ	K	Ca	Mg	Na	Soil type
	pН	pН	mS/cm	%	%	Ratio	%	ppm	ΕB	E Bases C mol (⁺)/Kg			
SS1KN1	6.3	5.5	0.03	0.9	0.15	5.93	1.53	15	1.22	6.15	2.64	1.51	Ferralic
SS2KN2	5.9	5.4	0.05	1.3	0.23	5.60	2.17	10	1.73	5.26	2.36	1.93	Ferralic
SS3KN3	6.2	5.9	0.03	0.9	0.15	5.87	1.52	12	1.78	7.1	2.6	2.18	Ferralic
SS4KN4	5.9	5.1	0.04	1.2	0.23	5.29	2.05	17	1.73	5.88	2.45	2.18	Ferralic
SS5KN5	6.3	5.8	0.04	0.8	0.15	5.53	1.43	10	2.09	5.84	2.68	2.6	Ferralic
SS6KN6	5.6	5.7	0.05	1.2	0.23	5.11	1.98	12	1.68	5.72	2.27	1.93	Ferralic
SS7KN7	6.3	5.6	0.03	0.9	0.15	6.27	1.62	9	1.37	5.83	2.45	1.43	Ferralic
SS8KN8	6.2	5.5	0.04	1.0	0.23	4.22	1.64	9	1.93	5.3	2.12	1.43	Ferralic
SS9KN9	6.6	6.1	0.05	1.0	0.23	4.22	1.64	19	1.93	6.92	2.31	2.1	Ferralic
SS10KN10	6.1	5.4	0.04	1.0	0.15	6.33	1.64	8	1.37	5.02	2.42	1.43	Ferralic
SS11KN11	6.5	5.9	0.04	1.3	0.19	6.99	2.26	13	1.68	6.15	2.2	1.85	Ferralic
SS12KN12	6.1	5.7	0.05	1.4	0.26	5.41	2.45	12	1.7	5.7	2.4	1.9	Ferralic
SS13KN13	6.3	5.7	0.04	1.0	0.19	5.44	1.76	9	1.48	6.09	2.46	1.6	Ferralic
SS14KN14	6.0	5.6	0.04	0.9	0.19	4.91	1.59	10	1.17	5.96	2.54	1.76	Ferralic
SS15KN15	6.2	5.8	0.04	1.1	0.15	7.40	1.91	9	1.32	6.21	2.46	1.85	Ferralic
SS16KN16	6.4	5.9	0.04	0.7	0.23	2.98	1.16	7	1.22	4.97	2.15	1.85	Ferralic
SS17KN17	6.2	5.6	0.04	0.7	0.15	4.87	1.26	13	1.22	4.59	2.13	1.93	Ferralic
SS18KN18	6.5	6.1	0.04	1.2	0.23	5.33	2.07	11	1.63	6.56	2.48	2.27	Ferralic
SS19KN19	6.3	5.7	0.05	1.1	0.26	4.34	1.97	19	1.58	6.45	2.38	2.1	Ferralic
SS20KN20	6.3	5.5	0.04	0.8	0.19	4.00	1.29	12	1.42	5.66	2.36	2.18	Ferralic
SS21KN21	5.7	4.9	0.04	0.7	0.23	2.98	1.16	12	1.02	4.67	2.2	1.43	Ferralic
SS22KN22	6.4	5.9	0.04	0.9	0.23	3.78	1.47	11	1.37	6.08	2.39	1.85	Ferralic
SS23KN23	6.6	5.6	0.04	1.2	0.15	8.00	2.07	14	1.53	7.24	2.39	2.1	Ferralic
SS24KN24	6.1	5.6	0.04	0.8	0.23	3.69	1.43	20	1.32	5.68	2.55	2.1	Ferralic
SS25KN25	6.2	6.1	0.03	0.6	0.23	2.71	1.05	12	1.07	5.72	2.47	1.68	Ferralic
SS26KN26	5.9	5.4	0.03	1.0	0.23	4.27	1.66	15	1.07	5.35	2.71	1.76	Ferralic of
SS27KN27	6.1	5.4	0.03	0.7	0.23	3.29	1.28	13	1.22	5.65	2.35	1.93	Ferralic of
SS28KN28	5.7	5.0	0.04	1.0	0.15	6.53	1.69	15	1.48	5.18	2.41	2.18	Ferralic
SS29KN29	6.2	5.7	0.03	0.8	0.15	5.20	1.34	15	1.32	5.58	2.24	1.93	Ferralic
SS30KN30	6.1	5.4	0.03	0.8	0.23	3.51	1.36	8	0.86	5.09	2.38	1.51	Ferralic of

е

cambisols/calcic xerosols cambisols/calcic xerosols

KAPKUN WET SEASON – Soil samples

Senders ref.	H ₂ O	CaCl ₂	Ec	С	Ν	C:N	SOM	Р	Κ	Ca	Mg	Na	Soil type
	pН	pH	µs/cm	%	%	Ratio	%	ppm	E Ba	ases C	mol ()/Kg	
KN01/SS	6.6	6.5	50	1.0	0.71	1.33	1.64	5	1.5	11.3	5.3	0.04	Ferralic cambisols/calcic xerosols
KN02/SS	6.5	6.3	53	1.4	1.04	1.33	2.38	22	1.7	7.1	5.7	0.01	Ferralic cambisols/calcic xerosols
KN03/SS	6.5	6.4	55	1.4	1.07	1.33	2.47	20	1.6	6.4	5.4	0.05	Ferralic cambisols/calcic xerosols
KN04/SS	6.4	6.3	30	0.9	0.69	1.33	1.59	10	1.5	5.5	5.6	0.07	Ferralic cambisols/calcic xerosols
KN05/SS	6.5	6.4	23	0.8	0.63	1.33	1.45	12	1.5	5.1	4.9	0.00	Ferralic cambisols/calcic xerosols
KN06/SS	6.6	6.4	41	1.3	0.94	1.33	2.16	20	1.5	6.5	5.1	0.04	Ferralic cambisols/calcic xerosols
KN07/SS	6.5	6.4	44	1.2	0.88	1.33	2.02	18	1.7	6.2	5.4	0.04	Ferralic cambisols/calcic xerosols
KN08/SS	6.5	6.3	43	1.1	0.80	1.33	1.84	15	1.6	6.7	5.6	0.07	Ferralic cambisols/calcic xerosols
KN09/SS	6.6	6.3	42	1.0	0.76	1.33	1.74	17	1.6	5.6	5.3	0.05	Ferralic cambisols/calcic xerosols
KN10/SS	6.5	6.4	32	0.9	0.64	1.33	1.47	7	1.2	7.2	6.1	0.06	Ferralic cambisols/calcic xerosols
KN11/SS	6.6	6.3	43	0.9	0.71	1.33	1.62	15	1.4	6.9	5.6	0.05	Ferralic cambisols/calcic xerosols
KN12/SS	6.6	6.2	33	1.0	0.15	6.80	1.76	10	2.5	11.7	6.4	0.31	Ferralic cambisols/calcic xerosols
KN13/SS	6.6	6.4	42	1.2	0.19	6.16	2.02	9	2	7.2	5.2	0.01	Ferralic cambisols/calcic xerosols
KN14/SS	6.7	6.5	51	0.8	0.11	7.36	1.40	8	1.2	6.1	5.6	0.06	Ferralic cambisols/calcic xerosols
KN15/SS	6.5	6.4	47	1.0	0.19	5.00	1.64	13	1.9	5.7	5.8	0.07	Ferralic cambisols/calcic xerosols
KN01/SS	7.3	6.3	43	0.9	0.19	4.95	1.62	18	2.3	7.9	4.9	0.17	Ferralic cambisols/calcic xerosols
KN02/SS	6.3	6.0	27	1.2	0.19	6.32	2.07	18	1.3	6.9	6.3	0.11	Ferralic cambisols/calcic xerosols
KN03/SS	7.4	6.0	29	1.3	0.11	12.09	2.29	14	1.5	6.5	3.2	0.16	Ferralic cambisols/calcic xerosols
KN04/SS	6.7	6.0	29	1.2	0.19	6.32	2.07	17	1.5	5.7	5.3	0.14	Ferralic cambisols/calcic xerosols
KN05/SS	6.7	5.8	33	1.3	0.19	6.58	2.16	30	1.8	7.3	6.3	0.27	Ferralic cambisols/calcic xerosols
KN06/SS	6.8	6.8	36	0.9	0.19	4.68	1.53	28	1.4	6.7	5.8	0.23	Ferralic cambisols/calcic xerosols
KN07/SS	6.8	6.5	44	1.1	0.19	5.53	1.81	12	1.7	6.4	5.9	0.39	Ferralic cambisols/calcic xerosols
KN08/SS	6.8	6.8	43	1.1	0.11	10.00	1.90	16	1.8	6.5	5.5	0.13	Ferralic cambisols/calcic xerosols
KN09/SS	7.05	6.0	22	1.0	0.19	5.11	1.67	9	1.6	7.7	6.1	0.19	Ferralic cambisols/calcic xerosols
KN10/SS	6.8	6.5	11	1.0	0.11	8.82	1.67	13	1.3	5.9	4.7	0.04	Ferralic cambisols/calcic xerosols
KN/11/SS	6.5	6.1	28	0.7	0.19	3.63	1.19	28	3.3	8.4	6.1	0.09	Ferralic cambisols/calcic xerosols
KN/12/SS	6.6	6.2	82	1.8	0.19	9.53	3.12	17	1.8	6.3	5.7	0.08	Ferralic cambisols/calcic xerosols
KN/13/SS	7.05	6.3	25	0.8	0.19	4.11	1.34	28	2.2	7.7	5.7	0.26	Ferralic cambisols/calcic xerosols
KN/14/SS	6.9	6.0	20	1.0	0.19	5.47	1.79	15	1.1	5.9	5.9	0.06	Ferralic cambisols/calcic xerosols
KN/15/SS	7.05	6.1	40	0.8	0.11	7.64	1.45	13	1.7	8.4	5.9	0.20	Ferralic cambisols/calcic xerosols

MAOI DRY SEASON – Soil samples

Senders ref.	H ₂ O	CaCl ₂	Ec	С	N	C:N	SOM	P	K	Ca	Mg	Na	Soil type
	pH	pH	µs/cm	%	%	Ratio	%	ppm	E Ba	ses C	mol (†)/Kg	
MI/01/SS	6.5	6.5	34	0.6	0.11	5.27	1.00	10	1.2	4.2	3.9	1.7	Ando-chromic cambisols/xerosols
MI/02/SS	6.7	6.3	26	0.8	0.19	4.42	1.45	13	1.3	5.2	4.9	1.7	Ando-chromic cambisols/xerosols
MI/03/SS	6.6	6.5	24	0.9	0.19	4.58	1.50	12	1.1	5.5	4.5	1.7	Ando-chromic cambisols/xerosols
MI/04/SS	6.6	6.4	40	0.9	0.11	8.09	1.53	12	1.3	5.4	4.8	1.9	Ando-chromic cambisols/xerosols
MI/05/SS	7.05	6.6	67	0.9	0.26	3.35	1.50	9	0.4	10.9	3.9	1.7	Ando-chromic cambisols/xerosols
MI/06/SS	6.3	6.2	64	1.0	0.19	5.05	1.66	15	1	5.2	4.1	1.9	Ando-chromic cambisols/xerosols
MI/07/SS	6.7	6.4	20	0.6	0.26	2.27	1.02	10	1.1	4.8	3.9	1.8	Ando-chromic cambisols/xerosols
MI/08/SS	6.7	6.4	23	0.4	0.11	3.45	0.66	12	0.9	4.9	3.9	1.8	Ando-chromic cambisols/xerosols
MI/09/SS	6.9	6.6	20	0.3	0.26	1.12	0.50	23	0.9	4.0	4.4	1.7	Ando-chromic cambisols/xerosols
MI/10/SS	6.7	6.1	12	0.3	0.11	2.82	0.53	9	0.7	3.1	4.2	1.8	Ando-chromic cambisols/xerosols
MI/11/SS	6.9	6.5	19	0.4	0.19	2.16	0.71	14	0.7	3.1	0.4	1.7	Ando-chromic cambisols/xerosols
MI/12/SS	6.7	6.5	18	0.3	0.19	1.74	0.57	14	0.9	3.0	4.4	1.8	Ando-chromic cambisols/xerosols
MI/13/SS	7.05	6.5	20	0.9	0.19	4.53	1.48	8	1	3.9	4.3	1.6	Ando-chromic cambisols/xerosols
MI/14/SS	7.05	6.3	26	0.4	0.19	2.21	0.72	22	0.8	3.4	4.2	1.7	Ando-chromic cambisols/xerosols
MI/15/SS	6.8	6.4	39	1.2	0.26	4.65	2.09	10	1.1	5.1	3.9	2.0	Ando-chromic cambisols/xerosols
MI/16/SS	6.4	6.3	32	0.6	0.19	3.05	1.00	12	0.8	4.4	3.4	1.9	Ando-chromic cambisols/xerosols
MI/17/SS	6.7	6.3	23	0.7	0.11	5.91	1.12	11	1.1	38.4	5.8	1.6	Ando-chromic cambisols/xerosols
MI/18/SS	6.6	6.4	37	0.6	0.11	5.45	1.03	35	1.2	3.7	3.6	1.8	Ando-chromic cambisols/xerosols
MI/19/SS	6.6	6.4	21	0.4	0.19	2.05	0.67	40	1.2	3.3	3.5	1.7	Ando-chromic cambisols/xerosols
MI/20/SS	6.6	6.2	18	0.4	0.11	3.82	0.72	45	1.2	3.7	3.7	1.8	Ando-chromic cambisols/xerosols
MI/21/SS	6.7	6.5	28	0.6	0.19	3.05	1.00	9	1.2	5.4	3.9	1.8	Ando-chromic cambisols/xerosols
MI/22/SS	6.7	6.5	54	0.6	0.11	5.27	1.00	32	1.6	4.1	2.6	1.7	Ando-chromic cambisols/xerosols
MI/23/SS	6.8	6.4	24	0.5	0.11	4.55	0.86	13	1	4.0	3.6	1.8	Ando-chromic cambisols/xerosols
MI/24/SS	7.2	6.6	14	0.3	0.11	2.55	0.48	8	0.8	3.3	4.5	1.7	Ando-chromic cambisols/xerosols
MI/25/SS	6.6	6.3	29	0.4	0.11	3.18	0.60	20	1.2	3.0	2.7	1.7	Ando-chromic cambisols/xerosols
MI/26/SS	7.05	6.5	21	0.6	0.04	14.50	1.00	3	1.2	4.6	4.3	1.9	Ando-chromic cambisols/xerosols
MI/27/SS	6.7	6.5	38	0.4	0.11	3.82	0.72	16	1.2	5.9	5	1.8	Ando-chromic cambisols/xerosols
MI/28/SS	7.2	6.7	54	1.2	0.19	6.32	2.07	13	1.5	7.8	3.6	1.8	Ando-chromic cambisols/xerosols
MI/29/SS	6.7	6.6	27	0.5	0.11	4.73	0.90	12	1.3	4.5	4.9	1.7	Ando-chromic cambisols/xerosols
MAOI WET SEASON – Soil samples

Senders ref.	H ₂ O	CaCl ₂	Ec	С	N	C:N	SOM	Р	K	Ca	Mg	Na	Soil type
	pН	pН	mS/cm	%	%	Ratio	%	ppm	ΕB	ases C	mol (*)/Kg	
M1/01/SS	5.9	5.8	0.16	1.02	0.19	5.37	1.76	11	1.1	4.15	1.57	5.3	Ando-chromic cambisols/xerosols
M1/02/SS	6.1	5.8	0.05	0.81	0.19	4.26	1.40	3	1.2	3.64	1.48	0.66	Ando-chromic cambisols/xerosols
M1/03/SS	6.5	6.2	0.06	0.81	0.19	4.26	1.40	5	1.2	3.92	1.75	0.7	Ando-chromic cambisols/xerosols
M1/04/SS	6.1	5.8	0.03	0.69	0.19	3.63	1.19	6	1.25	4.39	1.97	0.66	Ando-chromic cambisols/xerosols
M1/05/SS	6.3	6.0	0.07	0.84	0.19	4.42	1.45	3	1.45	5.3	1.78	0.26	Ando-chromic cambisols/xerosols
M1/06/SS	6.1	5.1	0.07	1.23	0.26	4.73	2.12	5	1.2	4.83	1.75	0.22	Ando-chromic cambisols/xerosols
M1/07/SS	6.1	5.9	0.05	0.95	0.19	5.00	1.64	25	1.25	3.88	1.66	0.31	Ando-chromic cambisols/xerosols
M1/08/SS	6.1	6.0	0.05	0.91	0.11	8.27	1.57	12	0.7	4.33	1.45	0.09	Ando-chromic cambisols/xerosols
M1/09/SS	6.3	6.0	0.07	0.79	0.26	3.04	1.36	8	1.8	3.83	1.41	0.44	Ando-chromic cambisols/xerosols
M1/10/SS	6.4	5.9	0.05	0.57	0.11	5.18	0.98	9	0.85	3.26	1.32	0.13	Ando-chromic cambisols/xerosols
M1/11/SS	6.5	6.0	0.05	1.09	0.19	5.74	1.88	10	1.2	4.59	1.64	0.44	Ando-chromic cambisols/xerosols
M1/12/SS	5.8	5.6	0.04	0.69	0.19	3.63	1.19	8	1.25	2.87	1.5	0.35	Ando-chromic cambisols/xerosols
M1/13/SS	5.9	5.9	0.04	0.83	0.19	4.37	1.43	10	0.7	4.86	1.78	0.31	Ando-chromic cambisols/xerosols
M1/14/SS	6.2	6.1	0.03	0.7	0.19	3.68	1.21	16	1.25	3.92	1.72	0.44	Ando-chromic cambisols/xerosols
M1/15/SS	6.2	6.1	0.07	0.8	0.19	4.21	1.38	8	1.15	4.2	1.85	0.4	Ando-chromic cambisols/xerosols
M1/16/SS	6.5	6.3	0.02	0.74	0.19	3.89	1.28	22	1.3	4.24	1.76	0.53	Ando-chromic cambisols/xerosols
M1/18/SS	6.0	5.7	0.05	1.04	0.19	5.47	1.79	11	1	3.7	1.46	0.31	Ando-chromic cambisols/xerosols
M1/19/SS	5.9	5.8	0.13	0.76	0.26	2.92	1.31	7	1.05	3.94	1.7	0.53	Ando-chromic cambisols/xerosols
M1/20/SS	6.3	5.6	0.06	0.95	0.26	3.65	1.64	11	1.1	5.12	1.51	0.4	Ando-chromic cambisols/xerosols
SS/M1/1	6.3	5.9	0.03	0.69	0.11	6.27	1.19	5	0.9	3.68	1.28	0.31	Ando-chromic cambisols/xerosols
SS/M1/2	6.2	5.6	0.03	0.38	0.19	2.00	0.66	8	1.1	3.18	1.84	0.48	Ando-chromic cambisols/xerosols
SS/M1/3	6.6	6.5	0.04	0.88	0.19	4.63	1.52	4	1.4	4.29	1.82	0.57	Ando-chromic cambisols/xerosols
SS/M1/4	6.3	6.1	0.03	0.32	0.26	1.23	0.55	12	1.1	3.13	1.64	0.48	Ando-chromic cambisols/xerosols
SS/M1/5	6.5	6.3	0.07	0.87	0.19	4.58	1.50	24	1.25	9.75	1.25	0.53	Ando-chromic cambisols/xerosols
SS/M1/6	6.4	6.3	0.03	0.69	0.19	3.63	1.19	10	2.25	3.2	1.71	0.97	Ando-chromic cambisols/xerosols
SS/M1/7	6.4	6.1	0.03	0.79	0.11	7.18	1.36	19	1.25	3.69	1.42	0.53	Ando-chromic cambisols/xerosols
SS/M1/8	6.3	6.1	0.02	0.47	0.11	4.27	0.81	15	1.3	5.44	1.93	0.62	Ando-chromic cambisols/xerosols
SS/M1/9	6.9	6.5	0.04	0.54	0.26	2.08	0.93	15	1.65	4.48	1.53	0.7	Ando-chromic cambisols/xerosols
SS/M1/10	6.9	6.5	0.03	0.35	0.19	1.84	0.60	12	0.9	3.56	2.03	0.48	Ando-chromic cambisols/xerosols

165

SS/M1/11	6.5	6.2	0.03	1.06	0.19	5.58	1.83	9	0.95	4.42	1.67	0.48	Ando-chromic cambisols/xerosols
SS/M1/12	6.4	5.9	0.05	0.03	0.19	0.17	0.06	14	1.7	5.56	1.5	0.84	Ando-chromic cambisols/xerosols
SS/M1/13	5.9	5.8	0.02	0.5	0.26	1.92	0.86	16	1.15	2.63	1.24	0.53	Ando-chromic cambisols/xerosols
SS/M1/14	5.9	5.6	0.02	0.79	0.19	4.16	1.36	10	0.85	3.29	1.49	0.53	Ando-chromic cambisols/xerosols
SS/M1/15	6.9	6.4	0.03	0.34	0.19	1.79	0.59	36	2.8	5.43	3.07	1.19	Ando-chromic cambisols/xerosols

SO	LIT: DRY S	SEASO	N - Tra	ce element	ts – EDT	A extracti	ions dry a	nd wet se	asons
Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn	
No.	Ref.	pH	pН	mS/cm	ppm	ppm	ppm	ppm	
170	ST/SSM1	7.9	7.3	0.09	0.1	14	20	0.5	
172	ST/SSM2	8.1	7.6	0.07	0.6	30	133	0.2	
174	ST/SSM3	8.2	7.1	0.07	0.1	32	34	0.2	
176	ST/SSM4	7.9	7.2	0.08	0.1	59	38	0.4	
178	ST/SSM5	8.0	7.7	0.05	0.1	59	53	1.7	
180	ST/SSM6	8.0	7.7	0.05	0.4	60	85	0.8	
182	ST/SSM7	8.1	7.1	0.06	0.6	53	79	1.0	
184	ST/SSM8	7.9	7.3	0.05	1.1	100	293	1.3	
186	ST/SSM9	8.1	7.2	0.06	1.2	102	304	1.0	
188	ST/SSM10	8.1	7.2	0.07	1.5	117	301	0.7	
190	ST/SSM11	7.5	7.2	0.08	2.3	122	326	2.4	
192	ST/SSM12	8.1	7.2	0.05	2.3	159	340	1.2	
194	ST/SSM13	8.4	7.3	0.05	1.8	124	246	0.7	
196	ST/SSM14	8.0	7.1	0.04	1.7	205	428	1.0	
198	ST/SSM15	8.1	7.3	0.06	2.1	199	312	1.5	
200	ST/SSL1	7.6	7.1	0.06	2.4	207	364	2.3	
202	ST/SSL2	7.7	7.05	0.09	2.1	208	381	2.4	
204	ST/SSL3	7.1	7.1	0.05	1.5	209	384	0.9	
206	ST/SSL4	8.0	7.1	0.08	1.6	247	445	0.6	
208	ST/SSL5	7.7	7.3	0.06	2.1	249	501	0.7	
210	ST/SSL6	7.7	7.05	0.06	2.9	256	461	0.6	
212	ST/SSL7	7.8	7.05	0.06	2.1	203	206	0.9	
214	ST/SSL8	7.2	7.1	0.05	2.6	232	228	0.8	
216	ST/SSL9	8.0	7.1	0.06	2.4	230	191	0.9	
218	ST/SSL10	7.8	7.1	0.06	3.0	244	226	0.2	
220	ST/SSL11	7.9	7.1	0.06	3.0	269	296	1.4	
222	ST/SSL12	7.9	7.05	0.05	3.1	263	207	0.6	
224	ST/SSL13	8.1	7.05	0.05	3.4	265	228	1.0	
226	ST/SSL14	8.2	7.1	0.05	3.4	266	232	1.0	
228	ST/SSL15	8.0	-7.5	0.06	3.5	311	238	1.1	
230	ST/SSL16	8.1	7.4	0.05	4.4	330	273	2.3	
232	ST/SSL17	8.2	7.1	0.06	4.2	346	449	2.6	

SOLIT: WET SEASON - EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	Ref.	pН	рН	□s/cm	ppm	ppm	ppm	ppm
343	ST01/SS	7.2	7.05	50	1.6	61	226	1.3
344	ST02/SS	7.1	6.6	100	1.9	44	107	1.8
345	ST03/SS	7.4	7.1	121	1.6	95	222	2.5
346	ST04/SS	7.1	6.7	68	1.3	98	247	2.5
347	ST05/SS	7.4	7.1	85	1.0	108	208	2.9
383	SS/ST/1	7.3	7.1	120	0.7	121	226	3.4
384	SS/ST/2	7.2	7.05	112	2.0	137	201	3.0
385	SS/ST/3	7.2	6.9	67	0.6	107	126	2.2
386	SS/ST/4	7.3	6.9	76	1.0	113	212	2.4
387	SS/ST/5	7.3	6.9	37	2.1	138	287	2.3
388	SS/ST/6	7.3	7.05	41	3.2	85	362	3.2
389	SS/ST/7	7.1	6.6	59	1.8	94	417	3.3
390	SS/ST/8	7.1	6.8	68	0.3	189	227	4.0
391	SS/ST/9	6.9	6.7	72	0.8	106	248	2.0
392	SS/ST/10	7.1	6.9	90	2.8	94	280	2.8
233	ST/01/SS	7.6	6.7	61	0.3	146	231	0.9
234	ST/02/SS	7.4	7.1	6	3.0	217	389	1.6
235	ST/03/SS	7.3	6.5	34	1.2	134	323	2.2
236	ST/04/SS	8.2	7.3	154	1.8	167	407	2.2
237	ST/05/SS	7.1	6.8	70	1.6	140	384	2.7
238	ST/05/SS	7.1	6.8	50	2.6	137	309	3.0
239	ST/06/SS	7.05	7.05	75	1.5	103	148	3.1
240	ST/07/SS	7.3	6.7	16	0.4	132	357	4.2
241	ST/08/SS	7.1	7.05	33	0.8	149	470	1.0
242	ST/09/SS	7.4	6.9	50	0.5	113	209	1.1
243	ST/10/SS	7.5	6.9	40	0.1	103	94	1.0
244	ST/11SS	7.8	6.9	47	0.2	118	204	1.7
245	ST/12/SS	7.6	7.2	86	2.1	110	217	2.5
246	ST/13/SS	8.0	6.8	53	1.2	173	502	3.2
247	ST/14/SS	7.3	6.9	68	1.0	101	83	3.4
248	ST/15/SS	7.2	6.9	62	0.5	123	346	0.8

KIMOROK: DRY SEASON - EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	Ref.	pН	pН	mS/cm	ppm	ppm	ppm	ppm
117	KK1SS	6.8	6.2	0.07	0.1	309	255	2.4
118	KK2SS	6.5	6.1	0.05	0.1	213	240	1.3
119	KK3SS	6.5	5.9	0.05	0.2	446	264	2.9
120	KK4SS	5.9	5.7	0.04	0.6	223	230	1.9
121	KK5SS	6.5	5.8	0.05	0.1	357	258	2.1
122	KK6SS	6.0	5.7	0.06	0.2	443	387	2.9
123	KK7SS	5.9	5.8	0.04	0.2	402	355	2.9
124	KK8SS	5.6	5.6	0.03	0.1	307	219	1.7
125	KK9SS	6.4	6.1	0.03	0.1	326	236	3.3
126	KK10SS	6.0	5.9	0.06	0.3	293	228	1.9
127	KK11SS	5.8	5.7	0.04	0.3	306	278	2.4
128	KK12SS	6.2	5.9	0.03	0.4	313	260	3.1
129	KK13SS	6.0	5.8	0.04	0.3	329	426	2.9
130	KK14SS	6.1	5.7	0.03	0.4	371	400	4.0
131	KK15SS	6.2	5.7	0.03	0.5	383	484	4.3
132	KK16SS	5.6	5.5	0.03	0.6	117	161	5.2
133	KK17SS	6.2	5.8	0.04	0.2	178	251	1.8
134	KK18SS	6.8	6.0	0.05	0.3	138	246	1.8
135	KK19SS	6.5	5.8	0.03	0.3	172	293	2.6
136	KK20SS	6.4	5.9	0.03	0.8	145	324	4.7
137	KK21SS	6.3	5.9	0.04	0.6	138	326	4.1
138	KK22SS	6.4	5.9	0.05	0.7	152	340	5.4
139	KK23SS	6.4	5.7	0.03	0.7	172	309	4.7
140	KK24SS	6.0	5.7	0.05	0.8	146	239	5.4
141	KK25SS	6.4	6.0	0.03	0.1	136	204	6.1
142	KK26SS	6.2	5.7	0.03	0.1	129	208	1.8
143	KK27SS	6.6	5.5	0.03	0.1	232	489	2.7
144	KK28SS	6.3	5.9	0.05	0.4	205	317	2.1
145	KK29SS	6.3	5.7	0.04	0.3	157	392	2.7
146	KK30SS	5.8	5.8	0.03	0.5	158	280	3.8

KIMOROK: WET SEASON – EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	Ref.	pН	pН	□s/cm	ppm	ppm	ppm	ppm
373	KK01/SS	6.5	6.3	41	0.8	149	213	14.4
374	KK02/SS	6.4	6.2	55	0.5	213	321	14.5
375	KK03/SS	6.6	6.4	85	0.6	187	374	14.3
376	KK04/SS	6.8	6.3	35	0.6	248	372	14.9
377	KK05/SS	6.3	6.1	42	0.5	186	306	16.0
378	KK06/SS	6.5	6.3	66	0.6	189	367	25.8
379	KK07/SS	6.5	6.2	65	0.7	221	364	16.9
380	KK08/SS	6.3	6.1	46	0.7	210	306	17.7
381	KK09/SS	6.5	6.4	55	0.7	239	304	3.3
382	KK10SS	6.1	6.0	61	0.9	198	452	2.3
383	KK11/SS	6.3	6.1	60	0.8	151	183	2.5
384	KK12/SS	6.4	6.1	58	0.9	326	398	3.1
385	KK13/SS	6.7	6.3	40	1.1	155	181	4.0
386	KK14/SS	6.6	6.3	39	1.2	249	352	4.7
387	KK15/SS	6.7	6.2	38	1.2	185	296	5.2
388	KK16/SS	6.8	6.3	42	1.1	227	376	4.0
389	KK17/SS	6.9	6.3	50	1.2	306	432	5.4
390	KK18/SS	6.6	6.3	40	1.4	164	310	14.5
391	KK19/SS	6.9	6.5	54	1.4	208	361	18.3
392	KK20/SS	6.5	6.4	57	1.5	162	259	16.3
393	KK21/SS	6.9	6.4	56	0.8	254	247	2.2
394	KK22/SS	6.9	6.3	41	1.2	288	482	14.4
395	KK23/SS	6.7	6.2	35	1.4	237	396	14.1
396	KK24/SS	6.8	6.2	38	1.4	217	356	13.9
397	KK25/SS	6.9	6.3	55	2.3	273	374	15.1
398	KK26/SS	6.8	6.5	47	2.0	279	404	14.8
399	KK27/SS	6.5	6.3	83	1.7	253	318	14.7
400	KK28/SS	6.5	6.3	39	1.6	295	472	14.4
401	KK29/SS	6.6	6.3	48	1.8	219	246	14.0
404	KK32/SS	6.8	6.2	41	1.6	237	367	14.4
296	KK/04/SS	7.6	6.6	26	0.3	173	290	14.2
298	KK/06/SS	6.6	6.1	11	0.5	191	444	14.3

KAPKUN: DRY SEASON – EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	Ref.	pН	pH	mS/cm	ppm	ppm	ppm	ppm
276	KN/SS1	6.3	5.5	0.03	1.1	147	264	3.3
278	KN/SS2	5.9	5.4	0.05	1.1	339	483	3.2
280	KN/SS3	6.2	5.9	0.03	0.3	300	419	2.6
282	KN/SS4	5.9	5.1	0.04	0.6	377	495	8.5
284	KN/SS5	6.3	5.8	0.04	0.8	300	361	6.9
286	KN/SS6	5.6	5.7	0.05	0.4	346	411	6.4
288	KN/SS7	6.3	5.6	0.03	0.4	378	446	6.7
290	KN/SS8	6.2	5.5	0.04	0.4	288	247	7.6
292	KN/SS9	6.6	6.1	0.05	0.9	288	427	10.9
294	KN/SS10	6.1	5.4	0.04	0.4	185	301	8.9
296	KN/SS11	6.5	5.9	0.04	0.4	279	455	9.1
298	KN/SS12	6.1	5.7	0.05	0.3	218	215	3.6
300	KN/SS13	6.3	5.7	0.04	0.6	291	413	2.5
302	KN/SS14	6.0	5.6	0.04	0.4	223	303	9.2
304	KN/SS15	6.2	5.8	0.04	0.7	291	364	11.7
306	KN/SS16	6.4	5.9	0.04	0.6	263	295	3.6
308	KN/SS17	6.2	5.6	0.04	0.5	257	274	2.6
310	KN/SS18	6.5	6.1	0.04	0.5	384	542	4.6
312	KN/SS19	6.3	5.7	0.05	1.3	423	527	9.8
314	KN/SS20	6.3	5.5	0.04	1.1	320	307	4.5
316	KN/SS21	5.7	4.9	0.04	0.9	379	330	4.9
318	KN/SS22	6.4	5.9	0.04	1.3	244	400	6.3
320	KN/SS23	6.6	5.6	0.04	1.1	368	581	7.8
322	KN/SS24	6.1	5.6	0.04	0.9	287	442	8.3
324	KN/SS25	6.2	6.1	0.03	1.0	197	327	7.6
326	KN/SS26	5.9	5.4	0.03	2.0	271	466	5.1
328	KN/SS27	6.1	5 . 4	0.03	1.3	206	386	5.7
330	KN/SS28	5.7	5.0	0.04	1.8	270	480	8.5
332	KN/SS29	6.2	5.7	0.03	1.8	177	360	6.9
334	KN/SS30	6.1	5.4	0.03	1.6	179	311	3.8

KAPKUN: WET SEASON – EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	Ref.	pН	pН	□s/cm	ppm	ppm	ppm	ppm
348	KN01/SS	6.6	6.5	50	0.2	189	292	2.6
349	KN02/SS	6.5	6.3	53	0.2	268	444	5.0
350	KN03/SS	6.5	6.4	55	0.3	197	295	3.1
351	KN04/SS	6.4	6.3	30	0.0	170	203	3.1
352	KN05/SS	6.5	6.4	23	0.4	170	180	3.0
353	KN06/SS	6.6	6.4	41	0.4	258	346	4.7
354	KN07/SS	6.5	6.4	44	0.2	217	340	5.4
355	KN08/SS	6.5	6.3	43	0.4	295	426	4.8
356	KN09/SS	6.6	6.3	42	0.2	194	284	4.5
357	KN10/SS	6.5	6.4	32	0.4	188	266	5.3
358	KN11/SS	6.6	6.3	43	0.2	239	305	5.8
359	KN12/SS	6.6	6.2	33	0.2	265	330	5.8
360	KN13/SS	6.6	6.4	42	0.3	235	330	7.7
361	KN14/SS	6.7	6.5	51	0.2	156	265	7.1
362	KN15/SS	6.5	6.4	47	0.4	166	389	9.4
249	KN01/SS	7.3	6.3	43	0.4	116	239	9.3
250	KN02/SS	6.3	6.0	27	0.5	179	275	9.7
251	KN03/SS	7.4	6.0	29	0.4	331	487	9.3
252	KN04/SS	6.7	6.0	29	0.3	232	266	1.0
253	KN05/SS	6.7	5.8	33	0.5	186	275	10.9
254	KN06/SS	6.8	6.8	36	0.4	146	139	8.8
255	KN07/SS	6.8	6.5	44	0.3	265	314	1.8
256	KN08/SS	6.8	6.8	43	0.1	290	403	2.0
257	KN09/SS	7.05	6.0	22	0.3	162	173	8.1
258	KN10/SS	6.8	6.5	11	0.3	186	175	2.3
259	KN/11/SS	6.5	6.1	28	0.7	244	374	6.1
260	KN/12/SS	6.6	6.2	82	0.6	222	239	4.9
261	KN/13/SS	7.05	6.3	25	0.3	227	264	5.5
262	KN/14/SS	6.9	6.0	20	0.5	240	325	7.7
263	KN/15/SS	7.05	6.1	40	0.6	258	323	7.7
264	KN/16/SS	7.3	6.3	24	0.2	287	380	2.7

MAOI - DRY SEASON -- EDTA Extractables

Lab	Senders	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.	ref.	pН	pН	□s/cm	ppm	ppm	ppm	ppm
265	MI/01/SS	6.5	6.5	34	0.2	91	201	0.9
266	MI/02/SS	6.7	6.3	26	0.2	126	318	2.0
267	MI/03/SS	6.6	6.5	24	0.2	154	310	3.1
268	MI/04/SS	6.6	6.4	40	0.4	167	389	3.4
269	MI/05/SS	7.05	6.6	67	0.7	159	684	5.4
270	MI/06/SS	6.3	6.2	64	0.1	198	408	2.7
271	MI/07/SS	6.7	6.4	20	0.4	129	276	2.3
272	MI/08/SS	6.7	6.4	23	0.3	131	245	3.0
273	MI/09/SS	6.9	6.6	20	0.2	143	246	3.5
274	MI/10/SS	6.7	6.1	12	0.2	140	151	1.6
275	MI/11/SS	6.9	6.5	19	0.3	158	242	1.9
276	MI/12/SS	6.7	6.5	18	0.0	206	314	3.8
277	MI/13/SS	7.05	6.5	20	0.3	189	318	3.1
278	MI/14/SS	7.05	6.3	26	0.2	182	271	2.9
279	MI/15/SS	6.8	6.4	39	0.3	264	400	3.5
280	MI/16/SS	6.4	6.3	32	0.4	260	420	4.0
420	MI/17/SS	6.7	6.3	23	0.5	205	204	2.8
421	MI/18/SS	6.6	6.4	37	0.7	233	278	4.1
422	MI/19/SS	6.6	6.4	21	0.5	221	272	4.2
423	MI/20/SS	6.6	6.2	18	0.4	213	201	2.6
424	MI/21/SS	6.7	6.5	28	0.7	272	330	3.7
425	MI/22/SS	6.7	6.5	54	0.7	221	215	5.8
426	MI/23/SS	6.8	6.4	24	0.4	212	130	3.0
427	MI/24/SS	7.2	6.6	14	0.6	229	186	4.2
428	MI/25/SS	6.6	6.3	29	0.4	251	193	2.8
429	MI/26/SS	7.05	6.5	21	0.7	276	275	2.3
430	MI/27/SS	6.7	6.5	38	0.7	259	244	2.9
431	MI/28/SS	7.2	6.7	54	1.2	359	406	4.8
432	MI/29/SS	6.7	6.6	27	0.9	260	212	2.8
433	MI/30/SS	6.8	6.4	20	0.8	295	295	2.5

MAOI WET: EDTA Extractables

	Sender's							
Lab	ref.	H ₂ O	CaCl ₂	EC	Cu	Fe	Mn	Zn
No.		pН	pН	mS/cm	ppm	ppm	ppm	ppm
305	M1/SS/01	5.9	5.8	0.16	0.5	404	569	7.8
306	M1/SS/02	6.1	5.8	0.05	0.2	320	478	7.6
307	M1/SS/03	6.5	6.2	0.06	0.2	339	580	4.4
308	M1/SS/04	6.1	5.8	0.03	0.5	291	434	3.6
309	M1/SS/05	6.3	6.0	0.07	0.1	388	646	3.5
310	M1/SS/06	6.1	5.1	0.07	0.7	375	545	7.4
311	M1/SS/07	6.1	5.9	0.05	0.7	397	569	9.6
312	M1/SS/08	6.1	6.0	0.05	0.4	431	548	8.0
313	M1/SS/09	6.3	6.0	0.07	0.3	339	444	7.1
314	M1/SS/10	6.4	5.9	0.05	0.3	311	290	7.6
315	M1/SS/11	6.5	6.0	0.05	0.5	476	635	7.4
316	M1/SS/12	5.8	5.6	0.04	0.6	428	499	8.9
317	M1/SS/13	5.9	5.9	0.04	0.7	314	369	6.3
318	M1/SS/14	6.2	6.1	0.03	0.5	355	372	4.3
319	M1/SS/15	6.2	6.1	0.07	0.6	410	482	2.1
320	M1/SS/16	6.5	6.3	0.02	0.4	371	577	5.7
321	M1/SS/17	6.2	5.8	0.04	0.6	422	496	6.4
322	M1/SS/18	6.0	5.7	0.05	0.6	315	439	5.6
323	M1/SS/19	5.9	5.8	0.13	0.4	290	361	2.6
324	M1/SS/20	6.3	5.6	0.06	1.0	432	490	8.8
400	M1/SS/21	6.3	5.9	0.03	0.7	444	466	6.2
401	M1/SS/22	6.2	5.6	0.03	0.8	133	264	4.9
402	M1/SS/23	6.6	6.5	0.04	1.0	373	426	6.5
403	M1/SS/24	6.3	6.1	0.03	0.8	128	234	4.2
404	M1/SS/25	6.5	6.3	0.07	0.6	289	516	13.2
405	M1/SS/26	6.4	6.3	0.03	1.0	162	434	5.7
406	M1/SS/27	6.4	6.1	0.03	1.0	468	501	9.1
407	M1/SS/28	6.3	-6.1	0.02	1.1	183	378	4.2
408	M1/SS/29	6.9	6.5	0.04	1.2	433	432	9.2
409	M1/SS/30	6.9	6.5	0.03	0.7	389	413	2.9
410	M1/SS/31	6.5	6.2	0.03	1.7	304	370	13.4
411	M1/SS/32	6.4	5.9	0.05	1.4	612	746	9.5
412	M1/SS/33	5.9	5.8	0.02	1.1	439	382	5.9
413	M1/SS/34	5.9	5.6	0.02	0.8	413	508	8.5
414	M1/SS/35	6.9	6.4	0.03	1.0	348	434	6.7

APPENDIX II

MICRONUTRIENTS UPTAKE BY ACACIA SENEGAL VARIETIES ON NORMAL,

LOW (-) AND HIGH (+) DOSAGE TREATMENTS FOR HEIGHTS IN SITES

			Mean ±	S.E.	(P < 0.0	5)
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN	1 2 3 4 5 6 7 8 9	$140.71 \\ 149.20 \\ 163.58 \\ 159.20 \\ 168.21 \\ 172.88 \\ 160.25 \\ 154.71 \\ 148.92 \\$	9.86 11.60 9.86 10.01 9.86 9.86 9.86 9.86 9.86 9.86	$\begin{array}{c} 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \end{array}$	1 2 3 4 5 6 7 8 9
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK	1 2 3 4 5 6 7 8 9	184.63 162.11 172.03 177.91 161.52 160.67 154.05 179.15 178.40	$13.18 \\ 13.22 \\ 11.64 \\ 11.61 \\ 13.23 \\ 13.20 \\ 13.18 \\ 11.60 \\ 11.6$	$\begin{array}{c} 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \end{array}$	10 11 12 13 14 15 16 17 18
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	MAOI MAOI MAOI MAOI MAOI MAOI MAOI MAOI	1 2 3 4 5 6 7 8 9	152.61 155.74 167.61 168.41 130.16 161.54 145.48 186.32 158.46	$10.58 \\ 10.58 \\ 13.75 \\ 11.60 \\ 11.61 \\ 10.58 \\ 10.58 \\ 11.59 \\ 10.58 \\ 10.5$	$\begin{array}{c} 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.05 \end{array}$	19 20 21 22 23 24 25 26 27
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT	1 2 3 4 5 6 7 8 9	144.94 206.69 160.63 172.85 189.37 170.42 155.41 172.20 145.98	10.58 10.58 11.59 11.60 11.59 10.58 10.58 10.58 10.57	0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05	28 29 30 31 32 33 34 35

MICRONUTRIENTS UPTAKE BY ACACIA SENEGAL VARIETIES ON NORMAL, LOW (-) AND HIGH (+) DOSAGE TREATMENTS FOR DIAMETERS AT GROUND LEVEL IN SITES

			Mean ±	S.E.	(P < 0.05)	
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN KAPKUN	1 2 3 4 5 6 7 8 9	5.91 4.85 5.80 5.21 5.75 5.20 8.10 5.65 5.16	0.49 0.58 0.49 0.51 0.49 0.49 0.49 0.49 0.49 0.49	0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05	1 2 3 4 5 6 7 8 9
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK KIMOROK	1 2 3 4 5 6 7 8 9	6.64 5.33 5.89 5.78 5.86 6.48 5.95 5.05 5.65	0.66 0.58 0.58 0.66 0.66 0.66 0.58 0.58	0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05	10 11 12 13 14 15 16 17 18
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	MAOI MAOI MAOI MAOI MAOI MAOI MAOI MAOI	1 2 3 4 5 6 7 8 9	5.61 5.71 4.91 5.45 5.18 5.80 5.63 5.25 5.35	0.53 0.53 0.69 0.58 0.58 0.53 0.53 0.53 0.53	0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05	19 20 21 22 23 24 25 26 27
Normal Cu - Cu + Fe - Fe + Mn - Mn + Zn - Zn +	SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT SOLIT	1 2 3 4 5 6 7 8 9	6.14 6.74 5.95 6.60 5.86 4.88 6.31 6.01 5.60	0.53 0.53 0.58 0.58 0.58 0.58 0.58 0.53 0.53 0.53	0.05 0.05 0.05 0.05 0.05 0.05 0.05 0.05	28 29 30 31 32 33 34 35 36

APPENDIX III: CORRELATIONS BETWEEN SOIL AND GUM CHEMICAL PROPERTIES



Figure 1 Correlation between Soil N and Gum N in Solit



Figure 2 Correlations between Soil Cu and Gum Cu in Solit



Figure 3 Correlations between Soil Fe and Gum Fe in Solit







Figure 5 Correlations between Soil Zn and Gum Zn in Solit



Figure 6 Correlations between Soil N and Gum N in Kapkun



Figure 7 Correlations between Soil Cu and Gum Cu in Kapkun



Figure 8 Correlations between Soil Fe and Gum Fe in Kapkun



Figure 9 Correlations between Soil Mn and Gum Mn in Kapkun







Figure 11 Correlations between Soil N and Gum N in Kimorok







Figure 13 Correlations between Soil Fe and Gum Fe in Kimorok







Figure 15 Correlations between Soil Zn and Gum Zn in Kimorok



Figure 16 Correlations between Soil N and Gum N in Maoi



Figure 17 Correlations between Soil Cu and Gum Cu in Maoi



Figure 18 Correlations between Soil Fe and Gum Fe in Maoi



Figure 19 Correlations between Soil Mn and Gum Mn in Maoi



Figure 20 Correlations between Soil Zn and Gum Zn in Maoi



Figure 21 Correlations between Soil N and Gum N in all sites







Figure 23 Correlations between Soil Fe and Gum Fe in all sites









APPENDIX IV: Gum arabic Analysis

Physical properties of Gum arabic of the four Study sites

	Moisture	Volatile	Internal	Ash
	content %	matter %	energy %	content %
Kimorok	16.0	62.8	34.2	3.0
	15.2	64.0	33.2	2.8
	16.8	63.4	33.8	2.8
	24.6	64.0	33.2	2.8
	14.9	64.4	32.6	3.0
Solit	14.6	6 5.4	31.4	3.2
	16.8	65.0	32.4	2.6
	16.5	64.6	32.4	3.0
	16.0	63.0	34.0	3.0
	15.5	63.2	34.6	2.2
Kapkun	15.2	63.4	33.0	3.6
	16.3	64.8	32.6	2.6
	16.0	63.4	33.4	3.2
	14.4	63.6	33.0	3.4
	17.4	63.8	33.2	3.0
Maoi	16.3	61.8	35.4	2.8
	14.2	64.2	33.4	24
	15.2	64.4	32.2	3.4
	15.7	63.8	33.8	2.4
	15.7	63.8	34.0	2.2

Chemical properties of Gum arabic of the four Study sites

	N	Zn	Fe	Mn	Cu
	%		ppm		
Site					
Kimorok					
1	0.31	43.48	1490.46	0.2	61.39
2	0.21	44.73	1336.92	11.94	29.51
3	0.42	54.79	1243.3	10.55	5.93
4	0.42	161.32	1552.6	21.9	13.64
5	0.42	23.49	1228	37.053	12.01
6	0.42	74.44	1197.42	45.99	14.98
7	0.21	47.32	1409.63	63.11	14.83
8	0.21	31.13	249.66	107.1	35
9	0.31	14.12	236.24	117.1	35.74
10	0.21	82.48	846.96	154.88	45.23
11	0.31	19.09	169.77	120.84	45.52
12	0.31	117.75	159.78	148.91	55.9
13	0.21	188.87	1461.13	75.01	90.98
14	0.42	151.56	1121.44	74.09	92.61
15	0.31	153.31	1168.31	72.9	93.43
16	0.31	164.13	953.69	78.58	6.38
17	0.31	38.12	1180.4	70.13	1.09

SOLIT	Ν	Zn	Fe	Mn	Cu
	%	ppm	ppm	ppm	ppm
1	0.42	34.41	1504.5	44	29.21
2	0.31	17.57	1560.05	61.12	21.2
3	0.21	37.65	1470.17	78.04	52.79
4	0.31	36.49	1551.62	67.69	27.43
5	0.21	45.29	- 23.41	73.66	23.28
6	0.42	34.11	22.47	72.66	26.25
7	0.42	39.68	343.9	107.9	35.88
8	0.21	12.69	644.43	113.08	40.93
9	0.21	9.76	563.76	252.83	40.63
10	0.21	98.28	625.71	106.08	38.55
11	0.21	188.87	1461.13	75.01	90.98
12	0.42	151.56	1121.44	74.09	92.61
13	0.31	152.58	933.96	70.92	96.41
14	0.21	153.68	910.28	68.41	0.81
15	0.52	156.49	1070.87	74.35	1.49
16	0.31	125.95	161.97	141.74	52.05
17	0.21	29.05	887.8	100.77	6.52

UNIVERSITY OF NAIROBI LIBRAR P. O. Box 30197 NAIROBI

MAOI	Ν	Zn	Fe	Mn	Cu
	%	ppm	ppm	ppm	ppm
1	0.31	11.37	926.8	71.58	12.49
2	0.42	8.41	827.64	76.34	17.79
3	0.42	12.61	751.16	69.73	20.1
4	0.42	2.45	698.13	72.77	20.51
5	0.31	11.77	872.04	78.71	24.71
6	0.31	7.38	833.06	77.66	24.85
7	0.31	24.34	1234.18	82.81	26.21
8	0.42	20.9	1112.81	82.15	27.29
9	0.31	10.53	977.38	77.52	29.88
10	0.42	23.86	1012.65	80.17	36.66
11	0.31	50.29	1014.38	89.94	44.95
12	0.31	15.79	839.23	81.88	36.66
13	0.31	28.87	885.85	83.33	32.73
14	0.42	50.14	1174.23	90.2	39.38
15	0.21	169.13	666.9	149.9	52.05
16	0.21	146.85	188.18	146.52	48.19
17	0.31	151.86	997.7	162.25	50.17

KAPKUN	Ν	Zn	Fe	Mn	Cu
	%	ppm	ppm	ppm	ppm
1	0.42	30	1005.25	96.54	35.71
2	0.31	95.82	1440.9	103.01	42.23
3	0.31	41.48	1021.78	89.81	38.84
4	0.31	150.42	1207.78	115.3	74.82
5	0.31	60.3	1189.28	95.22	38.97
6	0.31	76.13	1345.68	112.92	37.48
7	0.31	100.76	1092.83	103.67	41.15
8	0.21	119-18	1027.45	106.45	41.69
9	0.31	111.76	984.78	99.71	44.27
10	0.21	124.37	987.98	105.92	42.23
11	0.21	155.76	1112.81	106.05	41.55
12	0.31	150.94	990.94	105.65	41.69
13	0.21	152.43	970.22	108.03	45.76
14	0.21	162.01	1030.41	115.69	46.44
15	0.31	170.3	1192.98	127.055	49.7
16	0.21	170.52	1356.04	120.31	53.23
17	0.31	174.54	1192.73	122.16	47.26
18	0.31	184.37	1162.4	128.11	50.92